# Section 1: 10-Q (10-Q)

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington D.C. 20549

# **FORM 10-Q**

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2018

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from \_\_\_\_\_ to \_\_\_\_\_

Commission File Number: 0-24260



# **AMEDISYS, INC.**

(Exact Name of Registrant as Specified in its Charter)

Delaware

(State or other jurisdiction of incorporation or organization)

11-3131700

(I.R.S. Employer Identification No.)

**3854** American Way, Suite A, Baton Rouge, LA 70816 (Address of principal executive offices, including zip code)

(225) 292-2031 or (800) 467-2662 (Registrant's telephone number, including area code) Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes  $\boxtimes$  No  $\square$ 

Indicate by check mark whether the registrant has submitted electronically, every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes  $\boxtimes$  No  $\square$ 

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer	X	Accelerated filer	
Non-accelerated filer		Smaller reporting company	
Emerging growth company			

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.  $\Box$ 

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

The number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date, is as follows: Common stock, \$0.001 par value, 31,888,599 shares outstanding as of October 26, 2018.

# TABLE OF CONTENTS

1

# SPECIAL CAUTION CONCERNING FORWARD-LOOKING STATEMENTS

PART I.	FINANCIAL INFORMATION	
ITEM 1.	FINANCIAL STATEMENTS:	
	CONDENSED CONSOLIDATED BALANCE SHEETS AS OF SEPTEMBER 30, 2018 (UNAUDITED) AND DECEMBER 31, 2017	2
	<u>CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS FOR THE THREE AND NINE-MONTH PERIODS ENDED</u> <u>SEPTEMBER 30, 2018 AND 2017 (UNAUDITED)</u>	3
	CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS FOR THE NINE-MONTH PERIODS ENDED SEPTEMBER 30, 2018 AND 2017 (UNAUDITED)	4
ITEM 2.	MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	21
ITEM 3	QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK	34
ITEM 4.	CONTROLS AND PROCEDURES	34
PART II. OT	THER INFORMATION	
ITEM 1.	LEGAL PROCEEDINGS	36
ITEM 1A.	RISK FACTORS	36
ITEM 2.	UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS	36
ITEM 3.	DEFAULTS UPON SENIOR SECURITIES	36
ITEM 4.	MINE SAFETY DISCLOSURES	36
ITEM 5.	OTHER INFORMATION	36
ITEM 6.	<u>EXHIBITS</u>	37
SIGNATUR	ES	38

#### SPECIAL CAUTION CONCERNING FORWARD-LOOKING STATEMENTS

When included in this Quarterly Report on Form 10-Q, or in other documents that we file with the Securities and Exchange Commission ("SEC") or in statements made by or on behalf of the Company, words like "believes," "belief," "expects," "plans," "anticipates," "intends," "projects," "estimates," "may," "might," "would," "should" and similar expressions are intended to identify forward-looking statements as defined by the Private Securities Litigation Reform Act of 1995. These forward-looking statements involve a variety of risks and uncertainties that could cause actual results to differ materially from those described therein. These risks and uncertainties include, but are not limited to the following: changes in Medicare and other medical payment levels, our ability to open care centers, acquire additional care centers and integrate and operate these care centers effectively, changes in or our failure to comply with existing federal and state laws or regulations or the inability to comply with new government regulations on a timely basis, competition in the healthcare industry, our ability to integrate our personal care segment into our business efficiently, changes in the case mix of patients and payment methodologies, changes in estimates and judgments associated with critical accounting policies, our ability to maintain or establish new patient referral sources, our ability to attract and retain qualified personnel, changes in payments and covered services due to an economic downturn and deficit spending by federal and state governments, future cost containment initiatives undertaken by third-party payors, our access to financing, our ability to meet debt service requirements and comply with covenants in debt agreements, business disruptions due to natural disasters or acts of terrorism, our ability to integrate, manage and keep our information systems secure, our ability to comply with requirements stipulated in our corporate integrity agreement and changes in law or developments with respect to any litigation relating to the Company, including various other matters, many of which are beyond our control.

Because forward-looking statements are inherently subject to risks and uncertainties, some of which cannot be predicted or quantified, you should not rely on any forward-looking statement as a prediction of future events. We expressly disclaim any obligation or undertaking and we do not intend to release publicly any updates or changes in our expectations concerning the forward-looking statements or any changes in events, conditions or circumstances upon which any forward-looking statement may be based, except as required by law. For a discussion of some of the factors discussed above as well as additional factors, see our Annual Report on Form 10-K for the year ended December 31, 2017, filed with the SEC on February 28, 2018, particularly, Part I, Item 1A - Risk Factors therein, which are incorporated herein by reference and Part II, Item 1A. Risk Factors of this Quarterly Report on Form 10-Q. Additional risk factors may also be described in reports that we file from time to time with the SEC.

#### **Available Information**

Our company website address is www.amedisys.com. We use our website as a channel of distribution for important company information. Important information, including press releases, analyst presentations and financial information regarding our company, is routinely posted on and accessible on the Investor Relations subpage of our website, which is accessible by clicking on the tab labeled "Investors" on our website home page. Visitors to our website can also register to receive automatic e-mail and other notifications alerting them when new information is made available on the Investor Relations subpage of our website. In addition, we make available on the Investor Relations subpage of our website (under the link "SEC filings") free of charge our annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K, ownership reports on Forms 3, 4 and 5 and any amendments to those reports as soon as practicable after we electronically file such reports with the SEC. Further, copies of our Certificate of Incorporation and Bylaws, our Code of Ethical Business Conduct, our Corporate Governance Guidelines and the charters for the Audit, Compensation, Quality of Care, Compliance and Ethics and Nominating and Corporate Governance Committees of our Board are also available on the Investor Relations subpage of our website (under the link "Governance").

Additionally, the public may read and copy any of the materials we file with the SEC at the SEC's Public Reference Room at 100 F Street, NE, Room 1580, Washington, D.C. 20549. Information on the operation of the Public Reference Room may be obtained by calling the SEC at (800) SEC-0330. Our electronically filed reports can also be obtained on the SEC's internet site at http://www.sec.gov.



# PART I. FINANCIAL INFORMATION ITEM 1. FINANCIAL STATEMENTS

# AMEDISYS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (Amounts in thousands, except share data)

		otember 30, 2018 inaudited)	December 31, 2017		
ASSETS					
Current assets:					
Cash and cash equivalents	\$	14,010	\$	86,363	
Patient accounts receivable		195,030		201,196	
Prepaid expenses		11,679		7,329	
Other current assets		11,949		16,268	
Total current assets		232,668		311,156	
Property and equipment, net of accumulated depreciation of \$93,432 and \$146,814		30,215		31,122	
Goodwill		324,145		319,949	
Intangible assets, net of accumulated amortization of \$32,443 and \$30,610		44,309		46,061	
Deferred income taxes		41,149		56,064	
Other assets		53,406		49,130	
Total assets	\$	725,892	\$	813,482	
LIABILITIES AND EQUITY					
Current liabilities:					
Accounts payable	\$	24,519	\$	25,384	
Payroll and employee benefits		92,385		89,936	
Accrued expenses		101,177		89,104	
Current portion of long-term obligations		1,209		10,638	
Total current liabilities		219,290		215,062	
Long-term obligations, less current portion		54,853		78,203	
Other long-term obligations		6,253		3,791	
Total liabilities		280,396		297,056	
Commitments and Contingencies—Note 5					
Equity:					
Preferred stock, \$0.001 par value, 5,000,000 shares authorized; none issued or outstanding					
Common stock, \$0.001 par value, 60,000,000 shares authorized; 36,146,649 and 35,747,134 shares issued; and 31,877,278 and 33,964,767 shares outstanding		36		35	
Additional paid-in capital		592,539		568,780	
Treasury stock, at cost 4,269,371 and 1,782,367 shares of common stock		(240,536)		(53,713)	
Accumulated other comprehensive income		15		15	
Retained earnings		92,089		204	
Total Amedisys, Inc. stockholders' equity	_	444,143		515,321	
Noncontrolling interests		1,353		1,105	
Total equity		445,496		516,426	
Total liabilities and equity	\$	725,892	\$	813,482	

The accompanying notes are an integral part of these condensed consolidated financial statements.

# AMEDISYS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Amounts in thousands, except per share data)

(Unaudited)

	For the Three-Month Periods Ended September 30				_1	Month ptember 30		
		2018		2017		2018		2017
Net service revenue	\$	417,335	\$	373,704	\$	1,228,200	\$	1,113,311
Cost of service, excluding depreciation and amortization		249,739		227,269		730,612		664,139
General and administrative expenses:								
Salaries and benefits		79,367		77,130		232,213		226,532
Non-cash compensation		4,842		3,558		12,653		11,788
Other		40,335		38,189		124,119		120,223
Depreciation and amortization		3,164		4,185		9,882		13,139
Securities Class Action Lawsuit settlement, net		—				—		28,712
Operating expenses		377,447		350,331		1,109,479		1,064,533
Operating income		39,888		23,373		118,721		48,778
Other income (expense):								
Interest income		29		44		263		104
Interest expense		(1,991)		(1,335)		(5,834)		(3,600)
Equity in earnings from equity method investments		1,625		900		6,461		3,149
Miscellaneous, net		1,822		1,043		2,782		3,282
Total other income, net		1,485		652		3,672		2,935
Income before income taxes		41,373		24,025		122,393		51,713
Income tax expense		(9,825)		(9,364)		(29,984)		(17,324)
Net income		31,548		14,661		92,409		34,389
Net income attributable to noncontrolling interests		(171)		(103)		(524)		(240)
Net income attributable to Amedisys, Inc.	\$	31,377	\$	14,558	\$	91,885	\$	34,149
Basic earnings per common share:								
Net income attributable to Amedisys, Inc. common stockholders	\$	0.99	\$	0.43	\$	2.78	\$	1.02
Weighted average shares outstanding		31,815		33,838		33,075		33,640
Diluted earnings per common share:								
Net income attributable to Amedisys, Inc. common stockholders	\$	0.96	\$	0.42	\$	2.71	\$	1.00
Weighted average shares outstanding		32,691		34,363		33,852		34,255

The accompanying notes are an integral part of these condensed consolidated financial statements.

# AMEDISYS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Amounts in thousands) (Unaudited)

		For the N Periods Ende		
		2018		2017
Cash Flows from Operating Activities:				
Net income	\$	92,409	\$	34,389
Adjustments to reconcile net income to net cash provided by operating activities:				
Depreciation and amortization		9,882		13,139
Non-cash compensation		12,653		11,788
401(k) employer match		6,934		6,547
Loss (gain) on disposal of property and equipment		738		(22)
Deferred income taxes		14,916		17,228
Equity in earnings from equity method investments		(6,461)		(3,149)
Amortization of deferred debt issuance costs/debt discount		596		555
Return on equity investment		4,373		4,656
Changes in operating assets and liabilities, net of impact of acquisitions:				
Patient accounts receivable		6,166		(10,846)
Other current assets		(32)		(5,896)
Other assets		726		(12,202)
Accounts payable		(670)		(5,430)
Accrued expenses		14,758		22,584
Other long-term obligations		2,462		201
Net cash provided by operating activities		159,450		73,542
Cash Flows from Investing Activities:				
Proceeds from sale of deferred compensation plan assets		563		622
Proceeds from the sale of property and equipment		51		118
Investments in equity method investees		(3,477)		(436)
Purchases of property and equipment		(5,684)		(9,074)
Acquisitions of businesses, net of cash acquired		(4,074)		(24,128)
Net cash used in investing activities		(12,621)		(32,898)
Cash Flows from Financing Activities:		(,)		(,,-)
Proceeds from issuance of stock upon exercise of stock options and warrants		2,749		4,214
Proceeds from issuance of stock to employee stock purchase plan		1,787		1,798
Shares withheld upon stock vesting		(5,421)		(6,454)
Noncontrolling interest distribution		(530)		(0,434)
Proceeds from borrowings under revolving line of credit		127,500		(210)
Repayments of borrowings under revolving line of credit		(70,000)		
Principal payments of long-term obligations		(91,071)		(4,069)
Debt issuance costs		(2,433)		(1,007)
Purchase of company stock		(181,402)		
Repurchase of noncontrolling interest		(361)		
Net cash used in financing activities		(219,182)		(1 727)
Net (decrease) increase in cash and cash equivalents				(4,727)
		(72,353)		35,917
Cash and cash equivalents at beginning of period	<u>ф</u>	86,363	¢	30,197
Cash and cash equivalents at end of period	\$	14,010	\$	66,114
Supplemental Disclosures of Cash Flow Information:				
Cash paid for interest	\$	2,989	\$	2,188
Cash paid for income taxes, net of refunds received	\$	11,017	\$	315

The accompanying notes are an integral part of these condensed consolidated financial statements.

#### 1. NATURE OF OPERATIONS, CONSOLIDATION AND PRESENTATION OF FINANCIAL STATEMENTS

Amedisys, Inc., a Delaware corporation, (together with its consolidated subsidiaries, referred to herein as "Amedisys," "we," "us," or "our") is a multi-state provider of home health, hospice and personal care services with approximately 73% and 74% of our revenue derived from Medicare for the three and nine-month periods ended September 30, 2018, respectively, and approximately 76% of our revenue derived from Medicare for the three and nine-month periods ended September 30, 2017. As of September 30, 2018, we owned and operated 323 Medicare-certified home health care centers, 83 Medicare-certified hospice care centers and 15 personal-care care centers in 34 states within the United States and the District of Columbia.

#### **Basis of Presentation**

In our opinion, the accompanying unaudited condensed consolidated financial statements contain all adjustments (consisting solely of normal recurring adjustments) necessary to present fairly our financial position, our results of operations, and our cash flows in accordance with U.S. generally accepted accounting principles ("U.S. GAAP") for interim financial reporting. Our results of operations for the interim periods presented are not necessarily indicative of results of our operations for the entire year and have not been audited by our independent auditors.

This report should be read in conjunction with our consolidated financial statements and related notes included in our Annual Report on Form 10-K for the year ended December 31, 2017, as filed with the Securities and Exchange Commission ("SEC") on February 28, 2018 (the "Form 10-K"), which includes information and disclosures not included herein. Certain information and footnote disclosures normally included in annual financial statements prepared in accordance with U.S. GAAP have been condensed or omitted from the interim financial information presented, as allowed by such SEC rules and regulations.

#### **Recently Adopted Accounting Pronouncements**

On January 1, 2018, the Company adopted Accounting Standards Update ("ASU") 2014-09, *Revenue from Contracts with Customers (Topic 606)*: *Deferral of the Effective Date* (collectively, "ASC 606"), the new accounting standards issued by the Financial Accounting Standards Board ("FASB") on revenue recognition, using the full retrospective method. ASC 606 outlines a single comprehensive model to use in accounting for revenue arising from contracts with customers. The standards supersede existing revenue recognition requirements and eliminate most industry-specific guidance from U.S. GAAP. The core principle of the revenue recognition standard is to require an entity to recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which it expects to be entitled in exchange for those goods or services. As a result of the Company's adoption of ASC 606, the revenue and related estimated uncollectible amounts owed to us by non-Medicare payors that were historically classified as a provision for doubtful accounts are now considered a price concession in determining net service revenue. Accordingly, the Company reports uncollectible balances associated with patient responsibility as a reduction of the transaction price and therefore, as a reduction in net service revenue (or as it relates to Hospice room and board, an increase in cost of service, excluding depreciation and amortization) when historically these amounts were classified as a provision for doubtful accounts within operating expenses within our condensed consolidated statements of operations. In addition, the adoption of ASC 606 resulted in increased disclosure, including qualitative and quantitative disclosures about the nature, amount, timing and uncertainty of revenue and cash flows arising from contracts with customers.

In August 2016, the FASB issued ASU 2016-15, *Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments*, which provides specific guidance on eight cash flow classification issues not specifically addressed by U.S. GAAP. The ASU is effective for annual and interim periods beginning after December 15, 2017. The standard should be applied using a retrospective transition method unless it is impractical to do so for some of the issues. In such case, the amendments for those issues would be applied prospectively as of the earliest date practicable. Our adoption of this standard on January 1, 2018, using a retrospective transition method for each period presented, did not have an effect on our condensed consolidated financial statements.

In January 2017, the FASB issued ASU 2017-01, *Business Combinations (Topic 805): Clarifying the Definition of a Business*, which provides guidance to assist entities with evaluating whether transactions should be accounted for as an acquisition (or disposal) of assets or a business. The ASU is effective for annual and interim periods beginning after December 15, 2017. We

adopted this ASU effective January 1, 2018, on a prospective basis. The impact on our consolidated financial statements and related disclosures will depend on the facts and circumstances of any specific future transactions as evaluated under the new framework.

In January 2017, the FASB issued ASU 2017-04, *Intangibles—Goodwill and Other (Topic 350)—Simplifying the Test for Goodwill Impairment*, which eliminates the requirement to calculate the implied fair value of goodwill to measure a goodwill impairment charge (Step 2 of the goodwill impairment test). Instead, impairment will be measured using the difference between the carrying amount and the fair value of the reporting unit. The ASU is effective for annual and interim periods beginning after December 15, 2019. Early adoption is permitted. We adopted this ASU effective January 1, 2018, on a prospective basis and will apply this guidance to our future tests of goodwill impairment.

#### **Recently Issued Accounting Pronouncements**

In February 2016, the FASB issued ASU 2016-02, *Leases (Topic 842)*, which will require lessees to recognize a lease liability and right-of-use asset for all leases (with the exception of short-term leases) at the commencement date. The ASU is effective for annual and interim periods beginning on or after December 15, 2018. Early adoption is permitted. In July 2018, the FASB issued ASU 2018-11, *Leases (Topic 842): Targeted Improvements,* which provides for a prospective transition method for the recognition and disclosure requirements under the new guidance. While the Company expects adoption of this standard to lead to a material increase in the assets and liabilities recorded on our balance sheet, we are still evaluating the overall impact on our consolidated financial statements and related disclosures and the effect of the standard on our ongoing financial reporting.

# Use of Estimates

Our accounting and reporting policies conform with U.S. GAAP. In preparing the unaudited condensed consolidated financial statements, we are required to make estimates and assumptions that impact the amounts reported in the condensed consolidated financial statements and accompanying notes. Actual results could differ from those estimates.

# **Reclassifications and Comparability**

Certain reclassifications have been made to prior periods' financial statements in order to conform to the current period's presentation. Effective January 1, 2018, we adopted ASC 606 on a full retrospective basis which required the reclassification of certain previously reported results. See Note 2 - Summary of Significant Accounting Policies for further details on the impact of the adoption of ASC 606.

# **Principles of Consolidation**

These unaudited condensed consolidated financial statements include the accounts of Amedisys, Inc., and our wholly owned subsidiaries. All significant intercompany accounts and transactions have been eliminated in our accompanying unaudited condensed consolidated financial statements, and business combinations accounted for as purchases have been included in our unaudited condensed consolidated financial statements from their respective dates of acquisition. In addition to our wholly owned subsidiaries, we also have certain equity investments that are accounted for as set forth below.

#### **Equity Investments**

We consolidate investments when the entity is a variable interest entity and we are the primary beneficiary or if we have controlling interests in the entity, which is generally ownership in excess of 50%. Third party equity interests in our consolidated joint ventures are reflected as noncontrolling interests in our condensed consolidated financial statements.

We account for investments in entities in which we have the ability to exercise significant influence under the equity method if we hold 50% or less of the voting stock and the entity is not a variable interest entity in which we are the primary beneficiary. During the three-month period ended September 30, 2018, we made a \$3.3 million investment in a healthcare analytics company; this investment will be accounted for under the equity method. The book value of investments that we account for under the equity method of accounting was \$31.9 million and \$26.4 million as of September 30, 2018 and December 31, 2017, respectively and is reflected in other assets within our condensed consolidated balance sheets. We account for investments in entities in which we have less than a 20% ownership interest under the cost method of accounting if we do not have the ability to exercise significant influence over the investee.

# 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

# **Revenue Recognition**

Our adoption of ASC 606 on January 1, 2018, on a full retrospective basis, impacted the Company's previously reported results as follows (amounts in thousands, unaudited):

	As Pre	viously Reported	As Adjusted	
			1	
Condensed Consolidated Balance Sheets				
Patient accounts receivable	\$	201,196	\$	\$ 201,196
Allowance for doubtful accounts	\$	20,866	\$ 20,866	\$
		For the three-m	nonth period ended Sept	ember 30, 2017
Condensed Consolidated Statements of Operations				
Net service revenue	\$	380,163	\$ (6,459)	\$ 373,704
Cost of service, excluding depreciation and amortization	\$	226,642	\$ 627	\$ 227,269
Provision for doubtful accounts	\$	7,086	\$ (7,086)	\$
Net income attributable to Amedisys, Inc.	\$	14,558	\$	\$ 14,558
		For the nine-m	onth period ended Sept	ember 30, 2017
Condensed Consolidated Statements of Operations				
Net service revenue	\$	1,129,442	\$ (16,131)	\$ 1,113,311
Cost of service, excluding depreciation and amortization	\$	662,192	\$ 1,947	\$ 664,139
Provision for doubtful accounts	\$	18,078	\$ (18,078)	\$
Net income attributable to Amedisys, Inc.	\$	34,149	\$	\$ 34,149
Condensed Consolidated Statements of Cash Flows				
Provision for doubtful accounts	\$	18,078	\$ (18,078)	\$
Changes in operating assets and liabilities, net of impact of acquisitions:				
Patient accounts receivable	\$	(28,924)	\$ 18,078	\$ (10,846)

We earn net service revenue through our home health, hospice and personal care segments through the delivery of a variety of services that best suit our patients' needs, whether that is home-based recovery and rehabilitation after an operation or injury, care that empowers patients to manage a chronic disease, hospice care at the end of life, or providing assistance with daily activities through our personal care segment. We account for revenue from contracts with customers in accordance with ASC 606, and as such, we recognize revenue in the period in which we satisfy our performance obligations under our contracts by transferring our promised services to our customers in amounts that reflect the consideration to which we expect to be entitled in exchange for providing patient care, which are the transaction prices allocated to the distinct services. The Company's cost of obtaining contracts is not material.

Revenues are recognized as performance obligations are satisfied, which varies based on the nature of the services provided. Our performance obligation is the delivery of patient care services in accordance with the nature and frequency of services outlined in physicians' orders, which are determined by a physician based on a patient's specific goals.

The Company's performance obligations relate to contracts with a duration of less than one year; therefore, the Company has elected to apply the optional exemption provided by ASC 606 and is not required to disclose the aggregate amount of the transaction

price allocated to performance obligations that are unsatisfied or partially unsatisfied as of the end of the reporting period. The unsatisfied or partially unsatisfied performance obligations are generally completed when the patients are discharged, which generally occurs within days or weeks of the end of the reporting period.

We determine the transaction price based on gross charges for services provided, reduced by estimates for price concessions, which include items such as contractual adjustments provided to third-party payors, reserves provided for self-pay or uninsured patients or other payors, adjustments resulting from payment reviews and adjustments arising from our inability to obtain appropriate billing documentation, authorizations or face-to-face documentation. Subsequent changes to the estimate of the transaction price are recorded as adjustments to net service revenue in the period of change. Subsequent changes that are determined to be the result of an adverse change in the patient's ability to pay (i.e. change in credit risk) are recorded as a provision for doubtful accounts.

We determine our estimates of contractual adjustments and reserves for self-pay, uninsured patients and other payors by major payor class based on contractual agreements with individual third-party payors, our historical collection experience, aged accounts receivable by payor and current economic conditions. The Company assesses its ability to collect for the healthcare services provided at the time of patient admission based on the Company's verification of the patient's insurance coverage under Medicare, Medicaid, and other commercial or managed care insurance programs. Medicare contributes to approximately 74% of the Company's consolidated net service revenue. The price concession included in estimating the transaction price represents the difference between amounts billed and amounts we expect to collect based on our collection history with similar payors.

Amounts due from third-party payors, primarily commercial health insurers and government programs (Medicare and Medicaid), include variable consideration for retroactive revenue adjustments due to settlements of audits and reviews. We determine our estimates for price concessions related to payment reviews based on our historical experience and success rates in the claim appeals and adjudication process. Revenue is recorded at amounts we estimate to be realizable for services provided.

We determine our estimates for price concessions related to our inability to obtain appropriate billing documentation, authorizations, or face-to-face documentation based on our historical experience, which primarily includes a historical collection rate of over 99% on Medicare claims.

#### Home Health Revenue Recognition

#### Medicare Revenue

Net service revenue is recorded under the Medicare prospective payment system ("PPS") based on an established Federal Medicare home health episode payment rate, that is subject to adjustment based on certain variables, including, but not limited to (a) an outlier payment if our patient's care was unusually costly (capped at 10% of total reimbursement per provider number); (b) a low utilization payment adjustment ("LUPA") if the number of visits was fewer than five; (c) a partial payment if our patient transferred to another provider or we received a patient from another provider before completing the episode; (d) a payment adjustment based upon the level of therapy services required (with various incremental adjustments made for additional visits, with larger payment increases associated with the sixth, fourteenth and twentieth visit thresholds); (e) adjustments to payments if we are unable to perform periodic therapy assessments; (f) the number of episodes of care provided to a patient, regardless of whether the same home health provider provided care for the entire series of episodes; (g) changes in the base episode payments established by the Medicare Program; (h) adjustments to the base episode payments for case mix and geographic wages; and (i) recoveries of overpayments. Medicare rates are based on the severity of the patient's condition, service needs and goals, and other factors relating to the cost of providing services and supplies, bundled into an episode of care, not to exceed 60 days. An episode starts the first day a billable visit is performed and ends 60 days later or upon discharge, if earlier, with multiple continuous episodes allowed.

The Medicare home health benefit requires that beneficiaries be homebound (meaning that the beneficiary is unable to leave their home without a considerable and taxing effort), require intermittent skilled nursing, physical therapy or speech therapy services, and receive treatment under a plan of care established and periodically reviewed by a physician. All Medicare contracts are required to have a signed plan of care which represents a single performance obligation, comprising of the delivery of a series of distinct services that are substantially similar and have a similar pattern of transfer to the customer. Accordingly, the Company accounts for the series of services ("episode") as a single performance obligation satisfied over time, as the customer simultaneously receives and consumes the benefits of the goods and services provided. Expected Medicare revenue per episode is recognized based on a pro-rated service output method, utilizing our historical average length of episode prior to discharge.

The base episode payment can be adjusted based on each patient's health including clinical condition, functional abilities, and service needs, as well as for the applicable geographic wage index, low utilization, patient transfers and other factors. The services covered by the episode payment include all disciplines of care in addition to medical supplies. Medicare can also make various adjustments to payments received if we are unable to produce appropriate billing documentation or acceptable authorizations. In addition, we make adjustments to Medicare revenue if we find we are unable to obtain appropriate billing documentation, authorizations or face-to-face documentation. We estimate the impact of such adjustments based on our historical experience, which primarily includes a historical collection rate of over 99% on Medicare claims, and record this estimate during the period in which services are rendered as an estimated price concession and a corresponding reduction to patient accounts receivable.

A portion of reimbursement from each Medicare episode is billed near the start of each episode, and cash is typically received before all services are rendered. The amount of revenue recognized for episodes of care which are incomplete at period end is based on the company's average percentage of days complete on episodes as of the end of the year. As of September 30, 2018 and 2017, the difference between the cash received from Medicare for a request for anticipated payment ("RAP") on episodes in progress and the associated estimated revenue was immaterial and, therefore, the resulting credits were recorded as a reduction to our outstanding patient accounts receivable in our condensed consolidated balance sheets for such periods.

# Non-Medicare Revenue

*Episodic-based Revenue*. We recognize revenue in a similar manner as we recognize Medicare revenue for episodic-based rates that are paid by other insurance carriers, including Medicare Advantage programs; however, these rates can vary based upon the negotiated terms which generally range from 90% to 100% of Medicare rates.

*Non-episodic based Revenue.* Gross revenue is recorded on an accrual basis based upon the date of service at amounts equal to our established or estimated per-visit rates. Contractual adjustments are recorded for the difference between our standard rates and the contracted rates to be realized from patients, third parties and others for services provided and are deducted from gross revenue to determine net service revenue. We also make adjustments to non-episodic revenue for any implicit price concessions, based on historical experience, to reflect the estimated transaction price. We receive a minimal amount of our net service revenue from patients who are either self-insured or are obligated for an insurance co-payment.

# Hospice Revenue Recognition

# Hospice Medicare Revenue

Gross revenue is recorded on an accrual basis based upon the date of service at amounts equal to the estimated payment rates. The estimated payment rates are predetermined daily or hourly rates for each of the four levels of care we deliver. The four levels of care are routine care, general inpatient care, continuous home care and respite care. Routine care accounted for 98% and 97% of our total net Medicare hospice service revenue for each of the three and nine-month periods ended September 30, 2018, respectively, and 97% of our total net Medicare hospice service revenue for each of the three and nine-month periods ended September 30, 2017. There are two separate payment rates for routine care: payments for the first 60 days of care and care beyond 60 days. In addition to the two routine rates, we may also receive a service intensity add-on ("SIA"). The SIA is based on visits made in the last seven days of life by a registered nurse ("RN") or medical social worker ("MSW") for patients in a routine level of care.

The performance obligation is the delivery of hospice services to the patient, as determined by a physician, each day the patient is on hospice care.

We make adjustments to Medicare revenue for our inability to obtain appropriate billing documentation or acceptable authorizations and other reasons unrelated to credit risk. We estimate the impact of these adjustments based on our historical experience, which primarily includes a historical collection rate of over 99% on Medicare claims, and record it during the period services are rendered as an estimated price concession and as a reduction to our outstanding patient accounts receivable.

Additionally, our hospice service revenue is subject to certain limitations on payments from Medicare which are considered variable consideration. We are subject to an inpatient cap limit and an overall Medicare payment cap for each provider number. We monitor these caps on a provider-by-provider basis and estimate amounts due back to Medicare if we estimate a cap has been exceeded. We record these adjustments as a reduction to revenue and an increase in accrued expenses within our condensed consolidated balance sheet. Beginning for the cap year ending October 31, 2017, providers are required to self-report and pay their estimated cap liability by February 28th of the following year. As of September 30, 2018, we have settled our Medicare hospice reimbursements for all fiscal years through October 31, 2012. As of September 30, 2018, we have recorded \$1.6 million for estimated amounts due back to Medicare in accrued expenses for the Federal cap years ended October 31, 2013 through September 30, 2018. As of December 31,

2017, we had recorded \$0.9 million for estimated amounts due back to Medicare in accrued expenses for the Federal cap years ended October 31, 2013 through September 30, 2018.

#### Hospice Non-Medicare Revenue

Gross revenue is recorded on an accrual basis based upon the date of service at amounts equal to our established rates or estimated per day rates, as applicable. Contractual adjustments are recorded for the difference between our established rates and the amounts estimated to be realizable from patients, third parties and others for services provided and are deducted from gross revenue to determine our net service revenue. We also make adjustments to non-Medicare revenue for any implicit price concessions, based on historical experience, to reflect the estimated transaction price.

#### Personal Care Revenue Recognition

#### Personal Care Revenue

We generate net service revenues by providing our services directly to patients based on authorized hours, visits or units determined by the relevant agency, at a rate that is either contractual or fixed by legislation. Net service revenue is recognized at the time services are rendered based on gross charges for the services provided, reduced by estimates for price concessions. We receive payment for providing such services from payors, including state and local governmental agencies, managed care organizations, commercial insurers and private consumers. Payors include the following elder service agencies: Aging Services Access Points (ASAPs), Senior Care Options (SCOs), Program of All-Inclusive Care for the Elderly (PACE) and the Veterans Administration (VA).

#### Patient Accounts Receivable

We report accounts receivable from services rendered at their estimated transaction price, which includes price concessions based on the amounts expected to be due from payors. Our patient accounts receivable are uncollateralized and consist of amounts due from Medicare, Medicaid, other third-party payors and patients. As of September 30, 2018, there is only one single payor, other than Medicare, that accounts for more than 10% of our total outstanding patient receivables (approximately 11%). Thus, we believe there are no other significant concentrations of receivables that would subject us to any significant credit risk in the collection of our patient accounts receivable. We write off accounts on a monthly basis once we have exhausted our collection efforts and deem an account to be uncollectible. We believe the collectibility risk associated with our Medicare accounts, which represent 55% and 59% of our patient accounts receivable at September 30, 2018 and December 31, 2017, respectively, is limited due to our historical collection rate of over 99% from Medicare and the fact that Medicare is a U.S. government payor.

We do not believe there are any significant concentrations of revenues from any payor that would subject us to any significant credit risk in the collection of our accounts receivable.

#### Medicare Home Health

For our home health patients, our pre-billing process includes verifying that we are eligible for payment from Medicare for the services that we provide to our patients. Our Medicare billing begins with a process to ensure that our billings are accurate through the utilization of an electronic Medicare claim review. We submit a RAP for 60% of our estimated payment for the initial episode at the start of care or 50% of the estimated payment for any subsequent episodes of care contiguous with the first episode for a particular patient. The full amount of the episode is billed after the episode has been completed ("final billed"). The RAP received for that particular episode is then deducted from our final payment. If a final bill is not submitted within the greater of 120 days from the start of the episode, or 60 days from the date the RAP was paid, any RAPs received for that episode will be recouped by Medicare from any other claims in process for that particular provider number. The RAP and final claim must then be resubmitted.

#### Medicare Hospice

For our hospice patients, our pre-billing process includes verifying that we are eligible for payment from Medicare for the services that we provide to our patients. Our Medicare billing begins with a process to ensure that our billings are accurate through the utilization of an electronic Medicare claim review. We bill Medicare on a monthly basis for the services provided to the patient.

#### Non-Medicare Home Health, Hospice and Personal Care

For our non-Medicare patients, our pre-billing process primarily begins with verifying a patient's eligibility for services with the applicable payor. Once the patient has been confirmed for eligibility, we will provide services to the patient and bill the applicable payor. Our review and evaluation of non-Medicare accounts receivable includes a detailed review of outstanding balances and special consideration to concentrations of receivables from particular payors or groups of payors with similar characteristics that would subject us to any significant credit risk.

# **Property and Equipment**

Property and equipment is stated at cost and we depreciate it on a straight-line basis over the estimated useful lives of the assets. Additionally, we have internally developed computer software for our own use. Additions and improvements (including interest

costs for construction of qualifying long-lived assets) are capitalized. Maintenance and repair expenses are charged to expense as incurred. The cost of property and equipment sold or disposed of and the related accumulated depreciation are eliminated from the property and related accumulated depreciation accounts, and any gain or loss is credited or charged to other general and administrative expenses.

During the nine-month period ended September 30, 2018, we reviewed the balances of our property and equipment and as a result, eliminated those asset balances for which the asset was no longer in service. The following table summarizes the balances related to our property and equipment for the periods indicated (amounts in millions):

	As of S	eptember 30, 2018	As of Decer	nber 31, 2017
Building and leasehold improvements	\$	8.7	\$	7.8
Equipment and furniture		55.9		72.9
Computer software		59.0		97.2
		123.6		177.9
Less: accumulated depreciation		(93.4)		(146.8)
	\$	30.2	\$	31.1

# Fair Value of Financial Instruments

The following details our financial instruments where the carrying value and the fair value differ (amounts in millions):

			Fair Value at <b>R</b>	eporting D	ate Using				
			Quoted Prices in Active Markets for Identical		nificant Other	Signi	ificant		
<u>Financial Instrument</u>	•	ng Value as of 1ber 30, 2018	Items (Level 1)	Obs	servable Inputs (Level 2)	Unobservable Input (Level 3)			
Long-term obligations	\$	58.1	\$	\$	60.2	\$	_		

The fair value hierarchy is based on three levels of inputs, of which the first two are considered observable and the last unobservable, that may be used to measure fair value. The three levels of inputs are as follows:

- Level 1 Quoted prices in active markets for identical assets and liabilities.
- Level 2 Inputs other than Level 1 that are observable, either directly or indirectly, such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.
- Level 3 Unobservable inputs that are supported by little or no market activity and are significant to the fair value of the assets or liabilities.

Our deferred compensation plan assets are recorded at fair value and are considered a level 2 measurement. For our other financial instruments, including our cash and cash equivalents, patient accounts receivable, accounts payable, payroll and employee benefits and accrued expenses, we estimate the carrying amounts approximate fair value.

# Weighted-Average Shares Outstanding

Net income per share attributable to Amedisys, Inc. common stockholders, calculated on the treasury stock method, is based on the weighted average number of shares outstanding during the period. The following table sets forth, for the periods indicated, shares used in our computation of the weighted-average shares outstanding, which are used to calculate our basic and diluted net income attributable to Amedisys, Inc. common stockholders (amounts in thousands):

	For the 7 Month P Ended Septe	eriods		e Nine- Periods ptember 30,		
	2018	2017	2018	2017		
Weighted average number of shares outstanding - basic	31,815	33,838	33,075	33,640		
Effect of dilutive securities:						
Stock options	561	271	464	279		
Non-vested stock and stock units	315	254	313	336		
Weighted average number of shares outstanding - diluted	32,691	34,363	33,852	34,255		
Anti-dilutive securities	51	337	76	279		

# **3. ACQUISITIONS**

We complete acquisitions from time to time in order to pursue our strategy of increasing our market presence by expanding our service base and enhancing our position in certain geographic areas as a leading provider of home health, hospice and personal care services. The purchase price paid for acquisitions is negotiated through arm's length transactions, with consideration based on our analysis of, among other things, comparable acquisitions and expected cash flows. Acquisitions are accounted for as purchases and are included in our consolidated financial statements from their respective acquisition dates. Goodwill generated from acquisitions is recognized for the excess of the purchase price over tangible and identifiable intangible assets because of the expected contributions of the acquisitions to our overall corporate strategy. We typically engage outside appraisal firms to assist in the fair value determination of identifiable intangible assets for significant acquisitions. The preliminary purchase price allocation is adjusted, as necessary, up to one year after the acquisition closing date if management obtains more information regarding asset valuations and liabilities assumed.

On March 1, 2018, we acquired the assets of Christian Care at Home which provides home health services to the state of Kentucky for a total purchase price of \$2.3 million. The purchase price was paid with cash on hand on the date of the transaction. Based on our preliminary purchase price allocation, we recorded goodwill of \$2.3 million in connection with the acquisition during the three-month period ended March 31, 2018.

On May 1, 2018, we acquired the assets of East Tennessee Personal Care Services which owns and operates one personal-care care center servicing the state of Tennessee for a total purchase price of \$2.0 million (subject to certain adjustments, of which \$0.2 million was placed in a promissory note to be paid over 24 months, subject to any offsets or withholds for indemnification purposes). The purchase price was paid with cash on hand on the date of the transaction. During the three-month period ended June 30, 2018, we recorded goodwill of \$1.9 million and other intangibles - non-compete agreements of \$0.1 million in connection with the acquisition. The non-compete agreement will be amortized over a weighted-average period of 2.8 years.

# 4. LONG-TERM OBLIGATIONS

Long-term debt consisted of the following for the periods indicated (amounts in millions):

	September 30, 2018	December 31, 2017
\$100.0 million Term Loan; principal payments plus accrued interest payable quarterly; interest rate at Base Rate plus Applicable Rate or Eurodollar Rate plus the Applicable Rate; due August 28, 2020	\$ —	\$ 90.0
\$550.0 million Revolving Credit Facility; interest only payments; interest rate at Base Rate plus Applicable Rate or Eurodollar Rate plus the Applicable Rate (3.7% at September 30, 2018); due June 29, 2023	57.5	_
Promissory notes	0.6	0.7
Capital leases	1.7	—
Principal amount of long-term obligations	59.8	90.7
Deferred debt issuance costs	(3.7)	(1.9)
	56.1	88.8
Current portion of long-term obligations	(1.2)	(10.6)
Total	\$ 54.9	\$ 78.2

# **Credit Agreement**

On June 29, 2018, we entered into our Amended and Restated Credit Agreement ("Credit Agreement") that provides for a senior secured revolving credit facility in an initial aggregate principal amount of up to \$550.0 million (the "Revolving Credit Facility"). The Revolving Credit Facility provides for and includes within its \$550.0 million limit a \$25.0 million swingline facility and commitments for up to \$60.0 million in letters of credit. Upon lender approval, we may increase the aggregate loan amount under the Revolving Credit Facility by either i) \$125.0 million or ii) an unlimited amount subject to a leverage limit of 0.5x under the maximum allowable consolidated leverage ratio which is currently 3.0x per the Credit Agreement.

The funds available under the Revolving Credit Facility were used to pay off our existing indebtedness under our prior credit agreement, dated as of August 28, 2015 (the "Prior Credit Agreement"), with a principal balance of \$127.5 million. The final maturity of the Revolving Credit Facility is June 29, 2023 and there is no mandatory amortization on the outstanding principal balances which are payable in full upon maturity. The Revolving Credit Facility may be used to provide ongoing working capital and for general corporate purposes of the Company and our subsidiaries, including permitted acquisitions, as defined in the Credit Agreement.

The interest rate on borrowings under the Revolving Credit Facility shall be selected by us from the following: (i) the Base Rate plus the Applicable Rate. The "Base Rate" means a fluctuating rate per annum equal to the highest of (a) the federal funds rate plus 0.50% per annum, (b) the prime rate of interest established by the Administrative Agent, and (c) the Eurodollar Rate plus 1% per annum. The "Eurodollar Rate" means the quoted rate per annum equal to the London Interbank Offered Rate ("LIBOR") or a comparable successor rate approved by the Administrative Agent for an interest period of one, two, three or six months (as selected by us). The "Applicable Rate" is based on the consolidated leverage ratio and is presented in the table below. As of September 30, 2018, the Applicable Rate is 0.50% per annum for Eurodollar Rate Loans. We are also subject to a commitment fee and letter of credit fee under the terms of the Credit Agreement, as presented in the table below.

			Letter of Cred	it	Base Rate
Pricing Tier	<b>Consolidated Leverage Ratio</b>	<b>Commitment Fee</b>	Fee	<b>Eurodollar Rate Loans</b>	Loans
Ι	> 3.00 to 1.0	0.35%	2.00%	2.25%	1.25%
II	$\leq$ 3.00 to 1.0 but $>$ 2.00 to 1.0	0.30%	1.75%	2.00%	1.00%
III	$\leq$ 2.00 to 1.0 but $>$ 1.00 to 1.0	0.25%	1.50%	1.75%	0.75%
IV	$\leq$ 1.00 to 1.0	0.20%	1.25%	1.50%	0.50%

The Credit Agreement requires maintenance of two financial covenants: (i) a consolidated leverage ratio of funded indebtedness to EBITDA, as defined in the Credit Agreement, and (ii) a consolidated interest coverage ratio of EBITDA to cash interest charges, as defined in the Credit Agreement. Each of these covenants is calculated over rolling four-quarter periods and also is subject to certain exceptions and baskets. The Credit Agreement also contains customary covenants, including, but not limited to, restrictions on: incurrence of additional debt, sales of assets and other fundamental corporate changes, investments, and declarations of dividends. These covenants contain customary exclusions and baskets as detailed in the Credit Agreement.

The Revolving Credit Facility is guaranteed by substantially all of our wholly-owned direct and indirect subsidiaries. The Credit Agreement requires at all times that we (i) provide guarantees from wholly-owned subsidiaries that in the aggregate represent not less than 95% of our consolidated net revenues and adjusted EBITDA from all wholly-owned subsidiaries and (ii) provide guarantees from subsidiaries that in the aggregate represent not less than 70% of consolidated adjusted EBITDA, subject to certain exceptions.

In connection with entering into the Credit Agreement, we entered into (i) a Security Agreement with the Administrative Agent dated June 29, 2018 and (ii) a Pledge Agreement with the Administrative Agent dated as of June 29, 2018 for the purpose of securing the payment of our obligations under the Credit Agreement. Pursuant to the Security Agreement and the Pledge Agreement, as of the effective date of the Credit Agreement, our obligations under the Credit Agreement are secured by (i) the grant of a first lien security interest in the non-real estate assets of substantially all of our direct and indirect, wholly-owned subsidiaries (subject to exceptions) and (ii) the pledge of the equity interests in (a) substantially all of our direct and indirect, wholly-owned corporate, limited liability company and limited partnership subsidiaries and (b) those joint ventures which constitute subsidiaries under the Credit Agreement (subject, in the case of the Pledge Agreement, to exceptions). In connection with our entry into the Credit Agreement, we recorded \$2.4 million in deferred debt issuance costs as long-term obligations, less current portion within our condensed consolidated balance sheet during the three-month period ended June 30, 2018.

Our weighted average interest rate for our \$100.0 million Term Loan, under our Prior Credit Agreement, was 3.2% and 3.0% for the three and ninemonth periods ended September 30, 2017, respectively. Our weighted average interest rate for borrowings under our \$550.0 million Revolving Credit Facility was 3.6% and 3.8% for the three and nine-month periods ended September 30, 2018, respectively.

As of September 30, 2018, our consolidated leverage ratio was 0.4, our consolidated interest coverage ratio was 57.6 and we are in compliance with our covenants under the Credit Agreement. In the event we are not in compliance with our debt covenants in the future, we would pursue various alternatives in an attempt to successfully resolve the non-compliance, which might include, among other things, seeking debt covenant waivers or amendments.

As of September 30, 2018, our availability under our \$550.0 million Revolving Credit Facility was \$458.1 million as we have \$57.5 million outstanding in borrowings and \$34.4 million outstanding in letters of credit.

# 5. COMMITMENTS AND CONTINGENCIES

# Legal Proceedings - Ongoing

We are involved in the following legal actions:

#### Subpoena Duces Tecum Issued by the U.S. Department of Justice

On May 21, 2015, we received a Subpoena Duces Tecum ("Subpoena") issued by the U.S. Department of Justice. The Subpoena requests the delivery of information regarding 53 identified hospice patients to the United States Attorney's Office for the District of Massachusetts. It also requests the delivery of documents relating to our hospice clinical and business operations and related compliance activities. The Subpoena generally covers the period from January 1, 2011 through May 21, 2015. We are fully cooperating with the U.S. Department of Justice with respect to this investigation. Based on the information currently available to us, we cannot predict the timing or outcome of this investigation or reasonably estimate the amount or range of potential losses, if any, which may arise from this matter.

#### Civil Investigative Demands Issued by the U.S. Department of Justice

On November 3, 2015, we received a civil investigative demand ("CID") issued by the U.S. Department of Justice pursuant to the federal False Claims Act relating to claims submitted to Medicare and/or Medicaid for hospice services provided through designated facilities in the Morgantown, West Virginia area. The CID requests the delivery of information to the United States Attorney's Office for the Northern District of West Virginia regarding 66 identified hospice patients, as well as documents relating to our hospice clinical and business operations in the Morgantown area. The CID generally covers the period from January 1, 2009 through August 31, 2015. We are fully cooperating with the U.S. Department of Justice with respect to this investigation. Based on the information currently available to us, we cannot predict the timing or outcome of this investigation or reasonably estimate the amount or range of potential losses, if any, which may arise from this matter.

On June 27, 2016, we received a CID issued by the U.S. Department of Justice pursuant to the federal False Claims Act relating to claims submitted to Medicare and/or Medicaid for hospice services provided through designated facilities in the Parkersburg, West Virginia area. The CID requests the delivery of information to the United States Attorney's Office for the Southern District of West Virginia regarding 68 identified hospice patients, as well as documents relating to our hospice clinical and business operations in the Parkersburg area. The CID generally covers the period from January 1, 2011 through June 20, 2016. We are fully cooperating with the U.S. Department of Justice with respect to this investigation. Based on the information currently available to us, we cannot predict the timing or outcome of this investigation or reasonably estimate the amount or range of potential losses, if any, which may arise from this matter.

In addition to the matters referenced in this note, we are involved in legal actions in the normal course of business, some of which seek monetary damages, including claims for punitive damages. We do not believe that these normal course actions, when finally concluded and determined, will have a material impact on our consolidated financial condition, results of operations or cash flows.

# Legal Proceedings - Settled

#### Securities Class Action Lawsuits

As previously disclosed, between June 10 and July 28, 2010, several putative securities class action complaints were filed in the United States District Court for the Middle District of Louisiana (the "District Court") against the Company and certain of our former senior executives. The cases were consolidated into the first-filed action Bach, et al. v. Amedisys, Inc., et al. Case No. 3:10-cv-00395, and the District Court appointed as co-lead plaintiffs the Public Employees' Retirement System of Mississippi and the Puerto Rico Teachers' Retirement System (the "Co-Lead Plaintiffs").

The Plaintiffs were granted leave to file a First Amended Consolidated Complaint (the "First Amended Securities Complaint") on behalf of all purchasers or acquirers of Amedisys' securities between August 2, 2005 and September 30, 2011. The First Amended Securities Complaint alleged that the Company and seven individual defendants violated Section 10(b), Section 20(a), and Rule 10b-5 of the Securities Exchange Act of 1934 by materially misrepresenting the Company's financial results and concealing a scheme to obtain higher Medicare reimbursements and additional patient referrals by (1) providing medically unnecessary care to patients, including certifying and re-certifying patients for medically unnecessary 60-day treatment episodes; (2) implementing clinical tracks such as "Balanced for Life" and wound care programs that provided a pre-set number of therapy visits irrespective of medical need; (3) "upcoding" patients' Medicare forms to attribute a "primary diagnosis" to a medical condition associated with higher billing rates; and (4) providing improper and illegal remuneration to physicians to obtain patient certifications or recertifications. The First Amended Securities Complaint sought certification of the case as a class action and an unspecified amount of damages, as well as interest and an award of attorneys' fees.

On June 12, 2017, the Company reached an agreement-in-principle to settle this matter. All parties to the action executed a binding term sheet that, subject to final documentation and court approval, provided in part for a settlement payment of approximately

\$43.7 million, which we accrued as of June 30, 2017, and the dismissal with prejudice of the litigation. Approximately \$15.0 million of the settlement amount paid by the Company's insurance carriers during the three-month period ended September 30, 2017, was previously recorded within other current assets in our condensed consolidated balance sheet as of June 30, 2017. The net of these two amounts, \$28.7 million, was recorded as a charge in our condensed consolidated statements of operations during the three-month period ended June 30, 2017 and paid with cash on hand during the three-month period ended September 30, 2017. On December 19, 2017, the Court entered the final order and judgment on the case.

#### **Other Investigative Matters - Ongoing**

#### Corporate Integrity Agreement

On April 23, 2014, with no admissions of liability on our part, we entered into a settlement agreement with the U.S. Department of Justice relating to certain of our clinical and business operations. Concurrently with our entry into this agreement, we entered into a corporate integrity agreement ("CIA") with the Office of Inspector General-HHS ("OIG"). The CIA formalizes various aspects of our already existing ethics and compliance programs and contains other requirements designed to help ensure our ongoing compliance with federal health care program requirements. Among other things, the CIA requires us to maintain our existing compliance program, executive compliance committee and compliance committee of the Board of Directors; provide certain compliance training; continue screening new and current employees to ensure they are eligible to participate in federal health care programs; engage an independent review organization to perform certain auditing and reviews and prepare certain reports regarding our compliance with federal health care programs; and provide certain reports and management certifications to the OIG. Additionally, the CIA specifically requires that we report substantial overpayments that we discover we have received from federal health care programs, as well as probable violations of federal health care laws. Upon breach of the CIA, we could become liable for payment of certain stipulated penalties, or could be excluded from participation in federal health care programs. The corporate integrity agreement has a term of five years.

#### Third Party Audits - Ongoing

From time to time, in the ordinary course of business, we are subject to audits under various governmental programs in which third party firms engaged by the Centers for Medicare and Medicaid Services ("CMS") conduct extensive review of claims data to identify potential improper payments under the Medicare program.

In July 2010, our subsidiary that provides hospice services in Florence, South Carolina received from a Zone Program Integrity Contractor ("ZPIC") a request for records regarding a sample of 30 beneficiaries who received services from the subsidiary during the period of January 1, 2008 through March 31, 2010 (the "Review Period") to determine whether the underlying services met pertinent Medicare payment requirements. We acquired the hospice operations subject to this review on August 1, 2009; the Review Period covers time periods both before and after our ownership of these hospice operations. Based on the ZPIC's findings for 16 beneficiaries, which were extrapolated to all claims for hospice services provided by the Florence subsidiary billed during the Review Period, on June 6, 2011, the Medicare Administrative Contractor ("MAC") for the subsidiary issued a notice of overpayment seeking recovery from our subsidiary of an alleged overpayment. We dispute these findings, and our Florence subsidiary has filed appeals through the Original Medicare Standard Appeals Process, in which we are seeking to have those findings overturned. An administrative law judge ("ALJ") hearing was held in early January 2015. On January 18, 2016, we received a letter dated January 6, 2016 referencing the ALJ hearing decision for the overpayment issued on June 6, 2011. The decision was partially favorable with a new overpayment amount of \$3.7 million with a balance owed of \$5.6 million, including interest, based on 9 disputed claims (originally 16). We filed an appeal to the Medicare Appeals Council on the remaining 9 disputed claims and also argued that the statistical method used to select the sample was not valid. No assurances can be given as to the timing or outcome of the Medicare Appeals Council decision. As of September 30, 2018, Medicare has withheld payments of \$5.7 million (including additional interest) as part of their standard procedures once this level of the appeal process has been reached. In the event we are not able to recoup this alleged overpayment, we are entitled to be indemnified by the prior owners of the hospice operations for amounts relating to the period prior to August 1, 2009. As of September 30, 2018, we have an indemnity receivable of approximately \$4.9 million for the amount withheld related to the period prior to August 1, 2009.

In July 2016, the Company received a request for medical records from SafeGuard Services, L.L.C ("SafeGuard"), a ZPIC related to services provided by some of the care centers that the Company acquired from Infinity Home Care, L.L.C. The review period covers time periods both before and after our ownership of the care centers, which were acquired on December 31, 2015. In August 2017, the Company received Requests for Repayment from Palmetto GBA, LLC ("Palmetto") regarding Infinity Home Care of

Lakeland, LLC ("Lakeland Care Centers") and Infinity Home Care of Pinellas, LLC ("Clearwater Care Center"). The Palmetto letters are based on statistical extrapolation performed by SafeGuard which alleged an overpayment of \$34.0 million for the Lakeland Care Centers on a universe of 72 Medicare claims totaling \$0.2 million in actual claims payments using a 100% error rate and an overpayment of \$4.8 million for the Clearwater Care Center on a universe of 70 Medicare claims totaling \$0.2 million in actual claims payments using a 100% error rate.

The Lakeland Request for Repayment covers claims between January 2, 2014 and September 13, 2016. The Clearwater Request for Repayment covers claims between January 2, 2015 and December 9, 2016. As a result of partially successful Level I and Level II Administrative Appeals, the alleged overpayment for the Lakeland Care Centers has been reduced to \$26.0 million and the alleged overpayment for the Clearwater Care Center has been reduced to \$3.3 million. The Company has now filed Level III Administrative Appeals, and will continue to vigorously pursue its appeal rights, which include contesting the methodology used by the ZPIC contractor to perform statistical extrapolation. The Company is contractually entitled to indemnification by the prior owners for all claims prior to December 31, 2015, for up to \$12.6 million.

At this stage of the review, based on the information currently available to the Company, the Company cannot predict the timing or outcome of this review. The Company estimates a low-end potential range of loss related to this review of \$6.5 million (assuming the Company is successful in seeking indemnity from the prior owners and unsuccessful in demonstrating that the extrapolation method used by SafeGuard was erroneous). The Company has reduced its high-end potential range of loss from \$38.8 million (the maximum amount Palmetto claims has been overpaid for both the Lakeland Care Centers and the Clearwater Care Center of which amount \$12.6 million is subject to indemnification by the prior owners) to \$29.3 million based on the partial success achieved by the Company in prosecuting its Level I and II Administrative Appeals.

As of September 30, 2018, we have an accrued liability of approximately \$17.4 million related to this matter. We expect to be indemnified by the prior owners for approximately \$10.9 million of the total \$12.6 million available indemnification related to this matter and have recorded this amount within other assets in our condensed consolidated balance sheet as of September 30, 2018. The net of these two amounts, \$6.5 million, was recorded as a reduction in revenue in our condensed consolidated statements of operations during the three-month period ended September 30, 2017. As of September 30, 2018, \$1.6 million of receivables have been impacted by this payment suspension.

# Compliance

From time to time, the Company performs internal reviews of claims data to identify potential improper payments under the Medicare program. Any overpayments are recorded as a reduction in revenue in our condensed consolidated statements of operations. As of September 30, 2018, we have recorded \$6.5 million to accrued expenses in our condensed consolidated balance sheet as a result of these reviews.

# Insurance

We are obligated for certain costs associated with our insurance programs, including employee health, workers' compensation and professional liability. While we maintain various insurance programs to cover these risks, we are self-insured for a substantial portion of our potential claims. We recognize our obligations associated with these costs, up to specified deductible limits in the period in which a claim is incurred, including with respect to both reported claims and claims incurred but not reported. These costs have generally been estimated based on historical data of our claims experience. Such estimates, and the resulting reserves, are reviewed and updated by us on a quarterly basis.

Our health insurance has an exposure limit of \$1.0 million for any individual covered life. Our workers' compensation insurance has a retention limit of \$0.5 million per incident and our professional liability insurance has a retention limit of \$0.3 million per incident.

17

#### 6. SEGMENT INFORMATION

Our operations involve servicing patients through our three reportable business segments: home health, hospice and personal care. Our home health segment delivers a wide range of services in the homes of individuals who may be recovering from surgery, have a chronic disability or terminal illness or need assistance with completing important personal tasks. Our hospice segment provides palliative care and comfort to terminally ill patients and their families. Our personal care segment provides patients with assistance with the essential activities of daily living. The "other" column in the following tables consists of costs relating to executive management and administrative support functions, primarily information services, accounting, finance, billing and collections, legal, compliance, risk management, procurement, marketing, clinical administration, training, human resources and administration.

18

Management evaluates performance and allocates resources based on the operating income of the reportable segments, which includes an allocation of corporate expenses directly attributable to the specific segment and includes revenues and all other costs directly attributable to the specific segment. Segment assets are not reviewed by the company's chief operating decision maker and therefore are not disclosed below (amounts in millions).

		For the Three-Month Period Ended September 30, 2018								
		Home Tealth		Hospice		Personal Care		Other		Total
Net service revenue	\$	294.9	\$	103.4	\$	19.0	\$		\$	417.3
Cost of service, excluding depreciation and amortization		181.8		53.4		14.5		_		249.7
General and administrative expenses		68.8		21.4		3.0		31.3		124.5
Depreciation and amortization		0.8		0.3		0.1		2.0		3.2
Operating expenses		251.4		75.1		17.6		33.3		377.4
Operating income (loss)	\$	43.5	\$	28.3	\$	1.4	\$	(33.3)	\$	39.9
· · · · Ø · · · · ( · · · /	Ψ	15.5	Ψ	20.5	Ψ	1.1	Ψ	(55:5)	Ψ	57.7

	For the Three-Month Period Ended September 30, 2017								
	Home Health		Hospice		Personal Care		Other		Total
Net service revenue	\$ 264.1	\$	95.9	\$	13.7	\$	_	\$	373.7
Cost of service, excluding depreciation and amortization	168.3		48.4		10.6				227.3
General and administrative expenses	70.8		19.0		3.1		25.9		118.8
Depreciation and amortization	0.9		0.2				3.1		4.2
Operating expenses	240.0		67.6		13.7		29.0		350.3
Operating income (loss)	\$ 24.1	\$	28.3	\$	_	\$	(29.0)	\$	23.4

	For the Nine-Month Period Ended September 30, 2018									
	Hom	e Health		Hospice		Personal Care		Other		Total
Net service revenue	\$	870.5	\$	302.1	\$	55.6	\$		\$	1,228.2
Cost of service, excluding depreciation and amortization		532.7		155.2		42.7				730.6
General and administrative expenses		205.2		61.7		9.5		92.6		369.0
Depreciation and amortization		2.4		0.8		0.2		6.5		9.9
Operating expenses		740.3		217.7		52.4		99.1		1,109.5
Operating income (loss)	\$	130.2	\$	84.4	\$	3.2	\$	(99.1)	\$	118.7

	For the Nine-Month Period Ended September 30, 2017									
	Hom	e Health		Hospice		Personal Care		Other		Total
Net service revenue	\$	802.0	\$	269.8	\$	41.5	\$	—	\$	1,113.3
Cost of service, excluding depreciation and amortization		496.1		136.7		31.3		—		664.1
General and administrative expenses		207.8		56.2		9.2		85.4		358.6
Depreciation and amortization		2.7		0.7		0.1		9.6		13.1
Securities Class Action Lawsuit settlement, net						_		28.7		28.7
Operating expenses		706.6		193.6		40.6		123.7		1,064.5
Operating income (loss)	\$	95.4	\$	76.2	\$	0.9	\$	(123.7)	\$	48.8

# 7. SHARE REPURCHASE

On June 4, 2018, we purchased 2,418,304 of our common shares from affiliates of KKR Credit Advisors (US) LLC ("KKR"), representing one-half of KKR's holdings in the Company and 7.1% of the aggregate outstanding shares of the Company's common stock for a total purchase price of \$181.4 million including related direct costs. The Company repurchased the shares at \$73.96 which represents 96% of the closing stock price of the Company's common stock on June 4, 2018.

# 8. SUBSEQUENT EVENTS

On October 1, 2018, we acquired certain personal care operations from Bring Care Home in Massachusetts for a purchase price of \$5.8 million.

On October 9, 2018, we signed a definitive agreement to acquire Compassionate Care Hospice, a national hospice care provider headquartered in New Jersey, for a purchase price of \$340 million, which is inclusive of \$50 million in payments related to a tax asset and working capital. The acquisition is expected to close by February 1, 2019.

20

#### ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis provides information we believe is relevant to an assessment and understanding of our results of operations and financial condition for the three and nine-month periods ended September 30, 2018. This discussion should be read in conjunction with the condensed consolidated financial statements and notes thereto included herein, and the consolidated financial statements and notes thereto included herein, and the consolidated financial statements and notes and the related Management's Discussion and Analysis of Financial Condition and Results of Operations in our Annual Report on Form 10-K for the year ended December 31, 2017 filed with the Securities and Exchange Commission ("SEC") on February 28, 2018 (the "Form 10-K"), which are incorporated herein by this reference. Historical results that appear in the condensed consolidated financial statements should not be interpreted as being indicative of future operations.

Unless otherwise provided, "Amedisys," "we," "our," and the "Company" refer to Amedisys, Inc. and our consolidated subsidiaries.

#### Overview

We are a provider of high-quality in-home healthcare and related services to the chronic, co-morbid, aging American population, with approximately 73% and 74% of our revenue derived from Medicare for the three and nine-month periods ended September 30, 2018, respectively, and approximately 76% of our revenue derived from Medicare for the three and nine-month periods ended September 30, 2017.

Our operations involve servicing patients through our three reportable business segments: home health, hospice and personal care. Our home health segment delivers a wide range of services in the homes of individuals who may be recovering from an illness, injury or surgery. Our hospice segment provides care that is designed to provide comfort and support for those who are facing a terminal illness. Our personal care segment provides patients with assistance with the essential activities of daily living. As of September 30, 2018, we owned and operated 323 Medicare-certified home health care centers, 83 Medicare-certified hospice care centers and 15 personal-care care centers in 34 states within the United States and the District of Columbia.

# **Owned and Operated Care Centers (Includes Unconsolidated Joint Ventures)**

	Home Health	Hospice	Personal Care
As of December 31, 2017	323	83	15
Acquisitions/Startups	1		1
Closed/Consolidated	(1)		(1)
As of September 30, 2018	323	83	15

# **Recent Developments**

# Governmental Inquiries and Investigations and Other Litigation

See Note 5 – Commitments and Contingencies to our condensed consolidated financial statements for additional information regarding our corporate integrity agreement and for a discussion of and updates regarding other legal proceedings and investigations we are involved in. No assurances can be given as to the timing or outcome of these items.

#### Payment

On August 6, 2018, the Centers for Medicare and Medicaid Services ("CMS") issued a final rule to update hospice payment rates and the wage index for fiscal year 2019. CMS estimates hospices serving Medicare beneficiaries would see an estimated 1.8% increase in payments. This increase is the result of a 2.9% market basket adjustment less a 0.8% productivity adjustment, less 0.3% as required under the Patient Protection and Affordable Health Care Act and the Health Care and Education Reconciliation Act (collectively, "PPACA"). CMS also increased the aggregate cap amount by 1.8% to \$29,205.44. We expect our impact of the 2019 final rule to be in line with that of the hospice industry.

In July 2018, CMS issued proposed payment changes for Medicare home health providers for 2019 and 2020. For 2019, CMS estimates that the net impact of the payment provisions of the proposed changes will result in an increase of 2.1% in reimbursement to home health providers. This increase is the result of a 2.8% market basket increase less a 0.7% productivity adjustment. Additionally, the proposed rule includes changes to the home health prospective payment system ("HHPPS") case-mix adjustment methodology through the use of a new Patient-Driven Grouping Model ("PDGM") for home health payments. This change is proposed to be implemented January 1, 2020 and also includes a change in the unit of payment from 60-day episodes of care to

30-day periods of care and eliminates the use of therapy visits in the determination of payments. While the proposed rule is to be implemented in a budget neutral manner to the industry, the ultimate impact will vary by provider based on factors including patient mix and admission source. Additionally, in arriving at the calculation of a rate that is budget neutral, CMS has made assumptions about behavioral changes. We are currently evaluating the proposed rule's impact on our home health operations.

#### Acquisitions

On March 1, 2018, we acquired the assets of Christian Care at Home which provides home health services to the state of Kentucky.

On May 1, 2018, we acquired the assets of East Tennessee Personal Care Services which provides personal care services to the state of Tennessee.

On October 1, 2018, we acquired certain personal care operations from Bring Care Home in Massachusetts.

On October 9, 2018, we signed a definitive agreement to acquire Compassionate Care Hospice, a national hospice care provider headquartered in New Jersey. The acquisition is expected to close by February 1, 2019.

#### **Results of Operations**

#### Three-Month Period Ended September 30, 2018 Compared to the Three-Month Period Ended September 30, 2017

#### Consolidated

The following table summarizes our consolidated results of operations (amounts in millions):

	 For the Month Ended Sep	Period	ls
	2018		2017
Net service revenue	\$ 417.3	\$	373.7
Gross margin, excluding depreciation and amortization	167.6		146.4
% of revenue	40.2 %		39.2 %
Other operating expenses	127.7		123.0
% of revenue	30.6 %		32.9 %
Operating income	39.9		23.4
Total other income, net	 1.5		0.7
Income tax expense	(9.8)		(9.4)
Effective income tax rate	23.7 %		39.0 %
Net income	 31.6		14.7
Net income attributable to noncontrolling interests	(0.2)		(0.1)
Net income attributable to Amedisys, Inc.	\$ 31.4	\$	14.6

Overall, our operating income increased \$16 million on a revenue increase of \$44 million driven by growth within all three of our segments. Our 2017 operating results were negatively impacted \$9 million; these impacts included a \$7 million reduction in revenue as a result of the Florida ZPIC audit (see Note 5 - Commitments and Contingencies to our condensed consolidated financial statements for additional information regarding this matter) and approximately \$2 million in severance costs related to our home health closures and restructuring plan. Excluding the 2017 revenue reduction, our gross margin as a percentage of revenue remained flat despite a net reduction of \$1 million in net service revenue and gross margin due to the impact of the 2018 changes in reimbursement and planned wage increases that became effective during the three-month period ended September 30, 2018. Our home health segment continues to see improvement in clinician productivity which positively impacted our gross margin. We also continue to remain focused on costs as our other operating expenses as a percentage of revenue declined significantly from prior year.

Our income tax expense as a percentage of our income before income taxes decreased due to the enactment of H.R. 1 (Tax Cuts and Jobs Act) on December 22, 2017.

# Home Health Segment

The following table summarizes our home health segment results of operations:

	For th Montl Ended Se	1 Peri	ods
	 2018		2017
Financial Information (in millions):			
Medicare	\$ 208.0	\$	191.4
Non-Medicare	86.9		72.7
Net service revenue	294.9	•	264.1
Cost of service	181.8		168.3
Gross margin	113.1	·	95.8
Other operating expenses	69.6		71.7
Operating income	\$ 43.5	\$	24.1
Same Store Growth (1):			
Medicare revenue	10%		(7%)
Non-Medicare revenue	20%		18%
Total admissions	4%		1%
Total volume (2)	6%		5%
Total Episodic admissions (3)	3%		1%
Total Episodic volume (4)	5%		3%
Key Statistical Data - Total (5):			
Medicare:			
Admissions	46,371		46,823
Recertifications	 28,551		26,996
Total volume	74,922		73,819
Completed episodes	73,114		71,454
Visits	1,301,090		1,259,156
Average revenue per completed episode (6)	\$ 2,855	\$	2,820
Visits per completed episode (7)	17.6		17.4
Non-Medicare:			
Admissions	29,325		26,686
Recertifications	 14,539		12,263
Total volume	43,864		38,949
Visits	698,181		592,742
Total (5):			
Visiting Clinician Cost per Visit	\$ 82.78	\$	82.53
Clinical Manager Cost per Visit	8.18		8.30
Total Cost per Visit	\$ 90.96	\$	90.83
Visits	1,999,271		1,851,898

(1) Same store information represents the percent increase (decrease) in our Medicare, Non-Medicare, Total and Episodic revenue, admissions or volume for the period as a percent of the Medicare, Non-Medicare, Total and Episodic revenue, admissions or volume of the prior period.

(2) Total volume includes all admissions and recertifications.

(3) Total Episodic admissions includes admissions for Medicare and Non-Medicare payors that bill on a 60-day episode of care basis.

(4) Total Episodic volume includes admissions and recertifications for Medicare and Non-Medicare payors that bill on a 60-day episode of care basis.

- (5) Total includes acquisitions.
- (6) Average Medicare revenue per completed episode is the average Medicare revenue earned for each Medicare completed episode of care.
- (7) Medicare visits per completed episode are the home health Medicare visits on completed episodes divided by the home health Medicare episodes completed during the period.

#### **Operating Results**

Overall, our operating income increased \$19 million on a \$31 million increase in net service revenue. The \$19 million increase includes the \$7 million reduction in revenue related to the Florida ZPIC audit in 2017 (see Note 5 - Commitments and Contingencies to our condensed consolidated financial statements for additional information regarding this matter). Our growth in volumes and increases in clinician productivity positively impacted our gross margin as a percentage of revenue which increased despite the 2018 change in reimbursement and planned wage increases that became effective during the three-month period ended September 30, 2018. The impact of the 2018 change in reimbursement was a reduction in net service revenue and gross margin of approximately \$2 million.

#### Net Service Revenue

Our revenue increased \$31 million on a 6% increase in total volumes which is inclusive of a 5% increase in episodic volumes. The volume growth was driven by a 4% increase in admissions and a 120 basis point increase in our Medicare recertification rate. In addition to the increase in volumes, our revenue per episode is up approximately \$35 per episode as a result of an increase in the acuity level of our patients which offset the 70 basis point reimbursement reduction effective January 1, 2018. The increase in revenue for the three-month period ended September 30, 2018 was also positively impacted by a \$9 million reduction in our revenue price concessions which is inclusive of the \$7 million reduction in revenue related to the Florida ZPIC audit during the three-month period ended September 30, 2017.

#### Cost of Service, Excluding Depreciation and Amortization

Our cost per visit consists of costs associated with direct clinician care in the homes of our patients as well as the cost of clinical managers who monitor the overall delivery of care. Our cost of service increased 8% on an 8% increase in total visits. Our increase in total visits was driven by growth in volumes as well as an increase in visits per completed episode which is the result of an increase in the acuity level of our patients. Our cost per visit remained relatively flat as an increase in clinician productivity offset planned wage increases.

#### Other Operating Expenses

Other operating expenses decreased approximately \$2 million primarily due to a decrease in salaries and benefits expense as 2017 operating expenses included approximately \$2 million in severance costs related to our home health restructuring plan.

# **Hospice Segment**

The following table summarizes our hospice segment results of operations:

	 For the Three- Month Periods Ended September 30,				
	 2018	_	2017		
Financial Information (in millions):					
Medicare	\$ 98.0	\$	91.4		
Non-Medicare	5.4		4.5		
Net service revenue	 103.4		95.9		
Cost of service	53.4		48.4		
Gross margin	50.0		47.5		
Other operating expenses	21.7		19.2		
Operating income	\$ 28.3	\$	28.3		
Same Store Growth (1):					
Medicare revenue	7%		17%		
Non-Medicare revenue	19%		11%		
Hospice admissions	8%		7%		
Average daily census	11%		14%		
Key Statistical Data - Total (2):					
Hospice admissions	6,765		6,257		
Average daily census	7,768		7,026		
Revenue per day, net	\$ 144.71	\$	148.43		
Cost of service per day	\$ 74.72	\$	74.96		
Average discharge length of stay	101		95		

- (1) Same store information represents the percent increase (decrease) in our Medicare and Non-Medicare revenue, Hospice admissions or average daily census for the period as a percent of the Medicare and Non-Medicare revenue, Hospice admissions or average daily census of the prior period.
- (2) Total includes acquisitions.

# **Operating Results**

Overall, our operating income remained flat on an \$8 million increase in net service revenue. We experienced a decrease in gross margin as a percentage of revenue primarily due to planned wage increases that became effective during the three-month period ended September 30, 2018 and an increase in our estimated price concessions and amounts due back to Medicare for hospice caps.

#### Net Service Revenue

Our hospice revenue increased \$8 million on an 11% increase in our average daily census and a 1% increase in reimbursement effective for services provided from October 1, 2017. We experienced a \$3 million increase in our revenue price concessions and cap which negatively impacted our revenue for the three-month period ended September 30, 2018.

# Cost of Service, Excluding Depreciation and Amortization

Our hospice cost of service increased \$5 million (10%) as the result of an 11% increase in average daily census. Our cost of service per day decreased despite an increase in salary cost per day as a result of planned wage increases that became effective during the three-month period ended September 30, 2018.

#### Other Operating Expenses

Other operating expenses increased \$3 million due to increases in other care center related expenses, primarily salaries and benefits expense and travel and training expense.

# Personal Care Segment

The following table summarizes our personal care segment results of operations:

	 For the Month Ended Sej	Perio	ds
	2018		2017
Financial Information (in millions):			
Medicare	\$ 	\$	_
Non-Medicare	19.0		13.7
Net service revenue	 19.0		13.7
Cost of service	14.5		10.6
Gross margin	 4.5		3.1
Other operating expenses	3.1		3.1
Operating income	\$ 1.4	\$	_
Key Statistical Data:			
Billable hours	810,427		616,036
Clients served	12,380		7,978
Shifts	357,382		281,904
Revenue per hour	\$ 23.40	\$	22.21
Revenue per shift	\$ 53.06	\$	48.54
Hours per shift	2.3		2.2

On October 1, 2017, we acquired the assets of Intercity Home Care, which owned and operated four personal-care care centers, three of which were subsequently consolidated with our existing personal-care care centers. On May 1, 2018, we acquired the assets of East Tennessee Personal Care Services, which owned and operated one personal-care care center. Acquisitions are included in our consolidated financial statements from their respective acquisition dates. As a result, our personal care operating results for 2018 and 2017 are not fully comparable.

Operating income related to our personal care segment increased \$1 million on a \$5 million increase in net service revenue. The increase in net service revenue includes revenues associated with our acquisitions. Gross margin increased despite the incurrence of additional costs associated with our acquisitions and the newly enacted Employer Medical Assistance Contribution ("EMAC") program that became effective in the state of Massachusetts on January 1, 2018. Other operating expenses remained flat on a 39% increase in net service revenue.

# **Corporate**

The following table summarizes our corporate results of operations:

	For the Month Ended Seg	ds	
	2018		2017
Financial Information (in millions):			
Other operating expenses	\$ 31.3	\$	25.9
Depreciation and amortization	2.0		3.1
Total operating expenses	\$ 33.3	\$	29.0

Corporate expenses consist of costs relating to our executive management and corporate and administrative support functions, primarily information services, accounting, finance, billing and collections, legal, compliance, risk management, procurement, marketing, clinical administration, training, human resources and administration.

During the three-month period ended September 30, 2018, corporate operating expenses increased \$4 million due to increases in salaries and benefits expense, professional fees and travel and training expense which were offset by a decrease in depreciation and amortization.

# Nine-Month Period Ended September 30, 2018 Compared to the Nine-Month Period Ended September 30, 2017

# Consolidated

The following table summarizes our consolidated results of operations (amounts in millions):

	For the Nine- Month Periods Ended September 30,				
	2018		2017		
Net service revenue	\$ 1,228.2	\$	1,113.3		
Gross margin, excluding depreciation and amortization	497.6		449.2		
% of revenue	40.5 %		40.3 %		
Other operating expenses	378.9		371.7		
% of revenue	30.9 %		33.4 %		
Securities Class Action Lawsuit settlement, net			28.7		
Operating income	118.7		48.8		
Total other income (expense), net	3.7		2.9		
Income tax expense	(30.0)		(17.3)		
Effective income tax rate	24.5 %		33.5 %		
Net income	92.4		34.4		
Net income attributable to noncontrolling interests	(0.5)		(0.2)		
Net income attributable to Amedisys, Inc.	\$ 91.9	\$	34.1		

Overall, our operating income increased \$70 million on a revenue increase of \$115 million. Our 2017 operating results were negatively impacted \$38 million; these impacts included our Securities Class Action Lawsuit settlement, a \$7 million reduction in revenue as a result of the Florida ZPIC audit and approximately \$2 million in severance costs related to our home health closures and restructuring plan. Excluding these 2017 impacts, operating income increased \$32 million, driven by the improved performance of all three of our segments. In addition, our gross margin as a percentage of revenue was relatively flat despite a net reduction of \$2 million in net service revenue and gross margin due to the impact of the 2018 changes in reimbursement and planned wage increases during the three-month period ended September 30, 2018. Our improvement in gross margin was driven by continued growth in our home health and hospice segments and increases in clinical productivity in our home health segment. We continue to maintain cost discipline as our other operating expenses increased only 2% on a 10% increase in net service revenue.

Our income tax expense as a percentage of our income before income taxes decreased due to the enactment of H.R. 1 (Tax Cuts and Jobs Act) on December 22, 2017.

# Home Health Segment

The following table summarizes our home health segment results of operations:

	For t Montl Ended Se	ods	
	 2018	-	2017
Financial Information (in millions):			
Medicare	\$ 619.3	\$	588.4
Non-Medicare	251.2		213.6
Net service revenue	 870.5		802.0
Cost of service	532.7		496.1
Gross margin	337.8		305.9
Other operating expenses	207.6		210.5
Operating income	\$ 130.2	\$	95.4
Same Store Growth (1):			
Medicare revenue	7%		(5%
Non-Medicare revenue	18%		14%
Total admissions	4%		1%
Total volume (2)	7%		3%
Total Episodic admissions (3)	3%		1%
Total Episodic volume (4)	6%		3%
Key Statistical Data - Total (5):			
Medicare:			
Admissions	142,884		143,711
Recertifications	 84,218		78,878
Total volume	227,102		222,589
Completed episodes	220,726		217,190
Visits	3,933,290		3,794,001
Average revenue per completed episode (6)	\$ 2,841	\$	2,811
Visits per completed episode (7)	17.5		17.3
Non-Medicare:			
Admissions	88,485		80,244
Recertifications	 40,862		33,949
Total volume	129,347		114,193
Visits	2,049,662		1,727,618
Total (5):			
Visiting Clinician Cost per Visit	\$ 81.06	\$	81.41
Clinical Manager Cost per Visit	7.98		8.42
Total Cost per Visit	\$ 89.04	\$	89.83
Visits	5,982,952		5,521,619

(1) Same store information represents the percent increase (decrease) in our Medicare, Non-Medicare, Total and Episodic revenue, admissions or volume for the period as a percent of the Medicare, Non-Medicare, Total and Episodic revenue, admissions or volume of the prior period.

(2) Total volume includes all admissions and recertifications.

(3) Total Episodic admissions includes admissions for Medicare and Non-Medicare payors that bill on a 60-day episode of care basis.

- (4) Total Episodic volume includes admissions and recertifications for Medicare and Non-Medicare payors that bill on a 60-day episode of care basis.
- (5) Total includes acquisitions.
- (6) Average Medicare revenue per completed episode is the average Medicare revenue earned for each Medicare completed episode of care.
- (7) Medicare visits per completed episode are the home health Medicare visits on completed episodes divided by the home health Medicare episodes completed during the period.

# **Operating Results**

Overall, our operating income increased \$35 million on a \$69 million increase in net service revenue. The \$35 million increase includes the \$7 million reduction in revenue related to the Florida ZPIC audit in 2017. Our growth in volumes and increases in clinician productivity positively impacted our gross margin as a percentage of revenue, which increased despite the 2018 changes in reimbursement and planned wage increases that became effective during the three-month period ended September 30, 2018. The impact of the 2018 changes in reimbursement was a reduction in net service revenue and gross margin of approximately \$5 million.

# Net Service Revenue

Our revenue increased \$69 million on a 7% increase in total volumes which is inclusive of a 6% increase in episodic volumes. The volume growth was driven by a 4% increase in admissions and a 190 basis point increase in our Medicare recertification rate. In addition to the increase in volumes, our revenue per episode is up \$30 per episode as a result of an increase in the acuity level of our patients which offset the 70 basis point reimbursement reduction effective January 1, 2018. The increase in revenue for the nine-month period ended September 30, 2018 was also positively impacted by a \$5 million decrease in our revenue price concessions. Excluding the \$7 million reduction in revenue related to the Florida ZPIC audit in 2017, our revenue price concessions increased approximately \$2 million during the nine-month period ended September 30, 2018.

# Cost of Service, Excluding Depreciation and Amortization

Our cost of service increased 7% on an 8% increase in total visits. Our increase in total visits was driven by growth in volumes as well as an increase in visits per completed episode which is the result of an increase in the acuity level of our patients. Our cost per visit decreased 1% as an increase in clinician productivity offset planned wage increases.

# Other Operating Expenses

Other operating expenses decreased approximately \$3 million on a 9% increase in net service revenue primarily due to a decrease in salaries and benefits expense as 2017 operating expenses included approximately \$2 million in severance costs related to our home health restructuring plan. Additionally, we experienced decreases in professional fees and telecommunications expense which were offset by increases in information technology expense and travel and training expense.

#### Hospice Segment

The following table summarizes our hospice segment results of operations:

	 For the Nine- Month Periods Ended September 30,			
	 2018		2017	
Financial Information (in millions):				
Medicare	\$ 286.7	\$	257.9	
Non-Medicare	15.4		11.9	
Net service revenue	302.1		269.8	
Cost of service	155.2		136.7	
Gross margin	146.9		133.1	
Other operating expenses	62.5		56.9	
Operating income	\$ 84.4	\$	76.2	
Same Store Growth (1):				
Medicare revenue	11%		18%	
Non-Medicare revenue	28%		(1%)	
Hospice admissions	7%		13%	
Average daily census	12%		15%	
Key Statistical Data - Total (2):				
Hospice admissions	20,444		19,010	
Average daily census	7,514		6,705	
Revenue per day, net	\$ 147.28	\$	147.43	
Cost of service per day	\$ 75.65	\$	74.78	
Average discharge length of stay	98		92	

- Same store information represents the percent increase (decrease) in our Medicare and Non-Medicare revenue, Hospice admissions or average daily census for the period as a percent of the Medicare and Non-Medicare revenue, Hospice admissions or average daily census of the prior period.
- (2) Total includes acquisitions.

#### **Operating Results**

Overall, our operating income increased \$8 million on a \$32 million increase in net service revenue. The 12% increase in net service revenue was partially offset by a lower gross margin as a percentage of revenue primarily related to planned wage increases that became effective during the three-month period ended September 30, 2018, an increase in revenue price concessions and amounts due back to Medicare for hospice caps and an increase in other operating expenses.

#### Net Service Revenue

Our hospice revenue increased \$32 million on a 12% increase in our average daily census and a 1% increase in reimbursement effective for services provided from October 1, 2017. We experienced a \$3 million increase in our revenue price concessions and cap which negatively impacted our revenue for the nine-month period ended September 30, 2018.

#### Cost of Service, Excluding Depreciation and Amortization

Our hospice cost of service increased \$18 million (14%) as the result of a 12% increase in average daily census. Our cost of service per day increased 1% primarily due to an increase in salary cost per day as a result of planned wage increases that became effective during the three-month period ended September 30, 2018.

# Other Operating Expenses

Other operating expenses increased \$6 million on a 12% increase in net service revenue. The increase was related to other care center related expenses, primarily salaries and benefits expense, advertising expense, information technology expense, professional fees and travel and training expense as a result of the addition of resources to support census growth.

## **Personal Care Segment**

The following table summarizes our personal care segment results of operations:

	 For the Nine- Month Periods Ended September 30,			
	 2018		2017	
Financial Information (in millions):				
Medicare	\$ —	\$		
Non-Medicare	55.6		41.5	
Net service revenue	55.6		41.5	
Cost of service	42.7		31.3	
Gross margin	12.9		10.2	
Other operating expenses	9.7		9.3	
Operating income	\$ 3.2	\$	0.9	
Key Statistical Data:				
Billable hours	2,357,608		1,822,653	
Clients served	15,731		10,980	
Shifts	1,062,422		830,151	
Revenue per hour	\$ 23.57	\$	22.76	
Revenue per shift	\$ 52.30	\$	49.98	
Hours per shift	2.2		2.2	

Operating income related to our personal care segment increased approximately \$2 million on a \$14 million increase in net service revenue. The increase in net service revenue includes revenues associated with the acquisitions of Intercity Home Care and East Tennessee Personal Care Services, both of which occurred after September 30, 2017. The segment experienced a decrease in gross margin as a percentage of revenue related to additional costs associated with these acquisitions and the EMAC program that became effective in the state of Massachusetts on January 1, 2018. Other operating expenses remained relatively flat on a 34% increase in net service revenue.

# **Corporate**

The following table summarizes our corporate results of operations:

	For the Nine- Month Periods Ended September 30,				
	2018			2017	
Financial Information (in millions):					
Other operating expenses	\$	92.6	\$	85.4	
Depreciation and amortization		6.5		9.6	
Total operating expenses before Securities Class Action Lawsuit settlement, net		99.1		95.0	
Securities Class Action Lawsuit settlement, net				28.7	
Total operating expenses	\$	99.1	\$	123.7	

Excluding the Securities Class Action Lawsuit settlement accrual during the nine-month period ended September 30, 2017, corporate operating expenses increased 4% on a 10% increase in net service revenue. Increases in salaries and benefits expense, travel and training expense and a loss on disposal of assets were offset by decreases in personnel costs, telecommunications expense, professional fees and depreciation and amortization.

#### Liquidity and Capital Resources

#### Cash Flows

The following table summarizes our cash flows for the periods indicated (amounts in millions):

		For the Nine- Month Periods Ended September 30,		
	2	018		2017
Cash provided by operating activities	\$	159.4	\$	73.5
Cash used in investing activities		(12.6)		(32.9)
Cash used in financing activities		(219.2)		(4.7)
Net (decrease) increase in cash and cash equivalents		(72.4)		35.9
Cash and cash equivalents at beginning of period		86.4		30.2
Cash and cash equivalents at end of period	\$	14.0	\$	66.1

Cash provided by operating activities increased \$85.9 million during the nine-month period ended September 30, 2018 compared to the nine-month period ended September 30, 2017 primarily due to an increase in our net income and cash collections as compared to 2017. In addition, during 2017, operating cash flows were negatively impacted by \$28.7 million related to the Securities Class Action Lawsuit settlement (see Note 5 - Commitments and Contingencies to our condensed consolidated financial statements). For additional information regarding our operating performance and our days revenue outstanding, see "Results of Operations" and "Outstanding Patient Accounts Receivable," respectively.

Cash used in investing activities decreased \$20.3 million during the nine-month period ended September 30, 2018 compared to the nine-month period ended September 30, 2017 primarily due to a decrease in our acquisition activity (\$20.1 million) and a decrease in capital expenditures (\$3.4 million) offset by an increase in equity method investments (\$3.0 million).

Cash used in financing activities increased \$214.5 million during the nine-month period ended September 30, 2018 compared to the nine-month period ended September 30, 2017 primarily due to our repurchase of company stock and the repayment of borrowings under our Term Loan and Revolving Credit Facility offset by borrowings under our new Credit Agreement.

#### Liquidity

Typically, our principal source of liquidity is the collection of our patient accounts receivable, primarily through the Medicare program. In addition to our collection of patient accounts receivable, from time to time, we can and do obtain additional sources of liquidity by the incurrence of additional indebtedness.

During the nine-month period ended September 30, 2018, we spent \$5.7 million in capital expenditures as compared to \$9.1 million during the ninemonth period ended September 30, 2017. Our capital expenditures for 2018 are expected to be approximately \$7.0 million to \$9.0 million.

As of September 30, 2018, we had \$14.0 million in cash and cash equivalents and \$458.1 million in availability under our \$550.0 million Revolving Credit Facility.

Based on our operating forecasts and our debt service requirements, we believe we will have sufficient liquidity to fund our operations, capital requirements and debt service requirements.

#### **Outstanding Patient Accounts Receivable**

Our patient accounts receivable decreased \$6.2 million from December 31, 2017 to September 30, 2018. Our cash collection as a percentage of revenue was 104% and 102% for the nine-month periods ended September 30, 2018 and 2017, respectively. Our days revenue outstanding at September 30, 2018 was 40.6 days which is a decrease of 3.4 days from December 31, 2017.

Our patient accounts receivable includes unbilled receivables and are aged based upon our initial service date. We monitor unbilled receivables on a care center by care center basis to ensure that all efforts are made to bill claims within timely filing deadlines. Our unbilled patient accounts receivable can be impacted by acquisition activity, probe edits or regulatory changes which result in additional information or procedures needed prior to billing. The timely filing deadline for Medicare is one year from the date the episode was completed and varies by state for Medicaid-reimbursable services and among insurance companies and other private payors.

The following schedules detail our patient accounts receivable, by payor class, aged based upon initial date of service (amounts in millions, except days revenue outstanding):

	0-90	91-180	181-365	0	ver 365	Total
At September 30, 2018:						
Medicare patient accounts receivable	\$ 93.3	\$ 9.2	\$ 2.6	\$	1.8	\$ 106.9
Other patient accounts receivable:						
Medicaid	14.2	2.4	1.2		(1.4)	16.4
Private	56.7	 8.5	 4.3		2.2	 71.7
Total	\$ 70.9	\$ 10.9	\$ 5.5	\$	0.8	\$ 88.1
Total patient accounts receivable						\$ 195.0
Days revenue outstanding (1)						40.6
	 0-90	 91-180	 181-365	0	ver 365	Total
At December 31, 2017:						
Medicare patient accounts receivable	\$ 95.9	\$ 16.1	\$ 6.6	\$	0.6	\$ 119.2
Other patient accounts receivable:						
Medicaid	13.8	3.2	1.3		(1.1)	17.2
Private	 51.0	7.5	 4.1		2.2	 64.8
Total	\$ 64.8	\$ 10.7	\$ 5.4	\$	1.1	\$ 82.0
Total patient accounts receivable						\$ 201.2
Days revenue outstanding (1)						 44.0

(1) Our calculation of days revenue outstanding is derived by dividing our ending patient accounts receivable at September 30, 2018 and December 31, 2017 by our average daily patient revenue for the three-month periods ended September 30, 2018 and December 31, 2017, respectively.

#### Indebtedness

#### Credit Agreement

On June 29, 2018, we entered into our Amended and Restated Credit Agreement ("Credit Agreement") that provides for a senior secured revolving credit facility in an initial aggregate principal amount of up to \$550.0 million (the "Revolving Credit Facility"). The funds available under the Revolving Credit Facility were used to pay off our existing indebtedness under our prior credit agreement, dated as of August 28, 2015 (the "Prior Credit Agreement"), with a principal balance of \$127.5 million.

The final maturity of the Revolving Credit Facility is June 29, 2023 and there is no mandatory amortization on the outstanding principal balances which are payable in full upon maturity. The Revolving Credit Facility may be used to provide ongoing working capital and for general corporate purposes of the Company and our subsidiaries, including permitted acquisitions, as defined in the Credit Agreement.

Our weighted average interest rate for our \$100.0 million Term Loan, under our Prior Credit Agreement, was 3.2% and 3.0% for the three and ninemonth periods ended September 30, 2017, respectively. Our weighted average interest rate for borrowings under our \$550.0 million Revolving Credit Facility was 3.6% and 3.8% for the three and nine-month periods ended September 30, 2018, respectively. As of September 30, 2018, our consolidated leverage ratio was 0.4, our consolidated interest coverage ratio was 57.6 and we are in compliance with our covenants under the Credit Agreement.

As of September 30, 2018, our availability under our \$550.0 million Revolving Credit Facility was \$458.1 million as we have \$57.5 million outstanding in borrowings and \$34.4 million outstanding in letters of credit.

See Note 4 - Long Term Obligations to our condensed consolidated financial statements for additional details on our outstanding long-term obligations.

#### Share Repurchase

On June 4, 2018, we purchased 2,418,304 of our common shares from affiliates of KKR Credit Advisors (US) LLC ("KKR"), representing one-half of KKR's holdings in the Company and 7.1% of the aggregate outstanding shares of the Company's common stock for a total purchase price of \$181.4 million including related direct costs. The Company repurchased the shares at \$73.96 which represents 96% of the closing stock price of the Company's common stock on June 4, 2018.

#### Inflation

We do not believe inflation has significantly impacted our results of operations.

#### **Critical Accounting Estimates**

See Part II, Item 7 – Critical Accounting Estimates and our consolidated financial statements and related notes in Part II, Item 8 of our 2017 Annual Report on Form 10-K, for accounting policies and related estimates we believe are the most critical to understanding our condensed consolidated financial statements, financial condition and results of operations and which require complex management judgment and assumptions, or involve uncertainties. These critical accounting estimates include revenue recognition, patient accounts receivable, insurance, goodwill and other intangible assets, and income taxes. There have not been any changes to our significant accounting policies or their application since we filed our 2017 Annual Report on Form 10-K except for the changes related to the implementation of Accounting Standards Updates 2014-09 and 2015-14 as disclosed in Note 2 - Summary of Significant Accounting Policies to our condensed consolidated financial statements.

#### ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to market risk from fluctuations in interest rates. Our Revolving Credit Facility carries a floating interest rate which is tied to the Eurodollar rate (*i.e.* LIBOR) and the Prime Rate and therefore, our condensed consolidated statements of operations and our condensed consolidated statements of cash flows are exposed to changes in interest rates. As of September 30, 2018, the total amount of outstanding debt subject to interest rate fluctuations was \$57.5 million. A 1.0% interest rate change would cause interest expense to change by approximately \$0.6 million annually.

#### **ITEM 4. CONTROLS AND PROCEDURES**

#### **Evaluation of Disclosure Controls and Procedures**

We have established disclosure controls and procedures which are designed to provide reasonable assurance of achieving their objectives and to ensure that information required to be disclosed in our reports filed under the Securities Exchange Act of 1934 as amended (the "Exchange Act") is recorded, processed, summarized, disclosed and reported within the time periods specified in the SEC's rules and forms. This information is also accumulated and communicated to our management and Board of Directors to allow timely decisions regarding required disclosure.

In connection with the preparation of this Quarterly Report on Form 10-Q, as of September 30, 2018, under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, we conducted an evaluation of the effectiveness of our disclosure controls and procedures, as such term is defined under Rules 13a-15(e) and 15d-15(e) promulgated under the Exchange Act.

Based on this evaluation, our principal executive officer and principal financial officer concluded that our disclosure controls and procedures were effective at a reasonable assurance level as of September 30, 2018, the end of the period covered by this Quarterly Report.

#### **Changes in Internal Controls**

There have been no changes in our internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f)) that have occurred during the quarter ended September 30, 2018, that have materially impacted, or are reasonably likely to materially impact, our internal control over financial reporting.

#### Inherent Limitations on Effectiveness of Controls

Our management, including our principal executive officer and principal financial officer, does not expect that our disclosure controls or our internal controls over financial reporting will prevent or detect all errors and all fraud. A control system, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control system's objectives will be met. The design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Further, because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that misstatements due to error or fraud will not occur or that all control issues and instances of fraud, if any, have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty and that breakdowns can occur because of simple error or mistake. Controls can also be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the controls. The design of any system of controls is based in part on certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Projections of any evaluation of controls' effectiveness to future periods are subject to risks. Over time, controls may become inadequate because of changes in conditions or deterioration in the degree of compliance with policies and procedures. Our disclosure controls and procedures are designed to provide reasonable assurance of achieving their objectives and, based on an evaluation of our controls and procedures, our principal executive officer and our principal financial officer concluded our disclosure controls and procedures were effective at a reasonable assurance level as of September 30, 2018, the end of the period covered by this Quarterly Repor

#### PART II. OTHER INFORMATION

### **ITEM 1. LEGAL PROCEEDINGS**

See Note 5 - Commitments and Contingencies to the condensed consolidated financial statements for information concerning our legal proceedings.

#### **ITEM 1A. RISK FACTORS**

In addition to other information set forth in this Quarterly Report on Form 10-Q, you should carefully consider the risk factors included in Part I, Item 1A. – Risk Factors of our Annual Report on Form 10-K. These risk factors could materially impact our business, financial condition and/or operating results. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially adversely impact our business, financial condition and/or operating results.

### ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

The following table provides the information with respect to purchases made by us of shares of our common stock during each of the months during the three-month period ended September 30, 2018:

Period	(a) Total Number of Shares (or Units) Purchased		Paid pe	erage Price er Share (or Unit)	(c) Total Number of Shares (or Units) Purchased as Part of Publicly Announced Plans or Programs	(d) Maximum Number (or Approximate Dollar Value) of Shares (or Units) That May Yet Be Purchased Under the Plans or Programs	
July 1, 2018 to July 31, 2018	24,094	5	6	94.97	_	\$	—
August 1, 2018 to August 31, 2018	—				—		_
September 1, 2018 to September 30, 2018	2,457			122.22	—		_
	26,551 (1	) \$	6	97.50		\$	

(1) Includes shares of common stock surrendered to us by certain employees to satisfy tax withholding obligations in connection with the vesting of non-vested stock previously awarded to such employees under our 2008 Omnibus Incentive Compensation Plan.

#### ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

### **ITEM 4. MINE SAFETY DISCLOSURES**

Not applicable.

### **ITEM 5. OTHER INFORMATION**

None.



### **ITEM 6. EXHIBITS**

The exhibits marked with the cross symbol ( $\dagger$ ) are filed and the exhibits marked with a double cross ( $\dagger$  $\dagger$ ) are furnished with this Form 10-Q. Any exhibits marked with the asterisk symbol ( $\ast$ ) are management contracts or compensatory plans or arrangements filed pursuant to Item 601(b)(10)(iii) of Regulation S-K.

Exhibit Number	Document Description	Report or Registration Statement	SEC File or Registration Number	Exhibit or Other Reference
†2.1	Stock Purchase Agreement, dated as of October 9, 2018, by and among Milton Heching, the Heching 2012 Exempt Irrevocable Trust, Amedisys Hospice, L.L.C., Compassionate Care Hospice Group, Inc., and solely for purposes of Sections 3.4, 4.3(a), 4.15 and Article VIII thereof, Amedisys, Inc. (Schedules and other similar attachments have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Company hereby undertakes to furnish supplementally copies of any of the omitted schedules and attachments upon request by the SEC.)			
3.1	Composite of Certificate of Incorporation of the Company inclusive of all amendments through June 14, 2007	The Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2007	0-24260	3.1
3.2	Composite of By-Laws of the Company inclusive of all amendments through April 20, 2016	The Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2016	0-24260	3.2
*†10.1	Amendment to the Amedisys, Inc. 2008 Omnibus Incentive Compensation Plan, dated September 25, 2018			
*†10.2	Amedment to the Amedisys, Inc. 2018 Omnibus Incentive Compensation Plan, dated September 25, 2018			
*10.3	Amended and Restated Employment Agreement, dated as of September 27, 2018, by and among Amedisys, Inc., Amedisys Holding, L.L.C. and Paul B. Kusserow	The Company's Current Report on Form 8-K filed on October 3, 2018	0-24260	10.1
†31.1	Certification of Paul B. Kusserow, President and Chief Executive Officer (principal executive officer), pursuant to Section 302 of the Sarbanes-Oxley Act of 2002			
†31.2	Certification of Scott G. Ginn, Chief Financial Officer (principal financial officer), pursuant to Section 302 of the Sarbanes-Oxley Act of 2002			
††32.1	Certification of Paul B. Kusserow, President and Chief Executive Officer (principal executive officer), pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002			
††32.2	Certification of Scott G. Ginn, Chief Financial Officer (principal financial officer), pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002			
†101.INS	XBRL Instance			
†101.SCH	XBRL Taxonomy Extension Schema Document			
†101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document			
†101.DEF	XBRL Taxonomy Extension Definition Linkbase			
†101.LAB	XBRL Taxonomy Extension Labels Linkbase Document			
†101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document			

### SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

AMEDISYS, INC. (Registrant)

By:

/s/ SCOTT G. GINN

Scott G. Ginn, Principal Financial Officer and Duly Authorized Officer

Date: October 30, 2018

(Back To Top)

38

# Section 2: EX-2.1 (EXHIBIT 2.1)

Exhibit 2.1

**EXECUTION VERSION** 

# STOCK PURCHASE AGREEMENT

by and among

### AMEDISYS HOSPICE, L.L.C.,

### COMPASSIONATE CARE HOSPICE GROUP, INC.,

### **MILTON HECHING,**

### HECHING 2012 EXEMPT IRREVOCABLE TRUST,

AND

### AMEDISYS, INC.

Dated as of October 9, 2018

# **TABLE OF CONTENTS**

ARTICLE I	DEFINED TERMS	1
1.1	Definitions	1
ARTICLE II	PURCHASE AND SALE OF SHARES; PURCHASE PRICE; ETC	14
2.1	Purchase and Sale of Shares	14
2.2	Closing	14
2.3	Certain Definitions; Pre-Closing Deliveries	14
2.4	Payments at Closing	15
2.5	Deliveries at Closing by the Sellers	16
2.6	Deliveries at Closing by the Buyer	16
ARTICLE III	REPRESENTATIONS AND WARRANTIES	16
3.1	Representations and Warranties of the Company	16
3.2	Representations and Warranties of the Buyer	37
3.3	Representations and Warranties of the Sellers	39
3.4	Representations and Warranties of Amedisys	42
ARTICLE IV	COVENANTS	43
4.1	Conduct of the Acquired Companies' Businesses	43
4.2	Publicity	45
4.3	Confidentiality	46
4.4	Access to Information	47
4.5	Filings and Authorizations	47
4.6	Closing Efforts	49
4.7	Compliance with Employment Laws	49
4.8	Further Assurances	49
4.9	Exclusivity	50
4.10	Books and Records	50
4.11	Certain Tax Matters	50
4.12	Release by the Sellers	56
4.13	Supplemental Disclosure Schedules	57
4.14	Additional Covenants	57
4.15	Payment and Performance Guarantee by Amedisys	60
ARTICLE V	CONDITIONS OF PURCHASE	61
5.1	Conditions to Obligations of the Buyer	61
5.2	Conditions to Obligations of the Sellers	63
ARTICLE VI	TERMINATION	63
6.1	Termination of Agreement	63
6.2	Effect of Termination	64



ARTICLE VII	INDEMNIFICATION	65
7.1	Survival	65
7.2	Indemnification of the Buyer Indemnitees	65
7.3	Indemnification of the Sellers	68
7.4	Indemnification Procedures and Related Provisions and Additional Limitations	69
7.5	Insurance and Other Third Party Recoveries	71
7.6	Manner of Payment	71
7.7	Remedies Exclusive	72
7.8	Tax Treatment of Indemnity Payments	72
7.9	No Circular Recovery	72
ARTICLE VIII	ADDITIONAL OPERATIVE PROVISIONS	73
8.1	Assignment; Binding Effect	73
8.2	Choice of Law	73
8.3	Consent to Jurisdiction and Service of Process; Waiver of Jury Trial	73
8.4	Notices	74
8.5	Headings	76
8.6	Fees and Expenses	76
8.7	Entire Agreement; Schedules	76
8.8	Interpretation	76
8.9	Waiver and Amendment	77
8.10	Third-party Beneficiaries	77
8.11	Non-Recourse	77
8.12	Severability	77
8.13	Counterparts; Facsimile Signatures	77
8.14	Specific Performance	78
8.15	Provision Respecting Representation of the Sellers	78

ii

# <u>Exhibit</u>

- Exhibit A Sellers' Ownership of Acquired Companies
- Exhibit B Corporate Integrity Agreement
- Exhibit C Escrow Agreement
- Exhibit D Form of Note
- $\underline{\text{Exhibit E}}$  Form of Restrictive Covenant Agreement
- Exhibit F Stock Powers

# **Schedules**

Schedule I - Required Governmental Entity and Third Party Consents, Approvals and Notices

Schedule II – Agency Locations

<u>Schedule 4.2</u> – Employee Communication Plan

Schedule 4.14(h) – Acquired Companies to be Dissolved

Schedule 5.1(j)(vii) – Repayment and OCR Disclosure Certificate

Disclosure Schedule

Buyer's Disclosure Schedule

# iii

### STOCK PURCHASE AGREEMENT

This **STOCK PURCHASE AGREEMENT** is made and entered into and effective as of October 9, 2018, by and among **MILTON HECHING**, a resident of the lawful age of majority of the State of Florida ("<u>Heching</u>"), the **HECHING 2012 EXEMPT IRREVOCABLE TRUST**, a trust formed under the laws of the State of Illinois (the "<u>Trust</u>" and together with Heching the "<u>Sellers</u>," each, a "<u>Seller</u>"), **AMEDISYS HOSPICE, L.L.C.**, a Louisiana limited liability company ("<u>Buyer</u>"), and **COMPASSIONATE CARE HOSPICE GROUP, INC.**, a Florida corporation (the "<u>Company</u>"). Amedisys, Inc., a Delaware corporation ("<u>Amedisys</u>"), also joins in this Agreement solely for purposes of <u>Section 3.4</u>, <u>Section 4.3(a)</u>, <u>Section 4.15</u> and ARTICLE VIII hereof.

### BACKGROUND

As further detailed on <u>Exhibit A</u>, the Sellers collectively, directly or indirectly, own one hundred percent (100%) of the outstanding equity interests of the Acquired Companies (as defined herein).

The Sellers desire to sell to Buyer, and Buyer desires to purchase and acquire from the Sellers, all of the issued and outstanding equity interests of the Acquired Companies.

**NOW, THEREFORE,** for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

### ARTICLE I DEFINED TERMS

**1.1** <u>Definitions</u>. For purposes of this Agreement, the following capitalized terms have the meanings assigned to them in this <u>Section 1.1</u>. Capitalized terms used in this Agreement and not defined in this <u>Section 1.1</u>, have the meanings assigned to them elsewhere in this Agreement.

"280G Resolutions" has the meaning set forth in Section 4.14(h).

"<u>Acquired Companies</u>" means, collectively, the Company and its Subsidiaries, including the Non-Wholly Owned Subsidiaries, and "<u>Acquired Company</u>" means any one of the Acquired Companies.

"<u>Action</u>" means any action, claim, complaint, investigation, recoupment effort, claims review, suit, arbitration or similar proceeding, whether civil or criminal, at law or in equity by or before any Governmental Entity or Private Program.

"<u>Affiliate</u>" means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a specified Person. A Person shall be deemed to control another Person if such first Person possesses, directly or indirectly, the power to direct, or cause the direction of, the management and policies of such other Person, whether through the ownership of voting securities, by contract or otherwise.

"<u>Agencies</u>" means, collectively, the respective hospice agency operating locations, including branch offices, subunits and drop sites, owned by the Acquired Companies, the individual addresses of which are listed on <u>Schedule II</u>.

"<u>Agreement</u>" means this Stock Purchase Agreement, as the same may be amended or supplemented, together with all exhibits and schedules attached hereto.

"<u>Allocation</u>" has the meaning set forth in <u>Section 4.11(j)</u>.

"<u>Asserted Liability</u>" has the meaning set forth in <u>Section 7.4(c)</u>

"Balance Sheet" means the unaudited, consolidated balance sheet of the Acquired Companies as of the Balance Sheet Date.

"Balance Sheet Date" means May 31, 2018.

"Base Purchase Price" means \$340,000,000.

"<u>Benefit Plan</u>" means each "employee benefit plan" within the meaning of Section 3(3) of ERISA that is established, maintained, sponsored or contributed to by any of the Acquired Companies for the benefit of any current or former employees, contractors, directors or officers thereof, including, without limitation, any pension, retirement, savings, disability, medical, dental, health, life, death benefit, group insurance, profit sharing, deferred compensation, stock option, bonus, incentive, or performance awards, vacation pay, equity incentive, phantom ownership, tuition reimbursement, severance or termination pay, or other employee benefit plan, agreement, arrangement, practice, program or policy.

"<u>Board of Directors</u>" means (i) with respect to a corporation, its board of directors, (ii) with respect to a limited liability company, its board of managers (if any), managers (if manager managed), managing members (if member managed) or other governing body, and (iii) with respect to any other non-natural Person, its equivalent governing body.

"<u>Business Day</u>" means any day other than a Saturday, Sunday or a day on which banks are permitted or required to be closed in Parsippany, New Jersey and Baton Rouge, Louisiana.

"Buyer" has the meaning set forth in the first paragraph of this Agreement.

"Buyer's Disclosure Schedule" means the disclosure schedule of Buyer referred to in and delivered to the Company pursuant to this Agreement

"<u>Buyer Indemnitees</u>" has the meaning set forth in <u>Section 7.2(a)</u>.

"<u>Capital Stock</u>" means, with respect to: (i) any corporation, any share, or any depositary receipt or other certificate representing any share, of an equity ownership interest in that corporation; and (ii) any other Entity, any share, membership or limited liability company interest, joint venture interest, partnership interest or other ownership interest, unit of participation or other equivalent (however designated) of an equity interest in that Entity.

"<u>Claim Notice</u>" has the meaning set forth in <u>Section 7.4(a)</u>.

"<u>Closing</u>" has the meaning set forth in <u>Section 2.2</u>.

"Closing Cash" has the meaning set forth in Section 2.3(a)(i).

"<u>Closing Cash Target</u>" means \$7,000,000.

"<u>Closing Date</u>" has the meaning set forth in <u>Section 2.2</u>.

"<u>Closing Period</u>" has the meaning set forth in <u>Section 2.2</u>.

"Closing Time" has the meaning set forth in Section 2.2.

"<u>Code</u>" means the Internal Revenue Code of 1986, as amended.

"Company" has the meaning set forth in the first paragraph of this Agreement.

"<u>Company Common Stock</u>" means the Capital Stock of the Company, consisting of (a) 100 shares of Class A voting common stock with no par value (the "<u>Class A Company Common Stock</u>") and (b) 900 shares of Class B non-voting common stock with no par value (the "<u>Class B Company Common Stock</u>").

"<u>Company Contracts</u>" has the meaning set forth in <u>Section 3.1(k)</u>.

"<u>Company Expenses</u>" means (i) the out-of-pocket fees and expenses incurred by the Sellers and/or the Acquired Companies prior to the Closing in connection with the transactions contemplated by this Agreement that remain unpaid as of the Closing, including, without limitation, the fees and expenses of Lowenstein Sandler LLP and (ii) any retention bonus payment obligations of the Company pursuant to any retention agreements entered into by the Company and its employees after the date hereof in accordance with the terms of Part 4.1 of the Disclosure Schedule, provided, however, (x) the Company Expenses to be included pursuant to this clause (ii) shall not exceed the lesser of (A) \$300,000 and (B) an amount equal to the product of (I) \$300,000 times (II) a fraction, the numerator of which is the maximum aggregate amount of payments that may be made by the Company under any such retention agreements determined as of the Business Day immediately prior to the Closing, and the denominator of which is the maximum aggregate amount of payments that could have been made by the Company pursuant to any such retention as of the date such retention agreements were originally entered into by the Company and the related employees, excluding, in each case in this clause (II) employees on leaves of absence, and (y) it is

understood and agreed that the payees of such Company Expenses shall not receive payment unless and until they are entitled to payment in accordance with the terms of their respective retention agreements. For the avoidance of doubt, Company Expenses shall not include (1) any amount included in Indebtedness, (2) those accounting fees incurred after the date hereof pursuant to Section 4.14(a)(ii), or (3) fees and expenses of certain service providers as described on Part 1 of the Disclosure Schedule.

"Company Intellectual Property" means all Intellectual Property owned by or licensed to any of the Acquired Companies.

"Company Knowledge Group" has the meaning set forth in the definition of "Knowledge."

"Company Lease" has the meaning set forth in Section 3.1(h)(iii).

"Company Owned Real Property" means any real property owned by an Acquired Company.

"<u>Company Tax Return</u>" has the meaning set forth in <u>Section 4.11(a)</u>.

"<u>Confidential Information</u>" means, with respect to any Person, all trade secrets and other confidential, nonpublic or proprietary information prepared for, by or on behalf of, or in the possession of, that Person, including any such information derived from reports, investigations, research, studies, work in progress, codes, marketing, sales or service programs, customer lists, records relating to past service provided to customers, capital expenditure projects, cost summaries, equipment or production system designs or drawings, pricing formulae, contract analyses, financial information, projections, business plans, agreements with vendors, joint venture agreements, confidential filings with any Governmental Entity and all other confidential, nonpublic concepts, methods, techniques or processes of doing business, ideas, materials or information prepared or performed for, by or on behalf of that Person.

"Contract" means any contract, agreement, indenture, lease or sublease, purchase order, license, note, bond or mortgage.

"Core Service Contracts" has the meaning set forth in Section 3.1(k).

"<u>Corporate Integrity Agreement</u>" means that certain Corporate Integrity Agreement between the Office of Inspector General of the Department of Health and Human Services, Compassionate Care Hospice Group, Ltd. and Compassionate Care Hospice of New York, LLC, effective as of January 30, 2015, a copy of which is attached hereto as <u>Exhibit B</u>.

"<u>Derivative Securities</u>" of a specified Entity means any Capital Stock, debt security or other Indebtedness of the specified Entity or any other Person which is convertible into or exchangeable for, or any option, warrant or other right to acquire (i) any unissued Capital Stock of the specified Entity or (ii) any Capital Stock of the specified Entity which has been issued and is being held by the Entity directly or indirectly as treasury Capital Stock.

"Disclosure Schedule" means the disclosure schedule of the Company referred to in and delivered to Buyer pursuant to this Agreement.

"Distribution Schedule" has the meaning set forth in Section 2.3(b)(i).

"Employee Communication Plan" has the meaning set forth in Section 4.2.

"<u>Encumbrance</u>" means any charge, lien, encumbrance, security interest, option, pledge, mortgage, deed of trust, hypothecation, conditional sale or restriction on transfer of title or voting, whether imposed by agreement, understanding, law, equity or otherwise, except for any restrictions on transfer generally arising under any applicable federal or state securities laws, rules or regulations.

"Entity" means any sole proprietorship, corporation, partnership of any kind having a separate legal status, limited liability company, business or other trust, unincorporated organization or association, mutual company, joint stock company or joint venture.

"<u>Environmental Law</u>" means any Law enacted and in effect on or prior to the Closing concerning pollution or the protection of the environment or human exposure to Hazardous Materials, including any use, storage, recycling, treatment, generation, transportation, processing, handling, labeling, management, disposal, remediation, Release or threatened Release of, or exposure to, any Hazardous Material.

"Environmental Permit" means any Permit required under any applicable Environmental Law.

"<u>ERISA</u>" means the Employee Retirement Income Security Act of 1974, as amended, and the related regulations and published interpretations.

"<u>ERISA Affiliate</u>" means any trade or business, whether or not incorporated, that, together with any of the Acquired Companies, is treated as a "single employer" under Section 414(b), (c), (m), or (o) of the Code and any related Treasury regulation thereunder.

"Escrow Agent" has the meaning set forth in Section 2.4(a)(ii).

"<u>Escrow Agreement</u>" means the Escrow Agreement, dated as of the Closing Date, by and among Buyer, Sellers and the Escrow Agent, in the form attached hereto as <u>Exhibit C</u>.

"Escrow Fund" has the meaning set forth in Section 2.4(a)(ii).

"Escrow Period" has the meaning set forth in Section 7.6(a).

"<u>Financial Statements</u>" means the audited, consolidated balance sheet and related consolidated statements of operations, stockholders' equity and cash flow of the Acquired Companies as of and for the fiscal year ended December 31, 2017, including the notes thereto, and the unaudited interim balance sheet and related unaudited consolidated statements of operations,

stockholders' equity and cash flows of the Acquired Companies for the five-month period ending on the Balance Sheet Date.

"First Release Date" has the meaning set forth in Section 7.6(a).

"Fraud" means, with respect to the Acquired Companies and the Sellers, on the one hand, or Buyer and Amedisys, on the other hand, an actual and intentional fraud by such Person with respect to the making of the representations and warranties set forth in this Agreement, provided, that such actual and intentional fraud of such Person shall only be deemed to exist if any of the individuals included in the related "knowledge group" of such Person (as set forth in the definition of "Knowledge") had actual knowledge that the representations and warranties made by such Person in this Agreement were actually breached when made, with the express intention to deceive the counterparties to this Agreement and that such counterparties rely thereon to their detriment.

"Fundamental Representations" has the meaning set forth in Section 7.1.

"GAAP" means generally accepted accounting principles in the United States, as in effect from time to time.

"<u>Governmental Entity</u>" means any national, federal, state or local government, or any regulatory authority, administrative agency, bureau, board, commission, court, department, tribunal, arbitral body or other similar governmental, quasi-governmental or similar body exercising executive, legislative, judicial, regulatory or administrative authority or functions or instrumentality thereof or contractor therefor.

"Governmental Filings" has the meaning set forth in Section 3.1(c).

"<u>Government Program</u>" means any United States federal, state or local health care or reimbursement program administered by a Governmental Entity, including, without limitation, Medicare, Medicaid, CHAMPUS, or TRICARE, and any successor program to any of the above.

"Guaranteed Obligations" has the meaning set forth in Section 4.15.

"<u>Hazardous Material</u>" means any waste, chemical, material or other substance that is listed, defined, designated or classified as hazardous, radioactive or toxic or a pollutant or a contaminant (or words of similar import) by or pursuant to Environmental Laws.

"<u>Healthcare Laws</u>" means any Law or Legal Requirement relating to healthcare regulatory and reimbursement matters, including but not limited to (a) Title XVIII of the Social Security Act, 42 U.S.C. §§ 1395-1395hhh (the Medicare statute) (b) any joint federal or state health care or health insurance program, including, Title XIX of the Social Security Act, 42 U.S.C. §§ 1396-1396v (the Medicaid statute); (c) TRICARE, 10 U.S.C. § 1071 et seq.; (d) 42 U.S.C. §§ 1320a-7, 7a, and 7b, which are commonly referred to as the "Federal Fraud Statutes," and their state law counterparts; (e) 42 U.S.C. § 1395nn, which is commonly referred to as the "Stark Statute," and its state law counterparts; (f) 31 U.S.C. §§ 3729-3733, which is commonly referred to as the "federal False

Claims Act"; (g) the Program Fraud Civil Remedies Act, 31 U.S.C. §§ 3801-3812; (h) the Anti-Kickback Act of 1986, 41 U.S.C. §§ 51-58; (i) HIPAA and any other applicable Laws or Legal Requirements concerning the privacy and security of health information; (j) Laws and Legal Requirements relating to participation in or submission of claims to Government Programs and Private Programs; (k) applicable provisions of the Patient Protection and Affordable Care Act, 42 U.S.C. § 18001 et seq., as amended by the Health Care and Education Reconciliation Act of 2010 (Pub. L. 11-152), together with implementing regulations and any other rules or regulations promulgated thereunder; (l) any federal, state or local statute or regulation relevant to mail fraud, wire fraud, false statements or claims; (m) survey, certification, and standards as each relates to eligibility to obtain authorizations of Governmental Entities required to participate in Government Programs; (n) Medicare program conditions of participation and conditions of payment; and (o) all applicable federal, state, and local licensing, accreditation, regulatory, certificate of need, recordkeeping, referrals, quality, and safety Laws.

"<u>Healthcare Liability Claims</u>" means any liabilities of any Acquired Company (including any fines, penalties, or exclusions) relating to any audits, investigations, recoupment efforts, claims, actions, proceedings, lawsuits, demands or charges brought by U.S. Department of Health and Human Services Office of Inspector General, U.S. Department of Justice, a state attorney general, state Medicaid agency or Governmental Entities or Private Program to the extent applicable, related to healthcare fraud, abuse, Federal False Claims Act matters, qui tam or whistle blower actions, or the submission of healthcare claims prior to the Closing by any Acquired Company and any predecessors from whom any Acquired Company acquired any Medicare or Medicaid provider number.

"Heching" has the meaning set forth in the first paragraph of this Agreement.

"<u>HIPAA</u>" means the administrative simplification provisions of the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act, along with the regulations promulgated thereunder by the Secretary of the Department of Health and Human Services.

"Income Tax" means any federal, state, local or foreign income tax measured by or imposed on net income, including any interest, addition to tax or penalty imposed with respect thereto.

"Income Tax Return" means any return, report, declaration, information return or other document required to be filed with a Taxing Authority in connection with Income Taxes, and any claims for refunds of Income Taxes paid, including any amendments to any of the foregoing.

"<u>Indebtedness</u>" means, with respect to any Person, without duplication: (i) obligations for borrowed money, including all outstanding principal, interest, fees, and other amounts payable with respect thereto (including, for the avoidance of doubt, any prepayment penalties or make-whole payments required in connection with the repayment of the borrowed money prior to its stated maturity date); and (ii) guarantees of obligations described in clause (i) of any other Person. For the avoidance of doubt, Indebtedness shall not include obligations payable in installments or on a deferred basis that do not involve obligations for borrowed money, including the remaining payment

obligation under the Corporate Integrity Agreement, payments under real property or equipment leases and payments for the Company's insurance policies for the current annual coverage period.

"Indemnified Party" has the meaning set forth in Section 7.4(a).

"Indemnifying Party" has the meaning set forth in Section 7.4(a).

"Indemnity Escrow Amount" means \$28,000,000, which shall be available to satisfy all Losses relating to claims for indemnification pursuant to Section 7.4.

"Independent Accounting Firm" means Grant Thornton LLP or such other independent certified public accounting firm mutually agreed upon in writing by Buyer and Sellers prior to the Closing.

"Intellectual Property" means all registered and unregistered trademarks, service marks, trade names and Internet domain names and email addresses, slogans, designs, pictures or any other symbols used to identify any good and/or service (together with the goodwill symbolized by any of the foregoing), patents, patent applications, registered and unregistered copyrights, trade secrets, proprietary know-how and other confidential and proprietary information, moral rights and any other intellectual property rights recognized under any Laws or international conventions, and in any country or jurisdiction in the world, all applications, disclosures, renewals, extensions, continuations or reissues thereof, and all rights arising thereunder.

"<u>IT Systems</u>" means the hardware, software, data, databases, data communication lines, network and telecommunications equipment, Internet-related information technology infrastructure, wide area network and other information technology equipment owned, leased or licensed by an Acquired Company.

"<u>Knowledge</u>" with respect to (i) the Acquired Companies and the Sellers, means the actual knowledge of each of Heching, Aliza Heching, the Chief Financial Officer of the Company, Judith Grey, the Chief Executive Officer of the Company and Stella Hardy, the Director of Quality and Compliance of the Company (collectively, the "<u>Company Knowledge Group</u>"), as of the related date, together with the knowledge that such person would reasonably be expected to have following inquiry of all employees of the Acquired Companies reporting directly to such individuals, and (ii) with respect to Buyer and Amedisys, means the actual knowledge of Kristopher Novak, Senior Director of Mergers and Acquisitions of Amedisys, Scott G. Ginn, Chief Financial Officer of Amedisys, David L. Kemmerly, General Counsel of Amedisys and David B. Pearce, Chief Compliance Officer of Amedisys as of the related date, after due inquiry of all management-level employees of Amedisys reporting directly to such individuals.

"<u>Law</u>" means any statute, code, rule, regulation, order, ordinance, treaty, judgment or decree of any Governmental Entity having the effect of law, including without limitation Healthcare Laws.

"Legal Requirement" means at any time (i) any Law, injunction, writ, edict, award, authorization or other legally binding requirement of any Governmental Entity in effect at that time

or (ii) any legally binding obligation included in any certificate, certification, franchise, Permit or license issued by any Governmental Entity at that time.

"Look-Back Date" means October 1, 2015.

"Losses" to any specified Person means any loss, cost, damage, expense (including reasonable fees of and actual disbursements by attorneys, consultants, experts, auditors or other Representatives, including litigation costs), fine of, penalty on, or liability of any other nature of that Person; provided, however, in no event shall any Indemnifying Party be liable to any Indemnified Party for any consequential, special, exemplary, and punitive damages, other than those that are either awarded in a judgment or paid pursuant to a settlement entered into in accordance with the terms of this Agreement, in each case to a third party pursuant to a Third Party Claim. For the avoidance of doubt, damages that occur as a direct result of an incident shall not constitute consequential damages for purposes of the definition of Losses.

"<u>Mandatory Repayments</u>" shall mean any voluntary repayment required to be made by the Buyer or any Acquired Company following the Closing pursuant to any applicable Legal Requirements, including, without limitation, 42 C.F.R. 401.301 – 401.305 and any related "60 Day Rule" regulations, in each case to the extent the repayment obligation arises from the provision of healthcare services or the submission of healthcare claims prior to the Closing by any Acquired Company or any predecessors from whom any Acquired Company acquired any Medicare or Medicaid provider number.

"Material Adverse Effect" means any fact, event, series of events, change, effect or circumstance that, individually or in the aggregate with other facts, events, series of events, changes, effects or circumstances, has had or is reasonably likely to have a material adverse effect on the assets, properties, business, operations, condition (financial or otherwise), or results of operations of the Acquired Companies, taken as a whole; provided, that in no event shall any fact, event, series of events, change, effect or circumstance that arises out of or is attributable to any of the following, either alone or in combination, be deemed to constitute or contribute to a Material Adverse Effect, or shall otherwise be taken into account in the determination of whether a Material Adverse Effect has occurred or would be reasonably likely to occur: (i) any actual or proposed change in any Law, Legal Requirements or accounting standards (including GAAP) or interpretations or the enforcement thereof (except to the extent that such change has had, or is reasonably likely to have, a materially disproportionate effect on the Acquired Companies, taken as a whole, relative to other Persons in the hospice industry); (ii) conditions affecting any of the industries, industry sectors or geographic sectors in which the Acquired Companies operate or in which products of the Acquired Companies are used or distributed, or general business, financial, banking or economic conditions or debt, currency or capital markets (whether in the United States or any other country or in any international market), including changes in interest rates, exchange rates, commodity prices and fuel costs (except to the extent that such change has had, or is reasonably likely to have, a materially disproportionate effect on the Acquired Companies, taken as a whole, relative to other Persons in the hospice industry); (iii) acts of God, national or international political or social conditions, the engagement by the United States or other countries in hostilities, war or military operations, whether commenced before or after the date hereof, and whether or not pursuant to the declaration of a

national emergency or war, the occurrence of any military or terrorist attack, sabotage, any hurricane, flood, tornado, earthquake or other natural disaster, or any other force majeure event; (iv) the execution or delivery of this Agreement or the announcement, disclosure, pendency or consummation of the transactions contemplated by this Agreement; (v) effects resulting from the identity of Buyer or its Affiliates, or (vi) any failure by the Acquired Companies to meet any of their internal or published projections, forecasts or predications of revenues, earnings or cash flows for any period (although this <u>clause</u> (vi) shall not apply to the facts and circumstances that may have given rise or contributed to any such failure).

"<u>Medicaid</u>" means collectively, the healthcare assistance program established by Title XIX of the Social Security Act (42 U.S.C. §§1396 et seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders, guidelines or requirements pertaining to such program, including (a) all federal statutes (whether set forth in Title XIX of the Social Security Act or elsewhere) affecting such program, (b) all state statutes and plans for medical assistance enacted in connection with such program in the states in which the Company does business and federal rules and regulations promulgated in connection with such program, and (c) all applicable provisions of all rules, regulations, manuals, orders and administrative, reimbursement, guidelines and requirements of all Governmental Authorities promulgated in connection with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

"Medicare" means collectively, the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act (42 U.S.C. §§1395 et seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders or guidelines pertaining to such program, including (a) all federal statutes (whether set forth in Title XVIII of the Social Security Act or elsewhere) affecting such program, and (b) all applicable provisions of all rules, regulations, manuals, orders and administrative, reimbursement, guidelines and requirements of all Governmental Authorities promulgated in connected with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

"Medicare and Medicaid Recoupment Claims" shall mean any hospice cap liabilities (but only to the extent such cap was surpassed prior to the Closing Date for any particular provider or patient), and any cost-report, take-back, adjustment, recoupment or other liabilities of any Acquired Company related to any pre- or post-payment review of claims, reports, actions, audits, investigations, or proceedings conducted by or on behalf of any Government Programs, including, but not limited to, Department of Health and Human Services Office of Inspector General, Medicare administrative contractors or intermediaries, recovery audit contractors, zone program integrity contractors, specialty medical review contractors, or similar investigative agencies, in each case to the extent such liabilities arise from the provision of healthcare services or the submission of healthcare claims, or the incurrence of costs in excess of the hospice cap for any individual provider or patient, prior to the Closing by any Acquired Company or any predecessors from whom any Acquired Company acquired any Medicare or Medicaid provider number.

"Monthly Financial Statements" has the meaning set forth in Section 4.14(a)(i).

"<u>New York LLC</u>" means Compassionate Care Hospice of New York, LLC, a New York limited liability company.

"<u>Nondisclosure Agreement</u>" has the meaning set forth in <u>Section 4.3</u>.

"Non-Wholly Owned Subsidiaries" has the meaning set forth in the definition of "Subsidiary."

"<u>Note</u>" has the meaning set forth in <u>Section 2.4(a)(iv)</u>.

"<u>Notice Matter</u>" has the meaning set forth in <u>Section 7.4(b)</u>.

"<u>Notice Period</u>" has the meaning set forth in <u>Section 7.4(c)</u>

"<u>Organizational Documents</u>" means the documents by which any Person (other than an individual) establishes its legal existence or which govern its internal affairs (i.e., including, but not limited to, certificate of incorporation, certificate of formation, memorandum of association, articles of association, constitutional documents, by-laws, operating agreement, and shareholders agreement).

"<u>Outside Date</u>" has the meaning set forth in <u>Section 6.1(e)</u>.

"<u>Permits</u>" has the meaning set forth in <u>Section 3.1(n)(ii)</u>.

"Permitted Encumbrance" means (i) Encumbrances incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations; (ii) mechanics, carriers', workers', repairers', materialmen's, warehousemen's and similar Encumbrances which have arisen in the ordinary course of business; (iii) Encumbrances approved by Buyer in writing; (iv) Encumbrances for Taxes (A) not yet due and payable, or (B) if due, not yet delinquent or being contested in good faith; in each case provided that adequate reserves with respect thereto are maintained on the books of the related Person in accordance with GAAP; (v) requirements and restrictions of zoning, building and other Laws or Legal Requirements, which are incurred in the ordinary course of such Person's business or existing on property and not materially interfering with the ordinary conduct of such Person's business or such Person's use of such property; (vi) statutory liens of landlords for amounts not yet due and payable, provided that adequate reserves with respect thereto are maintained on the books of the related Person in accordance with GAAP; and (vii) liens arising under conditional sales contracts and equipment leases with third parties entered into in the ordinary course of business.

"<u>Person</u>" means an association, a corporation, an individual, a partnership, a limited liability company, a trust, or any other entity or organization, including a Governmental Entity.

"<u>Personal Information</u>" means (i) any "nonpublic personal information" as such term is defined under the Title V of the U.S. Gramm-Leach-Bliley Act, 15 U.S.C. § 6801 et seq., and the rules and regulations issued thereunder, (ii) any information that can specifically identify an individual, such as name, signature, address, social security number, telephone number or other unique identifier, together with any other information that relates to an individual who has been so identified in any format whether written, electronic or otherwise, (iii) information that can be used to authenticate an individual (including, without limitation, passwords or PINs, biometric data, unique identification numbers, answer to security questions, or other personal identifiers) in any format whether written, electronic or otherwise, or (iv) any personally identifiable medical, financial and other personal information about proposed, current and former patients.

"<u>Pre-Closing Period</u>" means any taxable period that ends on or before the Closing Date.

"<u>Privacy Laws</u>" means any applicable federal, or state laws relating to the use, collection, storage, disclosure and transfer of Personal Information.

"<u>Private Programs</u>" means such private, commercial and non-governmental programs, including without limitation any private, commercial and non-governmental insurance program, under which any of the Acquired Companies, directly or indirectly, is presently receiving payments or is eligible to receive payments.

"Pro Rata Portion" means the percentages for each of Heching and Trust, as set forth on the Distribution Schedule.

"<u>Purchase Price</u>" has the meaning set forth in <u>Section 2.3(a)(ii)</u>.

"<u>Release</u>" means the release, spill, emission, leaking, pumping, pouring, emptying, escaping, dumping, injection, deposit, disposal, discharge, dispersal, leaching or migrating into or through the environment (including natural resources, threatened or endangered species, indoor or ambient air, surface water, groundwater, drinking water supplies, creek, stream or river sediments, marsh or wetlands and land surface or subsurface strata or soil).

"<u>Sale Bonus Payment</u>" means the payment owing by the Company to Judith Grey upon consummation of the Closing pursuant to that certain Executive Employment Agreement by and between Ms. Grey and the Company, dated November 17, 2017.

"Second Release Date" has the meaning set forth in Section 7.6(a).

"Section 338(h)(10) Election" has the meaning set forth in Section 4.11(j).

"Special Health Care Representations" means those representations and warranties contained in clause (t) (Healthcare Matters) of Section 3.1

"Special Representations" has the meaning set forth in Section 7.1(b).

"Stock Powers" has the meaning set forth in Section 5.1(j)(i).

"Straddle Period" means any taxable period that includes (but does not end on) the Closing Date.

"Subsidiary" means any Person whose securities or other ownership interests having by their terms the power to elect a majority of the Board of Directors are owned or controlled, directly or indirectly, by the Company or one or more of its Subsidiaries. For purposes of this Agreement, and without limitation of the foregoing, each of Compassionate Care Hospice of Central Florida, Inc., f/k/a Compassionate Care Hospice of Miami Dade & Florida Keys, Inc., and Compassionate Care Hospice of Lake and Sumter, Inc., each a Florida corporation (the "Florida Non-Wholly Owned Subsidiaries"), and Pathways to Compassion of California, LLC, a California limited liability company ("Pathways" and collectively, the "Non-Wholly Owned Subsidiaries"), shall constitute a Subsidiary.

"Survival Expiration Date" has the meaning set forth in Section 7.1.

"<u>Tax</u>" means any and all federal, state, local, foreign and other taxes, including net income, gross income, gross receipts, sales, use, ad valorem, hospital, provider, unclaimed property, transfer, franchise, profits, license, lease, rent, service, service use, withholding, payroll, employment, excise, severance, privilege, stamp, occupation, premium, property, windfall profits, alternative minimum, estimated, customs, duties or other taxes, and other governmental fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts imposed with respect thereto.

"<u>Tax Claim</u>" has the meaning set forth in <u>Section 4.11(g)</u>.

"<u>Tax Return</u>" means any return, report, declaration, information return or other document required to be filed with a Taxing Authority in connection with any Taxes, and any claims for refunds of Taxes paid, including any amendments to any of the foregoing and any schedules or attachments thereto.

"Taxing Authority" means any Governmental Entity having or purporting to exercise jurisdiction with respect to any Tax.

"Threshold" has the meaning set forth in Section 7.2(b)(ii).

"Tipping Basket" has the meaning set forth in Section 7.2(b)(i).

"<u>Transfer Taxes</u>" means any sales, use, stock transfer, real property transfer (including controlling interest transfer), real property gains, transfer, stamp, registration, documentary, recording or similar duties or taxes incurred in connection with the transactions contemplated hereby, including any interest, addition or penalties imposed with respect thereto.

"<u>Unresolved Claim</u>" has the meaning set forth in <u>Section 7.6(a)</u>.

# ARTICLE II PURCHASE AND SALE OF SHARES; PURCHASE PRICE; ETC

2.1 <u>Purchase and Sale of Shares</u>. In accordance with, and subject to, the provisions of this Agreement, at the Closing, Buyer shall purchase from the Sellers, and the Sellers shall transfer, convey, assign, and deliver to Buyer, all of the Sellers' right, title and interest in and to (a) the shares of Company Common Stock owned by the Sellers, free and clear of any Encumbrances and (b) the Capital Stock of the Non-Wholly Owned Subsidiaries owned by the Sellers, free and clear of any Encumbrances. The shares of Company Common Stock of the Non-Wholly Owned Subsidiaries purchased pursuant to this <u>Section</u> 2.1 shall constitute 100% of the issued and outstanding Capital Stock in the Company and the Non-Wholly Owned Subsidiaries.

**2.2** <u>Closing</u>. Unless this Agreement is earlier terminated pursuant to <u>Section 6.1</u>, the consummation of the transactions contemplated hereby (the "<u>Closing</u>") will take place via the electronic exchange of documents and funds as follows: (a) if the conditions set forth in <u>Sections 5.1</u> and <u>5.2</u>, other than those conditions which, by their terms, are to be satisfied or waived at the Closing, have been satisfied or waived by the parties on or before the 20<sup>th</sup> day of the related calendar month, the Closing shall occur on the first day of the first calendar month following that in which such conditions were satisfied or waived, (b) if the conditions set forth in <u>Sections 5.1</u> and <u>5.2</u>, other than those conditions which, by their terms, are to be satisfied or waived at the Closing, have been satisfied or waived by the parties after the 20<sup>th</sup> day of the related calendar month, the Closing shall occur on a mutually agreeable date within twelve (12) calendar days after such conditions were satisfied or waived (the period in which the Closing is permitted to occur pursuant to <u>Sections 2.2(a)</u> and <u>2.2(b)</u>, as applicable, following the satisfaction or waived at the Closing, is hereinafter referred to as the "<u>Closing Period</u>"), or (c) such other date that is agreed to in writing by the Sellers and Buyer (in any case, the "<u>Closing Date</u>"). Unless otherwise agreed to in writing by the Sellers and Buyer, the Closing shall be effective as of 12:01 a.m., central time, on the Closing Date unless the Closing Date (such effective time, the "<u>Closing Time</u>").

# 2.3 Certain Definitions; Pre-Closing Deliveries.

(a) <u>Certain Definitions</u>. As used herein, the following terms shall have the following meanings:

(i) "<u>Closing Cash</u>" means, as of the Closing Time, the consolidated bank account balances of the Acquired Companies, *plus*, any deposits received by the Acquired Companies from Governmental Entities on the last Business Day immediately prior to the Closing Date which are shown as pending on the documentation to be delivered by the Sellers to Buyer pursuant to <u>Section 2.3(c)</u>.

(ii) "<u>Purchase Price</u>" means the Base Purchase Price, (A) *minus* the employer's share of all employment Taxes owing by the Company in connection with the payment of the Sale Bonus Payment, (B) *plus* or *minus*, the amount, if any, by which Closing Cash is greater than or less than Target Closing Cash, as determined pursuant to <u>Section 2.3(c)</u>.

(b) <u>Closing Payment Schedule</u>. At least five (5) Business Days prior to the Closing Date (or such later date as is acceptable to Buyer), the Sellers will furnish (or cause to be furnished) to Buyer the following:

(i) a spreadsheet (the "<u>Distribution Schedule</u>") setting forth the Pro Rata Portion for each Seller and the calculation of the Purchase Price that each Seller is entitled to receive in the form of a Note (as defined below) at the Closing, in accordance with <u>Section 2.4</u> below;

(ii) a list of the payees of Company Expenses that will receive payment at the Closing, the amounts payable thereto, and their wire instructions for payment; and

(iii) a signed payoff letter from each holder of outstanding Indebtedness, if any, in form and content reasonably acceptable to Buyer, (A) indicating the amount required to discharge such Indebtedness at the Closing, and the wire instructions for payment, and (B) if such Indebtedness is secured by any Encumbrances, agreeing to release such Encumbrances upon receipt of the payoff amount.

The parties agree that Buyer shall be entitled to rely on the Distribution Schedule in making payments under this ARTICLE II and shall not be responsible for the calculations or the determinations regarding such calculations in such Distribution Schedule.

(c) <u>Closing Cash Certificate</u>. On the last Business Day immediately prior to the Closing Date, the Sellers will furnish (or cause to be furnished) to Buyer a certificate, in form and content reasonably acceptable to Buyer, certifying the amount of Closing Cash, together with printed snapshots of the Acquired Companies' bank account balances and received but pending deposits from Governmental Entities at the close of business on the last Business Day immediately prior to the Closing Date. If Closing Cash is less than the Target Closing Cash, the Purchase Price will be decreased on a dollar for dollar basis to the extent of such shortfall, and if Closing Cash is greater than Target Closing Cash, the Purchase Price will be increased on a dollar for dollar basis to the extent of such overage.

# 2.4 Payments at Closing.

(a) <u>Payments at Closing</u>. At the Closing, Buyer shall make the following payments (in an amount, in the aggregate, equal to the Purchase Price) by wire transfer of immediately available funds:

(i) first, on behalf of the Acquired Companies to the respective holders of Indebtedness, if any, outstanding as of the Closing, the amounts specified in the payoff letters delivered by the Company to Buyer pursuant to Section 2.3(b)(iii);

(ii) second, to Fifth Third Bank (the "<u>Escrow Agent</u>"), an amount equal to \$28,000,000 (together with all earnings thereon, the "<u>Escrow Fund</u>"), to be held and disbursed pursuant to the terms of the Escrow Agreement and this Agreement;

(iii) third, on behalf of the Acquired Companies, to such payees of the Company Expenses as directed in writing by the Company pursuant to <u>Section 2.3(b)(ii)</u>; and

(iv) fourth, to the Company, an amount equal to the Sale Bonus Payment (and Buyer shall cause the Company to pay such amount to Ms. Grey, less withholdings required by Law), if applicable; and

(v) Buyer shall issue to each of the Sellers a one-day promissory note in the form and substance of the promissory note annexed hereto as <u>Exhibit D</u> (each a "<u>Note</u>" and collectively, the "<u>Notes</u>"), which shall become due and payable, in accordance with this <u>Section 2.4(a)(v)</u>, on the Business Day immediately following the Closing Date. The aggregate principal amount of the Note issued to each Seller shall be equal to the Pro Rata Portion of the Purchase Price to be paid to such Seller.

2.5 <u>Deliveries at Closing by the Sellers</u>. At the Closing, and upon satisfaction or waiver of the conditions set forth in <u>Section 5.2</u>, the Sellers will deliver or cause to be delivered the instruments, consents, certificates and other documents required of him or it by <u>Section 5.1</u>.

**2.6** Deliveries at Closing by the Buyer. At the Closing, and upon satisfaction or waiver of the conditions set forth in Section 5.1, Buyer will deliver or cause to be delivered the instruments, consents, certificates and other documents required of Buyer by Section 5.2.

# ARTICLE III <u>REPRESENTATIONS AND WARRANTIES</u>

**3.1** <u>Representations and Warranties of the Company</u>. Except as set forth on the Disclosure Schedule, subject to <u>Section 8.7</u>, the Company represents and warrants to Buyer that each of the statements contained in this <u>Section 3.1</u> is true and correct as of the date hereof:

(a) <u>Due Organization and Good Standing</u>. Each Acquired Company is duly organized or incorporated, validly existing and is in good standing (or the equivalent) under the Laws of the state of its organization or incorporation, which states of organization or incorporation are listed on <u>Part 3.1(a)</u> of the Disclosure Schedule. Each Acquired Company is qualified or otherwise authorized to transact business as a foreign limited liability company or corporation and is in good standing (or the equivalent) under the Laws of every other state in which such qualification or authorization is required under applicable Law, with such exceptions as would not, individually or in the aggregate, reasonably be expected to materially impair the operation of the business of any Acquired Company, or give rise to a material liability of the Acquired Companies, taken as a whole. Each Acquired Company has all requisite corporate or other requisite organizational power and authority to own, lease and operate its respective properties and to carry on its respective businesses as now conducted.

(b) <u>Authorization, Execution and Delivery; Valid and Binding Agreement</u>. The Company has the requisite power and authority and has taken all required action on its part necessary to permit it to execute and deliver and to carry out the terms of this Agreement and the other agreements, instruments and documents of the Company contemplated hereby. This Agreement has been, and each of such other agreement, instrument and document will be at or prior to the Closing, duly and validly executed and delivered by the Company, and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and such other agreements, instruments and documents, when so executed and delivered will constitute, the legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

(c) <u>Governmental Filings</u>. No filings or registration with, notification to, or authorization, consent or approval of any Governmental Entity (collectively, "<u>Governmental Filings</u>") are required in connection with the execution, delivery and performance of this Agreement or the other agreements, instruments and documents of the Acquired Companies contemplated hereby, except (i) Governmental Filings that become applicable or are required solely as a result of matters specifically related to Buyer or its Affiliates, (ii) the Governmental Filings set forth on <u>Part 3.1(c)</u> of the Disclosure Schedule and (iii) where the failure to obtain, give or make such Governmental Filings would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

### (d) Capital Structure; Subsidiaries.

(i) The issued and outstanding Capital Stock of the Company consists of 1,000 shares of stock, divided into 100 shares of Class A Company Stock and 900 shares of Class B Company Stock, all of which are owned by the Sellers free and clear of any Encumbrances. The issued and outstanding shares of Company Common Stock have been duly authorized, and are validly issued, fully paid and nonassessable.

(ii) For each of the Florida Non-Wholly Owned Subsidiaries, the Capital Stock is owned ninety percent (90%) by the Trust and ten percent (10%) by the Company, free and clear of any Encumbrances. The issued and outstanding Capital Stock of the Florida Non-Wholly Owned Subsidiaries has been duly authorized, and are validly issued, fully paid and nonassessable.

(iii) Ninety four percent (94%) of the Capital Stock of Pathways is owned by the Trust, five percent (5%) is owned by the Company, and one percent (1%) is owned by Heching, free and clear of any Encumbrances. The issued and outstanding Capital Stock of Pathways has been duly authorized, and are validly issued, fully paid and nonassessable.

(iv) <u>Part 3.1(d)(iv)</u> of the Disclosure Schedule accurately lists each Subsidiary, describes the Capital Stock of such Subsidiary that is outstanding and the holder thereof, and identifies whether such Capital Stock is certificated and, if so, includes the certificate number

and date of issuance thereof. Except as set forth on <u>Part 3.1(d)(iv)</u> of the Disclosure Schedule, all of the issued and outstanding equity ownership interests of each Subsidiary are directly or indirectly owned by the Sellers or another Subsidiary, free and clear of any Encumbrances (other than Permitted Encumbrances), except as set forth on <u>Part 3.1(d)(iv)</u> of the Disclosure Schedule. Each such equity ownership interest has been duly authorized, and is validly issued, fully paid and nonassessable (in each case, to the extent such concept is applicable to such equity ownership interest). Except for the New York LLC and as set forth on <u>Part 3.1(d)</u> (<u>iv)</u> of the Disclosure Schedule, neither Heching, the Trust nor any settlor, trustee or beneficiary of the Trust owns, directly or indirectly, any right, title or interest to or in any other Person engaged in the business of providing hospice care or palliative services.

(v) There are no outstanding options, warrants, convertible or exchangeable securities or other rights that would obligate any of the Acquired Companies to issue shares of its respective equity ownership interests. The offer, issuance and sale of all outstanding equity ownership interests of the Acquired Companies were made in compliance with all applicable federal and state securities laws. Except as set forth on Part 3.1(d)(v) of the Disclosure Schedule, there are no agreements, written or oral, to which any Acquired Company is a party relating to the acquisition, redemption, disposition, voting or registration under applicable securities laws of any equity ownership interest of such Acquired Company. There are no outstanding or authorized stock appreciation, phantom stock or similar rights with respect to any of the Acquired Companies.

(vi) The Distribution Schedule is and at all times shall be true and correct and accurately set forth the amount of the Pro Rata Portion of the Purchase Price owing to each Seller. Except as set forth on the Distribution Schedule, no payments shall be owing to the Sellers in connection with the transactions contemplated by this Agreement.

(e) <u>Financial Statements</u>. <u>Part 3.1(e)</u> of the Disclosure Schedule attaches true and complete copies of the Financial Statements. Except as set forth in the footnotes thereto, the Financial Statements have been prepared in accordance with GAAP, subject, in the case of unaudited financial statements, to normal recurring adjustments necessary for a fair presentation of interim results (in accordance with GAAP) and the absence of notes thereto. Except as <u>Part 3.1(e)</u> of the Disclosure Schedule sets forth, the Financial Statements fairly present, in all material respects, in accordance with GAAP the consolidated financial position of the Acquired Companies as of the dates thereof and the consolidated results of operations, cash flows and stockholders' equity for the periods then ended (<u>provided</u> that the Financial Statements do not present any changes in stockholders' equity of the Acquired Companies for the interim period ended on the Balance Sheet Date), subject, in the case of unaudited financial statements, to normal recurring adjustments necessary for a fair presentation of interim results (in accordance with GAAP) and the absence of notes thereto. None of the Acquired Companies has committed any act of bankruptcy, is insolvent, has proposed a compromise or arrangement to its creditors generally, has had any petition for a receiving order in bankruptcy filed against it, has made a voluntary assignment in bankruptcy, has taken any proceeding with respect to a compromise or arrangement in bankruptcy, has taken any proceeding to have a receiver appointed to any part of its assets, or has had any debtor take possession of any of its property.

(f) <u>No Conflict or Violation</u>. Except as set forth on <u>Part 3.1(c)</u> or <u>Part 3.1(f)</u> of the Disclosure Schedule, the execution, delivery and performance by the Company of this Agreement and the other agreements, instruments and documents of the Acquired Companies contemplated hereby and the consummation by the Company of the transactions contemplated hereby and thereby, as applicable, do not (i) assuming all Governmental Filings described or referenced in <u>Section 3.1(c)</u> have been obtained or made, violate, in any material respect, any applicable Law or Legal Requirements to which any Acquired Company is subject, (ii) (A) require a consent or approval under, (B) conflict with, result in a violation or breach of, or constitute a default under, or (C) result in the acceleration of, or a right to accelerate, terminate or cancel, in each case, in any material respect, any Company Contract or Company Lease, or (iii) violate any Acquired Company's Organizational Documents, except, in the case of clauses (i) or (ii), as would not, individually or in the aggregate, reasonably be expected to materially impair the operation of the business of any Acquired Company, or give rise to a material liability of the Acquired Companies, taken as a whole.

(g) <u>Legal Proceedings</u>. Except as set forth on <u>Part 3.1(g)</u> of the Disclosure Schedule, (i) there are no Actions pending, or, to the Knowledge of the Company, threatened against any Acquired Company and (ii) none of the Acquired Companies is the subject of any judgment, decree, injunction or order of any Governmental Entity under which any unsatisfied obligation remains outstanding.

### (h) Personal Property; Owned and Leased Real Property.

(i) The Company has made available to Buyer a true, correct and complete listing, as of May 31, 2018, of (x) all tangible personal property owned by the Acquired Companies as of the Balance Sheet Date with a depreciated book value of at least 20,000, and (y) all material devices, servers, hard drives and other equipment of the Acquired Companies on which information protected by HIPAA may be stored or contained. Except as may be reflected in the Financial Statements or set forth on Part 3.1(h)(i) of the Disclosure Schedule, the Acquired Companies have good and valid title or rights, free and clear of any Encumbrances (except for Permitted Encumbrances), to all of their respective tangible personal property, except for such tangible personal property as has been disposed of in the ordinary course of business consistent with past practice since the Balance Sheet Date. Except as set forth on Part 3.1(h)(i) of the Disclosure Schedule, all of the tangible personal property of the Acquired Companies is in good working order and condition in accordance with industry practice, ordinary wear and tear excepted, and adequate in all material respects for the purposes for which they presently are being used or held for use, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Acquired Companies hold or have the right to use all material assets, properties, rights and Contracts necessary for the operation of the Business as presently conducted.

(ii) Except as set forth on <u>Part 3.1(h)(ii)</u> of the Disclosure Schedule, there is no and, since January 1, 2010, there has never been any Company Owned Real Property.

(iii) <u>Part 3.1(h)(iii)</u> of the Disclosure Schedule sets forth a list of all real property leases (A) for business locations of the Acquired Companies where medical records are located, or (B) requiring annual payments of \$20,000 or more by the Acquired Companies (each a "<u>Company Lease</u>") and the address of the real property to which it relates. Each Company Lease

remains in full force and effect. No Acquired Company is in breach or default in any material respect under any Company Lease and, to the Company's Knowledge, no lessor thereunder is in breach or default in any material respect. True and complete copies of all Company Leases, together with all modifications, extensions, amendments and assignments thereof, if any, have been made available to Buyer. Except as set forth on <u>Part 3.1(h)(iii)</u> of the Disclosure Schedule, no Acquired Company has subleased or granted any other right to the use or occupy any real property that is the subject of any Company Lease to any Person other than an Acquired Company.

(i) <u>Taxes</u>. Except as set forth on <u>Part 3.1(i)</u> of the Disclosure Schedule:

(i) each Acquired Company has timely filed all income and other material Tax Returns required to be filed by it on or before the Closing Date (taking into account any applicable extensions) with the appropriate Taxing Authority, and has timely paid in full all Taxes shown to be due and payable by any Acquired Company thereon and has timely paid in full all material Taxes due and owing by such Acquired Company (whether or not shown on any Tax Return). Each such Tax Return is true, correct and complete in all material respects and were prepared in substantial compliance with all applicable Laws;

(ii) each Acquired Company has withheld and paid to the appropriate Taxing Authorities all material Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party, and all Forms W-2 and 1099 required with respect thereto have been properly completed in all material respects and timely filed;

(iii) the due and unpaid Taxes of the Acquired Companies for all Tax periods through the Balance Sheet Date do not exceed the accruals and reserves for Taxes set forth on the Balance Sheet;

(iv) no Acquired Company has received written notice from any Taxing Authority of a pending (and, to the Company's Knowledge, there is no threatened) claim, action, suit, proceeding, deficiency or investigation by a Taxing Authority for the assessment or collection of any Taxes with respect to any Acquired Company;

(v) there are no Encumbrances for Taxes against any Acquired Company's assets, other than Encumbrances that constitute a Permitted Encumbrance;

(vi) no Acquired Company has waived any statute of limitations in respect of Taxes or entered into any agreement extending the period for assessment or collection of any Taxes (except for any such waivers or extensions that have expired);

(vii) no Acquired Company is a party to any Income Tax allocation or sharing agreement;

(viii) all material Income Tax Returns filed by the Acquired Companies, examination reports, and statements of deficiencies assessed against or agreed to by the Acquired

Companies for Tax periods ending December 31, 2014, December 31, 2015 and December 31, 2016, have been made available to Buyer;

(ix) no Acquired Company is a party to any agreement, contract, arrangement, or plan that could result, separately or in the aggregate, in the payment of any "excess parachute payment" within the meaning of Code §280G(b)(1) (or any corresponding provision of state, local, or non-U.S. Tax law) in connection with the transactions contemplated by this Agreement. No Acquired Company has been a United States real property holding corporation within the meaning of Code §897 (c)(2) during the applicable period specified in Code §897(c)(1)(A)(ii). No Acquired Company has (A) been a member of an affiliated group, within the meaning of Code §1504(a) or any similar group defined under any similar provision of state, local or non-U.S. Tax law, filing a consolidated U.S. federal Income Tax Return (other than a group the common parent of which was the Company) or (B) any liability for the Taxes of any Person (other than the Acquired Companies) under Reg. §1.1502-6 (or any similar provision of state, local, or non-U.S. law), as a transferee or successor;

(x) within the past three (3) years, no Acquired Company has distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Code §355 or Code §361;

(xi) no Acquired Company is or has been a party to any "listed transaction," as defined in Code §6707A (c)(2) and Reg. §1.6011-4(b)(2);

(xii) no Acquired Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any:

(A) change in method of accounting for a taxable period ending on or prior to the Closing Date;

(B) "closing agreement" as described in Code §7121 (or any corresponding or similar provision of state, local, or non-U.S. income Tax law) executed on or prior to the Closing Date;

(C) installment sale or open transaction disposition made on or prior to the Closing Date;

- (D) prepaid amount received on or prior to the Closing Date;
- (E) election under Code §108(i);

(F) use of an improper method of accounting for a taxable period ending on or prior to the

Closing Date; or

(G) intercompany transactions or any excess loss account described in Treasury Regulations under Code §1502 (or any corresponding or similar provision of state, local, or non-U.S. income Tax law);

For the avoidance of doubt, the determination of whether an Acquired Company used an improper method of accounting for a taxable period ending on or prior to the Closing Date shall be determined without regard to any accounting method changes required as a result of the transactions provided for in this Agreement.

(xiii) no Acquired Company is currently the beneficiary of any extension of time within which to file any Tax Return;

(xiv) no written claim has been made within the past three (3) years by any Taxing Authority against an Acquired Company in a jurisdiction where such Acquired Company does not file Tax Returns that such Acquired Company is or may be subject to taxation by that jurisdiction;

(xv) no Acquired Company has received or, to the Knowledge of the Company is subject to, any letter ruling from the Internal Revenue Service (or any comparable ruling from any other Taxing Authority).

(j) <u>Absence of Certain Changes</u>. Since December 31, 2017, there has not occurred any Material Adverse Effect. Except as set forth on <u>Part 3.1(j)</u> of the Disclosure Schedule or as expressly required by this Agreement, since December 31, 2017, each Acquired Company has conducted its business in the ordinary course, consistent with past practice, in all material respects, and no Acquired Company has:

(i) made any dividend or distribution in respect of, or redemption of, any shares of Company Common Stock, Capital Stock or any equity ownership interests of the Subsidiaries, as applicable, except to another Acquired Company;

(ii) allowed any Encumbrance to be placed upon any of such Acquired Company's assets, other than Permitted Encumbrances;

(iii) cancelled or waived any claims with a potential value in excess of \$25,000;

(iv) terminated or canceled or waived any material right under any Company Contract, other than in the ordinary course of business consistent with past practice;

(v) acquired (by merger, consolidation or acquisition of stock or assets) any corporation, partnership or other business organization or division thereof or collection of assets constituting all or substantially all of a business or business unit;

(vi) sold, transferred or otherwise disposed of a material amount of its properties or assets, other than in the ordinary course of business consistent with past practice;

(vii) changed its accounting methods or principles theretofore adopted, except as required by GAAP or reflected in the Financial Statements;

(viii) experienced any damage, destruction or casualty loss (other than those covered by insurance) with respect to any of the assets or properties of such Acquired Company that, individually or in the aggregate with respect to all Acquired Companies, exceeds \$25,000;

(ix) made any material change in compensation paid or payable to any officer or employee expected to have total annual compensation exceeding \$150,000 per annum for the fiscal year ending December 31, 2018, other than any change in the ordinary course of business consistent with past practice, as required by the terms of an existing Benefit Plan or by applicable Law or in connection with a new hire;

(x) made any material change to its Benefit Plans or the benefits payable thereunder, other than in the ordinary course of business consistent with past practices, as required by applicable Law or the terms of such Benefit Plans;

(xi) experienced any work interruptions, material labor grievances or similar material claims filed;

(xii) incurred any Indebtedness (other than (1) borrowings under the Company's revolving line of credit in the ordinary course of business consistent with past practice, or (2) other ordinary course borrowings in an aggregate amount of no more than \$100,000), or guaranteed another Person's Indebtedness;

(xiii) made any equity investments in any other Person;

(xiv) made any capital expenditures in excess of \$25,000, in the aggregate;

(xv) made any material change in its customs or practices regarding cash management, collection of accounts receivable, payments of accounts payable, or cash distributions;

(xvi) made any material change in its practices with respect to timely payment of accounts payable or other obligations payable to vendors, suppliers or other third parties; or

(xvii) entered into any Contract or made any binding commitment to take any of the actions specified in this <u>Section 3.1(j)</u>.

(k) <u>Company Contracts</u>. <u>Part 3.1(k)</u> of the Disclosure Schedule sets forth a list of Contracts to which any Acquired Company is a party and which are in any one or more of the categories listed below (collectively, with the Core Service Contracts, the "<u>Company Contracts</u>"):

(i) any joint venture agreement, operating agreement, management agreement, cost sharing agreement, or partnership agreement (other than the Organizational Documents of any of the Acquired Companies);

(ii) any Contract related to an acquisition or divestiture of any corporation, partnership or other business organization or division thereof or collection of assets constituting all or substantially all of a business or business unit by any Acquired Company;

(iii) any Contract for the purchase, sale or lease of tangible personal property, in each case with future required scheduled payments of \$25,000 or more in any calendar year or \$50,000 in the aggregate;

(iv) any employment, severance, retention, deferred compensation or consulting agreement with (A) any officer, director or employee of any Acquired Company or (B) any independent contractor of any Acquired Company expected to have a total annual compensation in excess of \$150,000 for the fiscal year ending December 31, 2018;

(v) any Contract evidencing Indebtedness of such Company, or under which such Acquired Company has issued any note, bond, indenture, mortgage, security interest or other evidence of Indebtedness, or has directly or indirectly guaranteed Indebtedness (other than capitalized leases), liabilities or obligations of any Person;

(vi) any Contract containing non-competition, non-solicitation or other material restrictive covenants binding on any of the Acquired Companies or, to the Company's Knowledge, any of their officers (other than those in favor of the Acquired Companies);

(vii) each Contract with the Acquired Companies' top ten (10) vendors on a consolidated basis (based on the consolidated, aggregate dollar amount of purchases of vendor product by the Acquired Companies during the twelve months ending on the Balance Sheet Date);

(viii) any Contract other than of a type described in subsections (i) through (vii) of this <u>Section 3.1(k)</u> involving (A) future payments by such Acquired Company, (B) performance of services by such Acquired Company with a value, or (C) receipt of goods or services by such Acquired Company with a value, in each of the foregoing cases, in excess of \$750,000 on an annual basis;

(ix) any Contract providing for an exclusive relationship or the purchase from a supplier of all or substantially all of the requirements of any of the Acquired Companies of a particular product or service, including cell phone contracts, utilities, healthcare insurance, leases and the like;

(x) any Contract between any Acquired Company and any referral source or third party payor, or any immediate family member of any of the foregoing, including, without limitation, any Contract involving any remuneration, compensation, donation, sponsorship or other payments by or to an Acquired Company and the aforementioned parties, excluding, for the avoidance of doubt Contracts with any medical directors of the Acquired Companies to serve in their capacity as such;

(xi) any outstanding binding commitment to enter into any agreement of the type described in subsections (i) through (x) of this Section 3.1(k).

The Acquired Companies are parties to all hospice core service contracts required by the Medicare conditions of participation or conditions of payment (collectively, the "<u>Core Service Contracts</u>"). True, correct and complete copies of the Company Contracts have been made available to Buyer

(with the exception of the Core Service Contracts, which are to be provided to Buyer pursuant to <u>Section 4.14(i)</u>). Except as set forth on <u>Part 3.1(k)</u> of the Disclosure Schedule, (i) each Company Contract is in full force and effect, (ii) no Acquired Company is in breach of or default under any Company Contract in any material respect, and (iii) to the Knowledge of the Company, no counterparty is in breach of or default under any Company Contract in any material respect.

## (l) Employees and Compensation.

(i) Except as set forth on <u>Part 3.1(1)</u> of the Disclosure Schedule, no Acquired Company is or, since the Look-Back Date, has been, subject to any material labor dispute, labor-related arbitration, labor-related lawsuit or labor-related administrative proceeding, no such labor dispute, labor-related arbitration, labor-related lawsuit or labor-related administrative proceeding is pending, and, to the Knowledge of the Company, none is threatened by any Governmental Entity, and none is threatened in writing by any Person other than any Governmental Entity. None of the Acquired Companies' employees are represented by any labor union nor are any collective bargaining agreements otherwise in effect with respect to such employees in connection with their employment by any of the Acquired Companies and no union organizing activities involving such employees are pending or, to the Knowledge of the Company, threatened.

(ii) The Company has made available to Buyer a true and correct list of the name, state of residence, title, employer, date of hire, hourly pay rate or annual salary, fixed bonuses, and discretionary bonuses of each officer, employee or independent contractor of the Acquired Companies employed or engaged by any of the Acquired Companies as of October 7, 2018. The Company has also made available a list of each employee who has notified an Acquired Company in writing as of the date hereof that he or she is not fully available to perform his or her duties as a result of disability leave or other leave and sets forth the basis of such leave and the anticipated date of return to full service. Except (A) as provided by law, (B) as set forth on Part 3.1 (1) of the Disclosure Schedule or (C) pursuant to a contract, agreement, or plan made available pursuant to clause 3.1(1)(ii), the employment or engagement, as applicable, of each officer or employee of the Acquired Companies is "at will" and may be terminated without any severance liability of the Acquired Companies.

(iii) Except as set forth on <u>Part 3.1(1)(iii)</u> of the Disclosure Schedule, none of the Acquired Companies have misclassified any employees of the Acquired Companies as independent contractors pursuant to the applicable United States Department of Labor's and Internal Revenue Service's regulations.

(iv) No Acquired Company is delinquent in payments to any employees or independent contractors for any wages, salaries, bonuses, benefits or other compensation for services performed by them or any amounts required to be paid any employee or independent contractor for any post-employment or post-engagement obligations, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(v) Each Acquired Company is and has been, since the Look-Back Date, in compliance with its obligations pursuant to the Worker Adjustment Retraining and Notification Act, 29 U.S.C. § 2101 et seq. (as amended from time to time, "<u>WARN</u>" and, collectively with any

similar state or local law, the "<u>WARN Acts</u>") and with all other notification obligations arising under any statute or otherwise, in each case to the extent affecting, in whole or in part, any site of employment, facility, operating unit or employee of the Acquired Companies. Since the Look-Back Date, the Acquired Companies have not engaged in any transaction or engaged in layoffs, terminations or relocations sufficient in number to trigger any WARN Act obligation. To the Company's Knowledge, no former employees of the Acquired Companies have suffered an "employment loss" (as defined in WARN) in the ninety (90) days prior to the date hereof.

# (m) Employee Benefit Plans.

(i) <u>Part 3.1(m)</u> of the Disclosure Schedule sets forth a list of each material Benefit Plan. The Company has made available to Buyer, with respect to each Benefit Plan, true and complete copies of: (i) the current plan documents and any amendments thereto; (ii) the current summary plan descriptions and all summaries of material modifications thereto; (iii) the most recent annual report or financial statements, if applicable, with respect to such Benefit Plan. Each Benefit Plan has been established and administered in all material respects in accordance with its terms and applicable Law and Legal Requirements, including, as to each Benefit Plan that is subject to United States Law, ERISA or the Code.

(ii) Except as set forth on <u>Part 3.1(m)</u> of the Disclosure Schedule, the consummation of the transactions contemplated by this Agreement will not, in and of itself, (a) entitle any current or former employee, officer or independent contractor of any Acquired Company to severance pay, unemployment compensation or any other payment, or (b) accelerate the time of payment or vesting, or increase the amount of compensation due any such employee, officer or independent contractor.

(iii) Each pension plan that is intended to qualify under Section 401(a) of the Code (the "<u>Qualified Plans</u>"), and each trust funding a Benefit Plan which is intended to be exempt from U.S. federal income tax under Section 501(c)(9) of the Code, has received a favorable determination, opinion or advisory letter from the Internal Revenue Service and, to the Knowledge of the Company, nothing has occurred subsequent to the issuance of such determination, opinion or advisory letter which would reasonably be expected to cause such Qualified Plan or trust to lose its qualified status. Except as set forth on <u>Part 3.1(m)</u> of the Disclosure Schedule, none of the Acquired Companies, nor any of their respective ERISA Affiliates, either currently or at any time within the six (6) years prior to the date hereof, sponsored, maintained, or were otherwise obligated to contribute to a "multiemployer plan" as defined in Section 4001(a)(3) of ERISA, or any pension plan subject to Section 501(c)(9) of the Code, satisfies the requirements of that Section and has received a favorable determination letter from the IRS regarding that exempt status and has not, since receipt of the most recent favorable determination letter, been amended or operated in a way that would reasonably be expected to adversely affect that exempt status.

(iv) No Acquired Company has engaged in any non-exempt "prohibited transaction" (as defined in Section 4975 of the Code or Section 406 of ERISA) with respect to any Benefit Plan that would be reasonably likely to subject any Acquired Company any material Tax or penalty (civil or otherwise) imposed by ERISA, the Code or other applicable Law.

(v) No audits, inquiries, proceedings or administrative actions are pending or, to the Knowledge of the Company, threatened by a Governmental Entity with respect to any Benefit Plan.

(vi) All contributions, premiums or other payments for the Benefit Plans listed on <u>Part 3.1(m)</u> of the Disclosure Schedule attributable to all periods prior to the Closing Date have been made, in all material respects, on a timely basis and within the time limits prescribed by ERISA or other applicable Law or have been corrected, in all material respects, in accordance with the Department of Labor's guidance under the Voluntary Fiduciary Correction Program or other applicable correction program.

(vii) Each Benefit Plan that is in whole or in part a "nonqualified deferred compensation plan" (as such term is defined in Section 409A(d)(1) of the Code) complies in form, and has been maintained and operated in compliance, with Section 409A of the Code and all guidance and regulations issued thereunder. No director, officer, employee or consultant of the Acquired Companies is entitled to receive any gross-up or additional payment by reason of the tax required by Section 409A of the Code being imposed on such person.

(viii) With respect to each Benefit Plan that is a group health plan, each Acquired Company has complied, in all material respects, with all applicable continuation coverage requirements of the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("<u>COBRA</u>").

# (n) Compliance With Laws; Permits.

(i) Each Acquired Company operates and since the Look-Back Date has operated its business in compliance in all material respects with applicable Laws and Legal Requirements, including, without limitation, Laws and Legal Requirements respecting employment and employment practices, immigration, terms and conditions of employment, wages and hours, overtime pay, hiring practices, parental and family leave and pay, workplace health and safety, and Privacy Laws, and since the Look-Back Date, to the Knowledge of the Company, no Acquired Company has received notice of any Action by any Governmental Entity alleging material liability under, or material noncompliance with, any such Law or Legal Requirement. The Company makes no representation with respect to compliance with Healthcare Laws in this <u>Section 3.1(n)</u>.

(ii) Except as set forth on <u>Part 3.1(n)(ii)</u> of the Disclosure Schedule, each Acquired Company possesses all material approvals, permits, licenses, accreditations, certificates of need, provider numbers and provider agreements, qualifications, certifications and other authorizations required and/or issued by any Governmental Entity or third party payor (collectively, "<u>Permits</u>") required to conduct its business as it is currently conducted, and such Permits are valid and in full force and effect. Each Acquired Company is and, since the Look-Back Date, has been, in material compliance with the terms and conditions of such Permits.

Since the Look-Back Date, to the Knowledge of the Company, the Acquired Companies have not received any notice from a Governmental Entity of its intention to cancel, terminate, restrict, limit or otherwise qualify or not renew any of the Permits. The Company makes no representation with respect to Permits required by applicable Healthcare Laws in this Section 3.1(n).

# (o) Intellectual Property.

(i) <u>Part 3.1(o)</u> of the Disclosure Schedule identifies (A) each Acquired Company's registered tradenames, registered service marks, patents and registered copyrights, and any applications for the foregoing, and Internet domain names and email domain names, and (B) any license pursuant to which Company Intellectual Property is licensed to any Acquired Company by another Person (excluding license agreements pertaining to commercial off-the-shelf computer software products) or licensed to another Person by any Acquired Company. The Company Intellectual Property includes all Intellectual Property used or held for use by the Acquired Companies in, or necessary to, the conduct of their businesses as currently conducted in all material respects. The Acquired Companies either own or have valid licenses or rights to use all of the Company's business, to the Knowledge of the Company, does not infringe or otherwise violate any other Person's Intellectual Property, (X) there is no claim pending or, to the Knowledge of the Company, currently threatened in writing against any Acquired Company that the conduct of such Acquired Company's business infringes or otherwise violates any other Person's Intellectual Property, and (Z) no claims are pending or, to the Knowledge of the Company, threatened in writing against any Acquired Company that another Person is infringing or otherwise violating any Company Intellectual Property, and (Z) no claims are pending or, to the Knowledge of the Company, threatened in writing against any Person by any Acquired Company that another Person is infringing or otherwise violating any Company Intellectual Property. No Acquired Company has granted to any Person exclusive rights to any of the Company Intellectual Property.

(ii) The Acquired Companies take commercially reasonable measures to protect, maintain and enforce the Company Intellectual Property.

(iii) The IT Systems (A) operate as necessary for the conduct of the business of the Acquired Companies, and (B) to the Company's Knowledge, do not contain any "malware" that would reasonably be expected to interfere with the ability of the Buyer to operate such business post-closing as currently conducted. Since the Look-Back Date, there have been no material adverse events affecting the IT Systems that have caused a material impact on any Acquired Company or its respective business operations. The Acquired Companies have implemented, maintains, and complies with commercially reasonable business continuity and backup and disaster recovery plans and procedures with respect to the IT Systems.

(p) <u>Brokers' Fees</u>. No Acquired Company has dealt with any broker, finder or similar agent with respect to the transactions contemplated by this Agreement, and no Acquired Company is under any obligation to pay any broker's fee, finder's fee or commission in connection with the consummation of the transactions contemplated by this Agreement as a result of any agreement of any Acquired Company.

(q) <u>Affiliate Transactions</u>. Except as set forth on <u>Part 3.1(q)</u> of the Disclosure Schedule and for employment relationships, Benefit Plans and the payment of compensation and benefits in the ordinary course of business consistent with past practice, (i) since December 31, 2017, there have been no material transactions, and (ii) there are no Contracts, in each case between the Sellers or any of their Affiliates (other than the Acquired Companies), on the one hand, and any

Acquired Company, on the other hand. Any such Contracts will be terminated at or prior to the Closing except as otherwise set forth on Part 3.1(q) of the Disclosure Schedule.

(r) <u>Insurance</u>. Each Acquired Company is insured under the insurance policies listed on <u>Part 3.1(r)</u> of the Disclosure Schedule, true and correct copies of which have been made available to Buyer. The Acquired Companies have made available to Buyer true and correct copies of all insurance loss runs and worker's compensation claims for the Acquired Companies for the most recently ended three (3) policy years as of the date of this Agreement. All such insurance policies are in full force and effect, are valid and enforceable, and all premiums currently due thereunder have been paid. No Acquired Company has received any notice of cancellation or material modification in coverage amounts of any such insurance policies (except any generally applicable modification not specific to the Acquired Companies). Further, no material insurance carried by the Acquired Companies has been canceled by an insurer, and no Acquired Company has been denied any material coverage. Except as disclosed on <u>Part 3.1(r)</u> of the Disclosure Schedule, there are no pending material claims under any such insurance policy as to which the respective insurers have denied coverage. To the Knowledge of the Company, no Acquired Company has received any written notice of any material increase (or proposed material increase) after the date hereof in any deductibles, retained amounts or premiums payable under the insurance policies pursuant to which each Acquired Company is insured that is disproportionate to other increases occurring in the market, including annual increases in the ordinary course of business.

## (s) Environmental Matters.

(i) Each Acquired Company is and has been since the Look-Back Date in compliance in all material respects with all Environmental Laws applicable to its properties and is and has been since the Look-Back Date in compliance in all material respects with all Environmental Permits required for the operation of its business as it is has been conducted.

(ii) There has been no Release of, or human exposure to, any Hazardous Materials that has had or would reasonably be expected to result in material liability to the Acquired Companies under Environmental Laws (other than such matters covered or barred by workers' compensation and related insurance).

(iii) Since the Look-Back Date, to the Knowledge of the Company, no Acquired Company has received any notice, information request, demand, claim, or notice of violation concerning any violation or alleged violation of, or any actual or alleged liability of any Acquired Company arising under or related to, any applicable Environmental Law (other than those resolved without any ongoing costs or obligations), which notice, information request, demand or claim is written in the case of notice from a Person that is not a Governmental Entity.

# (t) <u>Healthcare Matters</u>.

(i) Except as set forth on <u>Part 3.1(t)(i)</u> of the Disclosure Schedule: (A) the business of the Acquired Companies is and since the Look-Back Date, has been at all times conducted in all material respects in compliance with all Healthcare Laws, and (B) since the Look-Back Date, to the Knowledge of the Company, no Acquired Company has received any notice from any

Governmental Entity regarding any actual or alleged violation of, or failure to comply with, any Healthcare Laws applicable to the Company or by which any properties or assets owned or used by the Company are bound or affected. The Acquired Companies maintain a compliance program consistent in all material respects with the criteria established by the Federal Sentencing Guidelines and the guidance of the Office of Inspector General of the Department of Health and Human Services, and the Acquired Companies are and have, since the Look-Back Date, been operated in compliance in all material respects with their compliance program.

(ii) Except as set forth on <u>Part 3.1(t)(ii)</u> of the Disclosure Schedule: (A) the Acquired Companies possess all Permits required by applicable Healthcare Laws necessary for the operation of the Acquired Companies, which Permits are set forth on <u>Part 3.1(t)(ii)</u> of the Disclosure Schedule; (B) the Acquired Companies are and have been since the Look-Back Date in material compliance with such Permits, and all of such Permits are valid, in good standing and in full force and effect; (C) there is no Action by or before any Governmental Entity pending, or, to the Company's Knowledge, threatened against any Acquired Company to revoke, withdraw, suspend, cancel, terminate or otherwise limit any such Permit; and (D) to the Knowledge of the Company, no Acquired Company has received any notice from any Governmental Entity regarding any violation of any such Permit or any revocation, withdrawal, suspension, cancellation or termination of any such Permit. Since the Look-Back Date, the Acquired Companies have filed all material reports and maintained and retained in all material respects all records required by applicable Healthcare Laws.

(iii) Except as set forth on <u>Part 3.1(t)(iii)</u> of the Disclosure Schedule: (A) all billing practices (including, without limitation, billing, coding, filing, and claims practices, and the related reports and filings, including cost and cap reports) of the Acquired Companies are and, since the Look-Back Date, have been in compliance in all material respects with applicable Healthcare Laws; (B) each of the Acquired Companies has paid or caused to be paid all known and undisputed material refunds, overpayments, discounts or adjustments, which have become due, since the Look-Back Date, including but not limited to any such obligation relating to hospice Medicare cap liabilities and/or the "60-day" rule; (C) none of the Acquired Companies has any material reimbursement, payment or payment rate appeals, disputes or contested positions pending before any Governmental Entity and, to the Knowledge of the Company, none are threatened; (D) none of the Acquired Companies has claimed or received reimbursements from Government Programs or Private Programs in material excess of amounts permitted by applicable Healthcare Laws; and (E) the right of the Acquired Companies to receive reimbursements pursuant to any Government Program or Private Program has not been terminated, rescinded, suspended, subject to pre-payment or post-payment review, or otherwise materially adversely affected as a result of any Action by a Governmental Entity or Private Program.

(iv) The Acquired Companies, as applicable, have timely filed all cost and cap reports required to be filed on or prior to the date hereof in accordance with applicable Government Program requirements. All cost and cap reports submitted by the Acquired Companies to Government Programs have been prepared in all material respects in accordance with and in compliance with applicable Healthcare Laws and are complete and accurate in all material respects. Except as set forth on <u>Part 3.1</u> (t)(iv) of the Disclosure Schedule, since the Look-Back Date, to the

Knowledge of the Company, none of the Acquired Companies has received notice of any dispute from any Governmental Entity regarding such cost or cap reports, in each case other than with respect to adjustments in the ordinary course of business. Except as set forth on Part 3.1(t)(iv) of the Disclosure Schedule, since the Look-Back Date, none of the Acquired Companies has received written notice of audits or actual, alleged, possible or potential violations with respect to any cost or cap reports, and, to the Knowledge of the Company, no such audits are threatened.

(v) Neither any Acquired Company nor any of their directors, officers or managing employees has been or is currently suspended, excluded or debarred from contracting with the United States federal or any state government or from participating in any Federal Health Care Program (as defined in 42 USC § 1320a-7b(f)) or, to the Knowledge of the Company, is subject to an investigation or proceeding by any Governmental Entity that has resulted in or would reasonably be expected to result in such suspension, exclusion, or debarment; nor has any Acquired Company, or to the Knowledge of the Company, or any of their directors, officers or managing employees received notice of any impending or potential exclusion listing. No Acquired Company has been sanctioned pursuant to 15 U.S.C. § 41 et seq. or 42 U.S.C. § 1320a-7a or 1320a-8, or been charged with or convicted of a crime described at 42 U.S.C. § 1320a-7b, and no such sanction or proceeding is pending or, to the Company's Knowledge, threatened. Since the Look-Back Date, to the Knowledge of the Company, no employee of the Acquired Companies nor any independent contractor or vendor has been or is currently suspended, excluded or debarred from contracting with the United States federal or any state government or from participating in any Federal Health Care Program (as defined in 42 USC § 1320a-7b(f)) or, to the Knowledge of the Company, is subject to an investigation or proceeding by any Governmental Entity that has resulted in or would reasonably be expected to result in such suspension, exclusion, or debarment.

(vi) Neither any Acquired Company nor any of their directors, officers, employees, or, to the Knowledge of the Company, their contractors or agents, while acting on behalf of the Acquired Companies, have, solicited, received, paid or offered to pay any remuneration, directly or indirectly, overtly or covertly, in cash or in kind, for any referral in violation of any applicable Healthcare Law.

(vii) Except as set forth on <u>Part 3.1(t)(vii)</u> of the Disclosure Schedule, no Acquired Company, nor, to the Knowledge of the Company, any of their respective directors, officers, employees, contractors or agents, while acting on behalf of the Acquired Companies, are or, since the Look-Back Date, have been: (A) a party to or otherwise subject to a civil investigative demand, subpoena, corporate integrity or any other consent decree, judgment, order, deferred or non-prosecution agreement, settlement or similar agreement with a Governmental Entity, (B) to the Company's Knowledge, the subject of any investigation, program integrity review or audit conducted by any federal, state or local Governmental Entity or agent thereof, or any Private Program to the extent applicable, (C) to the Company's Knowledge, a defendant or named party in any qui tam/False Claims Act litigation, (D) subject to any mandatory or discretionary exclusion or suspension from Federal program participation, or (E) been the subject of any investigation, program integrity review, audit or survey conducted by any federal, state or local Governmental Entity or trade association, professional review organization, accrediting organization, licensing or certifying agency that resulted in a finding of any alleged improper activity or the Acquired Company's receipt

of either a notice of deficiency or other notice of any type of violation of any Healthcare Law. The Company has made available to Buyer true, correct and complete copies of each healthcare-related survey of the Acquired Companies conducted in the preceding three (3) years.

(viii) Except as set forth on Part 3.1(t)(viii) of the Disclosure Schedule, no Acquired Company has since the Look-Back Date: (A) had any security or data breaches compromising or otherwise involving Protected Health Information, as that term is defined in HIPAA, that required notification under 45 C.F.R. § 164.406 or 45 C.F.R. § 164.408(b); (B) received any claim or notice from any Governmental Entity or agent thereof alleging or referencing a violation of HIPAA or the investigation of any breach or the improper use, disclosure or access to any Protected Health Information in its possession, custody or control that has not been resolved; (C) to the Knowledge of the Company, received any communication from any Governmental Entity alleging that any of the Acquired Companies are not in compliance with HIPAA that has not been resolved; and (D) had occur any event that required it to provide notification to any Governmental Entity under any state privacy and/or breach notification Laws. The Acquired Companies, as applicable, have established and implemented such policies, programs, procedures, contracts and systems as are necessary to comply with HIPAA in all material respects. Without limitation of the foregoing, each Acquired Company is and has been since the Look-Back Date a lawful party to or beneficiary of a business associate agreement with each third party providing claims review, claims audit and related services on behalf of the Acquired Companies, including such third party(ies) situated in the Philippines, and all such arrangements by and between the Acquired Companies and, to the Company's Knowledge, those third parties providing such claims review, claims audit and related services are and have been at all relevant times in compliance with applicable Laws in all material respects, including, without limitation, HIPAA, HITECH and any and all corresponding state privacy laws.

(ix) Except as set forth on <u>Part 3.1(t)(ix)</u> of the Disclosure Schedule, none of the Acquired Companies, or any predecessor from which it acquired a provider number, has since the Look-Back Date conducted any home health operations, provided any skilled or non-skilled home health services, or participated in, or received any funds from, any Government Programs with respect to the provision of any home health services.

(x) Except as set forth on Part 3.1(t)(x) of the Disclosure Schedule, (A) no referral source of the Acquired Companies maintains an ownership interest in, or compensation arrangement with, the Acquired Companies; and (B) (x) none of the Acquired Companies has entered into any joint venture, partnership, co-ownership or other financial arrangement involving any ownership, lease or investment interest in or by the Acquired Companies with an individual known by it to be a physician or an immediate family member of a physician; and (y) no physician who has, or whose immediate family member has, a financial relationship (as such terms are defined in the Stark Law), with any Acquired Company directly or indirectly makes (or has made) referrals, as that term is defined in the Stark Law, to any Acquired Company or any predecessor business, in each case in (x) and (y) above, without complying with an applicable exception from the Stark Law's referral prohibition as set forth on Part 3.1 (t)(x) of the Disclosure Schedule.

(xi) All employees and independent contractors of the Acquired Companies possess the necessary licenses, certificates and qualifications, as required by the applicable Healthcare Laws, to carry out their respective duties to the Acquired Companies, with such exceptions as are not material to the Acquired Companies taken as a whole. All employees and independent contractors of the Acquired Companies required to be licensed, certified, registered, accredited, or similarly approved by a Governmental Entity are in compliance with such licensing requirements and do not engage in activities subject to licensure or such other regulatory approval in jurisdictions in which such license or other regulatory approval is required but has not been obtained and maintained, with such exceptions as are not material to the Acquired Companies taken as a whole. The Acquired Companies audit the license status of all employees and independent contractors on a monthly basis and remove any such employees or independent contractors from active services if such person's license is not in good standing or has expired.

(xii) The Acquired Companies are not relying on any exemption from or deferral of any Healthcare Laws (other than applicable exceptions from the Stark Law's referral prohibition as set forth on Part 3.1(t)(x) of the Disclosure Schedule).

(xiii) The hospice patients of the Acquired Companies, currently and since the Look-Back date, satisfy and have satisfied, as applicable, all of the criteria for hospice appropriateness, eligibility and billing under the rules and regulations of the Government Programs and the requirements of the Private Programs, with such exceptions as are not material to the Acquired Companies, taken as a whole. None of the Acquired Companies has made any material claims to, or received material reimbursements from, the Government Programs or any Private Program (A) for patients who have not been certified as terminally ill in accordance with 42 C.F.R. § 418.22; (B) for services which are not reasonable and necessary for the palliation and management of the terminal illness and related conditions; (C) for services which do not comply with the plan of care established for the patient and periodically reviewed by the attending physician, the medical director, and the interdisciplinary group of the hospice program as set forth in 42 C.F.R. § 418.56; (D) for an item or service that was either not provided as claimed or false or fraudulent; (E) for a patient that has not elected hospice care in accordance with 42 C.F.R. § 418.24; and (F) for amounts in excess of the amounts permitted by applicable Laws or the Government Programs or Private Programs, including without limitation any amounts in excess of the applicable Medicare hospice cap, with such exceptions for clauses (A) through (F) of this <u>Section 3.1</u> (t)(xiii) as are not material to the Acquired Companies, taken as a whole.

(xiv) The Company has made available to Buyer the patient census of each provider number of the Acquired Companies as of August 23, 2018 (to be updated by the Acquired Companies as of Closing), and shows by provider number the start of care date, the last billing date, the residence, and the level of service for each patient; provided, that such census does not contain any Personal Information of any such patient.

(xv) Each of the Acquired Companies has and has had at all times since the Look-Back Date effective and enforceable written contracts with all skilled nursing facilities in which its patients reside, with such exceptions as would not give rise to a material liability of the Acquired Companies.

(xvi) The Acquired Companies have (i) verified, or in the case of independent contractors, received verification from such independent contractors with respect to, the credentials of employees and independent contractors providing patient care or other services reimbursable under Government Programs, (ii) conducted criminal background checks on all such employees and independent contractors, and (iii) screened all such employees and independent contractors under the List of Excluded Individuals/Entities. The Acquired Companies and their respective employees and independent contractors supporting their respective business operations satisfy all applicable conditions of participation for the provision of services by the Agencies.

(xvii) All specialty programs currently utilized by the Acquired Companies or utilized by the Acquired Companies since the Look-Back Date, including, but not limited to, programs or protocols targeting specific diseases states, are set forth on Part 3.1(t)(xviii) of the Disclosure Schedule.

(xviii) The Acquired Companies are and have been at all times in compliance with the terms and requirements of the Corporate Integrity Agreement, including, without limitation, having timely made all reports, repayments and filings with the Office of Inspector General required thereby.

(xix) The Acquired Companies have submitted bills to Medicare and all other payors for all patients receiving hospice services from the Acquired Companies in accordance with the sequential billing requirements of the related Government Programs and Private Programs, with such exceptions as would not give rise to a material liability of the Acquired Companies or materially interfere with, disrupt or delay the billing or collection activities of the Acquired Companies after the Closing.

# (u) <u>Miscellaneous</u>.

(i) <u>Part 3.1(u)(i)</u> of the Disclosure Schedule contains a list of the members of the Board of Directors, managers and officers of each Acquired Company.

(ii) <u>Part 3.1(u)(ii)</u> of the Disclosure Schedule contains a list of each bank or other financial institution in which a Acquired Company has an account, safe deposit box or lock box arrangement, the name of the Acquired Company in whose name such account, box or arrangement is held, the identifying numbers or symbols of the account, box or arrangement, and the name of each person authorized to draw thereon or to have access thereto.

(iii) <u>Part 3.1(u)(iii)</u> of the Disclosure Schedule contains a list of any assets owned by the Acquired Companies constituting negotiable instruments, promissory notes, tangible chattel paper and documents of title.

(iv) The Acquired Companies do not have any liabilities or obligations of any nature (whether known or unknown and whether absolute, accrued, contingent or otherwise) relating to the operation of their respective businesses, as applicable, that are required under GAAP to be accrued on the face of a balance sheet, except for (i) current liabilities incurred in the ordinary course of business, consistent with past practice, since the Balance Sheet Date, (ii) expenses incurred

in connection with the transactions contemplated by this Agreement, (iii) liabilities set forth in the Balance Sheet, and (iv) liabilities set forth on Part 3.1(u)(iv) of the Disclosure Schedule.

(v) The Company does not hold any direct or indirect ownership interest in the New York LLC or any Entity having hospice operations in the State of New York. There are not any agreements between CCH and the New York LLC.

(vi) True and complete copies of all records of quarterly meetings of the Boards of Directors of the Acquired Companies have been made available for inspection by Buyer.

(v) Disclaimer of Warranties. EXCEPT AS SPECIFICALLY SET FORTH IN THIS SECTION 3.1 AND IN SECTION 3.3, NONE OF THE ACQUIRED COMPANIES OR ANY OF THEIR RESPECTIVE DIRECTORS, AFFILIATES, MANAGERS, MEMBERS, OFFICERS. SECURITYHOLDERS, PARTNERS, EMPLOYEES, CONSULTANTS, AGENTS, COUNSEL OR ADVISORS MAKES OR HAS MADE ANY REPRESENTATION OR WARRANTY OF ANY KIND OR NATURE WHATSOEVER, ORAL OR WRITTEN, EXPRESS OR IMPLIED, TO BUYER OR ANY OF ITS AFFILIATES OR FINANCING SOURCES. ANY REPRESENTATIONS AND WARRANTIES NOT SPECIFICALLY SET FORTH IN THIS SECTION 3.1 OR IN SECTION 3.3, WHETHER EXPRESS OR IMPLIED (INCLUDING ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, USAGE, SUITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSES OR NON-INFRINGEMENT), ARE DISCLAIMED BY THE COMPANY AND EACH OTHER ACQUIRED COMPANY. EXCEPT AS SET FORTH IN THIS SECTION 3.1 AND IN SECTION 3.3, THE COMPANY AND EACH OTHER ACQUIRED COMPANY SPECIFICALLY DISCLAIM ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY, USAGE, SUITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE WITH RESPECT TO THE ASSETS OF THE ACQUIRED COMPANIES, ANY PART THEREOF, THE WORKMANSHIP THEREOF, AND THE ABSENCE OF ANY DEFECTS THEREIN, WHETHER LATENT OR PATENT, IT BEING UNDERSTOOD THAT EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS SECTION 3.1 AND IN SECTION 3.3 (AS QUALIFIED BY THE APPLICABLE PART OF THE DISCLOSURE SCHEDULE) SUCH SUBJECT ASSETS ARE "AS IS, WHERE IS" ON THE CLOSING DATE, AND IN THEIR PRESENT CONDITION, AND BUYER SHALL RELY ON ITS OWN EXAMINATION AND INVESTIGATION THEREOF AS WELL AS THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS SECTION 3.1 AND IN SECTION 3.3. THE COMPANY, AND THE OTHER ACOUIRED COMPANIES, ARE NOT, DIRECTLY OR INDIRECTLY, MAKING ANY REPRESENTATIONS OR WARRANTIES REGARDING THE PRO-FORMA FINANCIAL INFORMATION, FINANCIAL PROJECTIONS OR OTHER FORWARD-LOOKING STATEMENTS OF THE ACQUIRED COMPANIES.

(w) <u>Reliance</u>. The Company acknowledges that none of Buyer, Amedisys or any other Person has made any representation or warranty, express or implied, written or oral, as to the accuracy or completeness of any information that Buyer or Amedisys have furnished or made available to the Company or its representatives, other than as explicitly set forth in <u>Section 3.2</u> and <u>Section 3.4</u>, and none of Buyer, Amedisys or any other Person (including any representative of Buyer or Amedisys) shall have or be subject to any liability to the Company, resulting from its use

of any such information, documents or material made available to the Company or its representatives in any "data rooms," management presentations, due diligence or in any other form in expectation of the transactions contemplated herein that is not explicitly set forth in <u>Section 3.2</u> or <u>Section 3.4</u>. The Company acknowledges that, except for the representations and warranties contained in <u>Section 3.2</u> or <u>Section 3.4</u>, none of Buyer, Amedisys nor any other Person has made, and the Company has not relied on, any other representation or warranty, express or implied, written or oral, by or on behalf of Buyer or Amedisys. The Company acknowledges that neither it nor any other Person, directly or indirectly, has made, and it has not relied on, any representation or warranty regarding the pro-forma financial information, financial projections or other forward-looking statements of Buyer or Amedisys.

**3.2** <u>Representations and Warranties of the Buyer</u>. Buyer represents and warrants to the Acquired Companies and the Sellers that each of the statements contained in this <u>Section 3.2</u> is true and correct as of the date hereof.

(a) <u>Due Organization and Good Standing</u>. Buyer is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Louisiana, with all requisite power and authority to own its properties and to carry on its business as such business is now conducted.

(b) <u>Authorization and Execution</u>. Buyer has the power and authority and has taken all required action on its part necessary to permit it to execute and deliver and to carry out the terms of this Agreement and the other agreements, instruments and documents of Buyer contemplated hereby. This Agreement has been, and each of such other agreement, instrument and document to which it is a party will be at or prior to the Closing, duly and validly executed and delivered by Buyer, and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and such other agreements, instruments and documents, when so executed and delivered will constitute, the legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

(c) <u>Governmental Filings</u>. No Governmental Filings are required in connection with the execution, delivery and performance by Buyer of this Agreement or the other agreements, instruments or documents contemplated hereby, or the consummation by Buyer of the transactions contemplated hereby or thereby, as applicable, except (i) Governmental Filings that become applicable as a result of matters specifically related to the Company, (ii) Governmental Filings set forth on <u>Part 3.2(c)</u> of Buyer's Disclosure Schedule or (iii) such other Governmental Filings the failure of which to be obtained or made would not materially impair the ability of Buyer to consummate the transactions contemplated hereby.

(d) <u>No Conflict or Violation</u>. Except as set forth on <u>Part 3.2(d)</u> of Buyer's Disclosure Schedule, the execution, delivery and performance by Buyer of this Agreement and the other agreements, instruments and documents contemplated hereby, and the consummation by Buyer

of the transactions contemplated hereby and thereby do not (i) assuming all authorizations, consents and approvals described or referred to in <u>Section 3.2(c)</u> have been obtained or made, violate any applicable Law or Legal Requirements to which Buyer is subject, (ii) require a consent, approval or notification under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate or cancel any Contract to which Buyer is a party, or (iii) violate the Organizational Documents of Buyer, except with respect to the foregoing clauses (i) and (ii) as would not, individually or in the aggregate, impair the ability of Buyer to consummate the transactions contemplated hereby.

(e) <u>Legal Proceedings</u>. There are no Actions pending or, to the Knowledge of Buyer, threatened which challenge the validity or enforceability of this Agreement or seek to enjoin or prohibit consummation of the transactions contemplated hereby. Buyer is not subject to any judgment, decree, injunction or order of any Governmental Entity which would materially impair its ability to consummate the transactions contemplated hereby.

(f) <u>Brokers' Fees</u>. Buyer has not dealt with any broker, finder or similar agent with respect to the transactions contemplated by this Agreement, and is not under any obligation to pay any broker's fee, finder's fee, commission or similar amount in connection with the consummation of the transactions contemplated by this Agreement.

(g) <u>Financial Ability</u>. Buyer has immediately available funds or funds available under its credit facility sufficient to consummate the transactions contemplated by this Agreement, including (i) the payment of the Purchase Price, (ii) the funding of all of Buyer's costs and expenses arising in connection therewith, and (iii) any capital reserves required by any applicable Governmental Authority to be maintained. There are no bankruptcy, reorganization, or arrangement proceedings pending against, being contemplated by, or threatened against Buyer.

(h) <u>Solvency</u>. Immediately after the consummation of the Closing (including, without limitation, after the payment in full of the Purchase Price), each Acquired Company will be Solvent, except to the extent any failure of each Acquired Company to be solvent results from a breach of the Sellers' or the Company's representations and warranties.

(i) <u>Reliance</u>. Buyer acknowledges that none of the Acquired Companies, the Sellers or any other Person has made any representation or warranty, express or implied, written or oral, as to the accuracy or completeness of any information that the Acquired Companies or the Sellers have furnished or made available to Buyer, Amedisys or their representatives, other than as explicitly set forth in <u>Section 3.1</u> and <u>Section 3.3</u>, and none of the Acquired Companies, the Sellers or any other Person (including any representative of the Acquired Companies or the Sellers) shall have or be subject to any liability to Buyer, Amedisys, its Affiliates, or any other Person, resulting from Buyer's or Amedisys' use of any such information, documents or material made available to Buyer, Amedisys or their representatives in any "data rooms," management presentations, due diligence or in any other form in expectation of the transactions contemplated herein that is not explicitly set forth in <u>Section 3.1</u> or <u>Section 3.3</u>. Buyer acknowledges that, except for the representations and warranties contained in <u>Section 3.1</u> or <u>Section 3.3</u>, none of the Sellers, the Company nor any other Person has made, and Buyer has not relied on, any other representation or warranty, express or implied, written or oral, by or on behalf of the Sellers or the Company. Buyer

acknowledges that neither the Sellers, the Company nor any other Person, directly or indirectly, has made, and Buyer has not relied on, any representation or warranty regarding the pro-forma financial information, financial projections or other forward-looking statements of the Acquired Companies.

(j) <u>No Other Negotiations.</u> As of the date of this Agreement, neither Buyer, nor any of its Affiliates, are involved in substantive negotiations with respect to the acquisition of any business that would reasonably be expected to delay or impede the ability of the parties hereto to make or obtain any of the Governmental Filings required in connection with the consummation of the transactions contemplated by this Agreement.

(k) <u>Disclaimer of Other Representations and Warranties</u>. EXCEPT AS SPECIFICALLY SET FORTH IN THIS <u>SECTION 3.2</u>, NEITHER BUYER NOR ANY OF ITS DIRECTORS, OFFICERS, SECURITYHOLDERS, AFFILIATES, MANAGERS, MEMBERS, PARTNERS, EMPLOYEES, CONSULTANTS, AGENTS, COUNSEL OR ADVISORS MAKES OR HAS MADE ANY REPRESENTATION OR WARRANTY OF ANY KIND OR NATURE WHATSOEVER, ORAL OR WRITTEN, EXPRESS OR IMPLIED, TO SELLERS OR ANY OF ITS AFFILIATES. ANY REPRESENTATIONS AND WARRANTIES NOT SPECIFICALLY SET FORTH IN THIS <u>SECTION 3.2</u> ARE DISCLAIMED BY THE BUYER.

**3.3** <u>**Representations and Warranties of the Sellers.** Except as set forth on the Disclosure Schedule, subject to <u>Section 8.7</u>, each Seller represents and warrants to Buyer, solely with respect to himself or itself, that each of the statements contained in this <u>Section 3.3</u> is true and correct as of the date hereof:</u>

(a) Execution. If such Seller is a natural person, (i) such Seller has the requisite capacity, power and authority to execute and deliver this Agreement, to perform his obligations hereunder and to consummate the transactions contemplated hereby and (ii) this Agreement has been duly and validly executed and delivered by such Seller and, assuming the due authorization, execution and delivery thereof by the other parties thereto, constitutes a legal, valid and binding obligation of such Seller, enforceable against him in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law). If such Seller is a trust: (i) the duly appointed trustee of such Seller is named on the signature page of such Seller; (ii) such Seller's trustee has the requisite power and authority to execute and deliver this Agreement and to perform and consummate the transactions contemplated hereby on behalf of such Seller; (iii) the execution and delivery by the trustee of such Seller (on behalf of such Seller) of this Agreement, the performance by such trustee (on behalf of such Seller) of such Seller's obligations hereunder have been duly authorized by all necessary action on the part of such trustee, and no other authorization or consent of such trustee is necessary under applicable Law; (iv) the trust is duly organized and validly existing under the Laws of its state of organization; and (v) the trustee of each Seller that is a trust has not approved or proposed any amendment to such Seller's organizational documents (including, without limitation, such trust's declaration of trust and any other document pertaining to the duties and powers of the trustee thereof in such capacity as in effect as of the date of this Agreement).

(b) <u>Governmental Filings</u>. No Governmental Filings are required in connection with such Seller's execution, delivery and performance of this Agreement or the other agreements, instruments or documents contemplated hereby, or the consummation by him or it of the transactions contemplated hereby or thereby, as applicable, except the Governmental Filings set forth on <u>Part 3.3(b)</u> of the Disclosure Schedule and where the failure to obtain, give or make such Governmental Filings would not, individually or in the aggregate, reasonably be expected to materially impair the ability of such Seller to perform his or its obligations under this Agreement or the other agreements, instruments or documents contemplated hereby, or the consummation by him or it of the transactions contemplated hereby.

(c) <u>No Conflict or Violation</u>. The execution, delivery and performance by such Seller or such Seller's trustee (on behalf of such Seller) of this Agreement, and the consummation by such Seller of the transactions contemplated hereby and thereby do not (i) assuming all authorizations, consents and approvals described or referred to in <u>Section 3.3(c)</u> have been obtained or made, violate any applicable Law or Legal Requirements to which such Seller is subject, or (ii) require a consent, approval or notification under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate or cancel any Contract to which such Seller is a party, except, with respect to the foregoing clauses (i) and (ii) as would not, individually or in the aggregate, (x) reasonably be expected to be materially adverse to such Seller, and (y) would not materially impair the ability of such Seller to perform his or its obligations under this Agreement or the other agreements, instruments or documents contemplated hereby, or the consummation by him or it of the transactions contemplated hereby.

(d) <u>Legal Proceedings</u>. There are no Actions pending against such Seller, or such Seller's trustee (in her capacity as such) or, to the Knowledge of such Seller, threatened against such Seller which challenge the validity or enforceability of this Agreement or seek to enjoin or prohibit consummation of the transactions contemplated hereby. Such Seller is not subject to any judgment, decree, injunction or order of any Governmental Entity which would materially impair his or its ability to consummate the transactions contemplated hereby.

(e) <u>Brokers' Fees</u>. Such Seller has not dealt with any broker, finder or similar agent with respect to the transactions contemplated by this Agreement, and he or it is not under any obligation to pay any broker's fee, finder's fee, commission or similar amount in connection with the consummation of the transactions contemplated by this Agreement.

(f) <u>Ownership</u>.

(i) Heching is the lawful record and beneficial owner of (A) 100 shares of Class A Company Common Stock; and (B) 554 shares of Class B Company Common Stock, and one percent (1%) of the Capital Stock of Pathways, and has good and marketable title to such shares, free and clear of any Encumbrances.

(ii) The Trust is the lawful record and beneficial owner of (A) 346 shares of Class B Company Common Stock; (B) ninety percent (90%) of the Capital Stock of each of the

Florida Non-Wholly Owned Subsidiaries; and (C) ninety-four percent (94%) of the Capital Stock of Pathways, and has good and marketable title to such shares, free and clear of any Encumbrances.

(g) <u>Solvency</u>. Such Seller is solvent, will not be rendered insolvent as a result of the transactions contemplated hereby, and will remain solvent at all times prior to the expiration of the latest Survival Expiration Date. For purposes hereof, the term "solvency" means that: (i) such Seller is able to pay his or its debts or obligations in the ordinary course as they mature; and (ii) such Seller has capital sufficient to carry on his or its business and all business in which he or it is about to engage.

(h) <u>Disclaimer of Other Representations and Warranties</u>. THE REPRESENTATIONS AND WARRANTIES MADE BY SELLERS IN THIS <u>SECTION 3.3</u> (AS QUALIFIED BY THE APPLICABLE PART OF THE DISCLOSURE SCHEDULE) ARE THE EXCLUSIVE REPRESENTATIONS AND WARRANTIES MADE BY OR CONCERNING THE SELLERS AND THE SELLERS' OWNERSHIP OF COMPANY COMMON STOCK, CAPITAL STOCK OF THE FLORIDA NON-WHOLLY OWNED SUBSIDIARIES AND PATHWAYS. EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS <u>SECTION 3.3</u> (AS QUALIFIED BY THE APPLICABLE PART OF THE DISCLOSURE SCHEDULE), SELLERS EXPRESSLY DISCLAIM ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, WRITTEN OR ORAL, AS TO SELLERS, AND BUYER SHALL RELY ON ITS OWN EXAMINATION AND INVESTIGATION THEREOF AS WELL AS THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS <u>SECTION 3.3</u> (AS QUALIFIED BY THE APPLICABLE PART OF THE DISCLOSURE SCHEDULE). SELLERS ARE NOT, DIRECTLY OR INDIRECTLY, MAKING ANY REPRESENTATIONS OR WARRANTIES REGARDING THE VALUE OF SELLERS' OWNERSHIP OF COMPANY COMMON STOCK OR ANY PRO-FORMA FINANCIAL INFORMATION, FINANCIAL PROJECTIONS OR OTHER FORWARD-LOOKING STATEMENTS WITH RESPECT TO THE OWNERSHIP OF COMPANY COMMON STOCK OR THE ACQUIRED COMPANIES.

(i) <u>Reliance</u>. The Sellers acknowledge that none of Buyer, Amedisys or any other Person has made any representation or warranty, express or implied, written or oral, as to the accuracy or completeness of any information that Buyer or Amedisys have furnished or made available to the Sellers or their representatives, other than as explicitly set forth in <u>Section 3.2</u> and <u>Section 3.4</u>, and none of Buyer, Amedisys or any other Person (including any representative of Buyer or Amedisys) shall have or be subject to any liability to the Sellers, resulting from its use of any such information, documents or material made available to the Sellers or their representatives in any "data rooms," management presentations, due diligence or in any other form in expectation of the transactions contemplated herein that is not explicitly set forth in <u>Section 3.2</u> or <u>Section 3.4</u>. The Sellers acknowledge that, except for the representations and warranties contained in <u>Section 3.2</u> or <u>Section 3.4</u>, none of Buyer, Amedisys nor any other Person has made, and the Sellers have not relied on, any other representation or warranty, express or implied, written or oral, by or on behalf of Buyer or Amedisys. The Sellers acknowledge that neither they nor any other Person, directly or indirectly, has made, and it has not relied on, any representation or warranty regarding the pro-forma financial information, financial projections or other forward-looking statements of Buyer or Amedisys.

**3.4** <u>Representations and Warranties of Amedisys</u>. Except as set forth on the related Part of the Buyer Disclosure Schedule, Amedisys represents and warrants to the Sellers and the Company that each of the statements contained in this <u>Section</u> <u>3.4</u> is true and correct as of the date hereof.

(a) <u>Due Organization and Good Standing</u>. Amedisys is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware, with all requisite power and authority to own its properties and to carry on its business as such business is now conducted.

(b) <u>Authorization and Execution</u>. Amedisys has the power and authority and has taken all required action on its part necessary to permit it to execute and deliver and to carry out the terms of this Agreement. This Agreement has been duly and validly executed and delivered by Amedisys, and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes the legal, valid and binding obligation of Amedisys, enforceable against Amedisys in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

(c) <u>Solvency</u>. Amedisys is not insolvent and the execution of this Agreement and the consummation of the transactions contemplated herein will not render Amedisys insolvent.

(d) <u>Governmental Filings</u>. No Governmental Filings are required in connection with Amedisys' execution, delivery and performance of this Agreement or the other agreements, instruments or documents contemplated hereby, or the consummation by it of the transactions contemplated hereby or thereby, as applicable.

(e) <u>No Conflict or Violation</u>. The execution, delivery and performance by Amedisys of this Agreement, and the consummation by Amedisys of the transactions contemplated hereby and thereby do not (i) violate any applicable Law or Legal Requirements to which Amedisys is subject, (ii) require a consent, approval or notification under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate or cancel any Contract to which Amedisys is a party, or (iii) violate the Organizational Documents of Amedisys.

(f) <u>Legal Proceedings</u>. There are no Actions pending against Amedisys or, to the Knowledge of Buyer, threatened against Amedisys which challenge the validity or enforceability of this Agreement or seek to enjoin or prohibit consummation of the transactions contemplated hereby. Amedisys is not subject to any judgment, decree, injunction or order of any Governmental Entity which would materially impair its ability to consummate the transactions contemplated hereby.

(g) <u>Reliance</u>. Amedisys acknowledges that none of the Acquired Companies, the Sellers or any other Person has made any representation or warranty, express or implied, written

or oral, as to the accuracy or completeness of any information that the Acquired Companies or the Sellers have furnished or made available to Buyer, Amedisys or their representatives, other than as explicitly set forth in <u>Section 3.1</u> and <u>Section 3.3</u>, and none of the Acquired Companies, the Sellers or any other Person (including any representative of the Acquired Companies or the Sellers) shall have or be subject to any liability to Buyer, Amedisys, its Affiliates, or any other Person, resulting from Buyer's or Amedisys' use of any such information, documents or material made available to Buyer, Amedisys or their representatives in any "data rooms," management presentations, due diligence or in any other form in expectation of the transactions contemplated herein that is not explicitly set forth in <u>Section 3.1</u> or <u>Section 3.3</u>. Amedisys acknowledges that, except for the representations and warranties contained in <u>Section 3.1</u> or <u>Section 3.3</u>, none of the Sellers, the Company nor any other Person has made, and Amedisys has not relied on, any other representation or warranty, express or implied, written or oral, by or on behalf of the Sellers or the Company. Amedisys has not relied on, any representation or warranty regarding the pro-forma financial information, financial projections or other forward-looking statements of the Acquired Companies.

(h) <u>Disclaimer of Other Representations and Warranties</u>. EXCEPT AS SPECIFICALLY SET FORTH IN THIS <u>SECTION 3.4</u>, NEITHER AMEDISYS OR ANY OF ITS DIRECTORS, OFFICERS, SECURITYHOLDERS, AFFILIATES, MANAGERS, MEMBERS, PARTNERS, EMPLOYEES, CONSULTANTS, AGENTS, COUNSEL OR ADVISORS MAKES OR HAS MADE ANY REPRESENTATION OR WARRANTY OF ANY KIND OR NATURE WHATSOEVER, ORAL OR WRITTEN, EXPRESS OR IMPLIED, TO THE SELLERS OR THE ACQUIRED COMPANIES. ANY REPRESENTATIONS AND WARRANTIES NOT SPECIFICALLY SET FORTH IN THIS <u>SECTION 3.4</u> ARE DISCLAIMED BY AMEDISYS.

# ARTICLE IV COVENANTS

## 4.1 <u>Conduct of the Acquired Companies' Businesses</u>.

(a) During the period from the date of this Agreement until the earlier of the Closing or the termination of this Agreement, except as required by applicable Law or Legal Requirements, set forth on <u>Part 4.1</u> of the Disclosure Schedule, required by this Agreement or consented to by Buyer in writing (which consent shall not be unreasonably withheld, conditioned or delayed), the Company shall, and shall cause each other Acquired Company to:

(i) conduct its business and operations in the ordinary course of business consistent with past practice and in accordance with applicable Law and Legal Requirements in all material respects, except as expressly contemplated by this Agreement; and

(ii) maintain its legal existence.

(b) During the period from the date of this Agreement until the earlier of the Closing or the termination of this Agreement, except as required by applicable Law or Legal Requirements, set forth on <u>Part 4.1(b)</u> of the Disclosure Schedule, required by this Agreement or

consented to by Buyer in writing (which consent shall not be unreasonably withheld, conditioned or delayed), the Company shall not, and shall not permit any other Acquired Company to:

(i) effect any amendment to or change its Organizational Documents in any respect;

(ii) subject to <u>Section 4.1(c)</u>, make any dividend or distribution on the shares of Company Common Stock or Capital Stock, or issue, repurchase or redeem any equity interests or any options, warrants, convertible or exchangeable securities or other rights to acquire equity interests of any such Company Common Stock or Capital Stock;

(iii) issue any note, bond, or other debt security, or create, incur, assume or guarantee any Indebtedness or any capitalized lease obligation;

(iv) except in the ordinary course of business consistent with past practice, accelerate, terminate, modify or cancel any material right under any Company Contract;

(v) sell, lease, transfer or otherwise dispose of any of the material property or assets of any Acquired Company which are material to the Acquired Companies taken as a whole;

(vi) cancel, compromise or settle any material claim, or intentionally waive or release any material rights, of any Acquired Company;

(vii) except as required by applicable Law from time to time in effect or by the terms of any Benefit Plans: (A) increase the annual level of compensation of any director or employee of the Acquired Companies, except for increases in the ordinary course of business consistent with past practice (provided that any annual compensation increase of \$20,000 or more for any director or employee, whether or not in the ordinary course of business, shall require the prior written consent of Buyer, which consent shall not be unreasonably withheld, conditioned or delayed), (B) grant any increased bonus, benefit or other direct or indirect compensation to any director or employee of the Acquired Companies, except in the ordinary course of business consistent with past practice, (C) increase the coverage or benefits available under any (or create any new) Benefit Plan, except in the ordinary course of business consistent with past practice, or (D) enter into any employment, deferred compensation, severance, consulting, non-competition or similar agreement (or amend any such agreement) to which any of the Acquired Companies is a party, except in the ordinary course of business consistent with past practice to the extent otherwise permitted pursuant to this Agreement;

(viii) subject to any Encumbrance any of the properties or assets of the Acquired Companies, except for Permitted Encumbrances;

(ix) make or change any material Tax election, other than as required by applicable Law or Legal Requirements or as may be consistent with past practices, or settle or compromise any material liability relating to Taxes with a Taxing Authority;

(x) adopt or change any material accounting method, except as required by GAAP, file any material amended Tax Return, enter into any material closing agreement, surrender any right to claim a material refund of Taxes, consent to any extension or waiver of the limitation period applicable to any material Tax claim or assessment relating to an Acquired Company, or take any other similar action relating to the filing of any Tax Return or the payment of any Tax, if such election, adoption, change, amendment, agreement, surrender, consent or other action would have the effect of materially increasing the Tax liability of any Acquired Company for any period ending after the Closing Date or materially decreasing any Tax attribute of any Acquired Company existing on the Closing Date, other than as required by applicable Law or Legal Requirements or as may be consistent with past practices;

(xi) acquire any corporation, partnership or other business organization or division thereof or collection of assets constituting all or substantially all of a business or business unit, whether by merger or consolidation, purchase of substantial assets or equity interests or any other manner;

(xii) adopt a complete or partial plan of liquidation or resolutions authorizing or providing for such a liquidation or dissolution, consolidation, recapitalization, reorganization or bankruptcy, or make a general assignment for the benefit of creditors;

(xiii) make any material change in its customs or practices regarding cash management, collection of accounts receivable, payments of accounts payable, or cash distributions;

insider; or

(xiv) engage in any transaction, arrangement or contract with any officer, director, shareholder or other

(xv) agree or otherwise commit to take any of the actions prohibited by the foregoing clauses (i) through (xiv) or that would require disclosure under <u>Section 3.1(j)</u>.

(c) Notwithstanding anything to the contrary herein, the Company may distribute to the Sellers prior to the Closing, by distribution or otherwise, all or any cash and cash equivalents of the Company and the Subsidiaries in excess of the Closing Cash Target.

(d) Buyer acknowledges and agrees that (i) nothing contained in this Agreement shall give Buyer, directly or indirectly, the right to control or direct the operations of the Acquired Companies prior to the Closing, (ii) prior to the Closing, the Sellers shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over the operations of the Acquired Companies, and (iii) notwithstanding anything to the contrary set forth in this Agreement, no consent of Buyer shall be required with respect to any matter set forth in <u>Section 4.1</u> or elsewhere in this Agreement to the extent that the requirement of such consent would, upon advice of the Company's counsel, violate any applicable Law.

# 4.2 <u>Publicity</u>.

Prior to the Closing, no party shall (or permit any Affiliate or any agent or representative thereof to) issue a press release or make any other public announcement concerning the transactions

contemplated by this Agreement without the prior written consent of the Sellers and Buyer, except as such release or announcement may be required by Law or the rules or regulations of the United States or foreign securities exchange, in which case the party required to make the release or announcement shall allow each of the other parties reasonable time to comment on such release or announcement in advance of such issuance. Upon execution of this Agreement, the parties have agreed that (i) a mutually agreeable public announcement will be made, (ii) the parties will implement a mutually agreed joint communication plan to patients/payors, referral sources, vendors and suppliers of the Acquired Companies, and (iii) the parties will implement the mutually agreed upon communication and preparedness plan with respect to the employees and contractors of the Acquired Companies, which plan is attached hereto as <u>Schedule 4.2</u> (such plan, as it may be amended by the mutual agreement of Buyer and the Sellers, the "<u>Employee Communication Plan</u>"). After the Closing, no press releases related to this Agreement and the transactions contemplated herein, will be issued without the approval of Buyer (which approval, in each case, shall not be unreasonably withheld, conditioned or delayed).

#### 4.3 Confidentiality.

(a) Any information regarding the Acquired Companies heretofore or hereafter obtained from the Acquired Companies by Buyer shall be subject to the terms of the Confidentiality Agreement dated July 13, 2017, by and between Amedisys and the Company (the "<u>Nondisclosure Agreement</u>"), and, except as otherwise permitted or contemplated by this Agreement, such information shall be held by Buyer in accordance with the terms of the Nondisclosure Agreement, which shall continue in full force and effect. Amedisys and the Company hereby agree that the term of the Nondisclosure Agreement is hereby extended until the earlier of (i) the twelve month anniversary of the date of the Agreement and (ii) the Closing.

(b) The Sellers acknowledge that they have had in the past, currently have and in the future may have access to Confidential Information with respect to the business of the Acquired Companies. Following the Closing, the Sellers agree that they will keep confidential all such Confidential Information furnished to them and, except with the specific prior written consent of Buyer, will not use any such Confidential Information and will not disclose any such Confidential Information to any Person except (i) representatives of Buyer and its Affiliates and (ii) their own representatives, provided that these representatives agree to the confidentiality provisions of this Section 4.3(b); provided, however, that, for purposes of this Section 4.3(b), Confidential Information does not include such information as (i) becomes known to the public generally through no fault of the Sellers or their representatives or (ii) is required to be disclosed by applicable Law or Legal Requirements, provided, however, that prior to the disclosure by the Sellers of any information under this clause (ii), the Sellers will give, to the extent permitted by Law, prior written notice to Buyer of the circumstances requiring disclosure and provide Buyer with the opportunity to contest that disclosure. Notwithstanding the above, and for the avoidance of doubt, in the event that this Agreement is terminated prior to Closing, the Sellers shall have no obligation to Buyer and its Affiliates with respect to the Confidential Information of the Acquired Companies.

## 4.4 Access to Information.

(a) During the period from the date of this Agreement to the Closing Date, if reasonably requested by Buyer, the Sellers shall, and shall cause each Acquired Company to, afford the officers, directors, employees and other agents of Buyer (i) reasonable access during normal business hours to the properties, books and records, data systems, offices and other facilities of the Acquired Companies, and (ii) reasonable access during normal business hours to the employees and contractors of the Acquired Companies in accordance with the terms of the Employee Communication Plan. The parties shall work together in good faith to develop a mutually acceptable plan whereby Buyer will contact and communicate with the payors, referral sources, vendors, and suppliers of the Acquired Companies in a manner reasonably necessary for Buyer to prepare for its operation of the Acquired Companies after the Closing, subject to Buyer's compliance with all applicable Legal Requirements in connection therewith. Any such access or contact shall be managed by and conducted through those representatives identified by the Seller, and shall not be permitted to the extent that (i) it would unreasonably disrupt the operations of the Acquired Companies or (ii) would require the Acquired Companies to disclose information that, in the reasonable judgment and good faith of counsel to the Acquired Companies, is subject to attorney-client privilege or may conflict with any applicable Law or confidentiality obligations to which the Acquired Companies are bound.

(b) Subject to <u>Section 4.13</u>, no information obtained by Buyer pursuant to this <u>Section 4.4</u> or otherwise shall be deemed to amend or supplement the Disclosure Schedule, to prevent or cure any breach of warranty, or breach of covenant, or to otherwise limit or affect any rights of the Buyer Indemnitees under ARTICLE VII.

# 4.5 Filings and Authorizations.

(a) Each of Buyer and the Company shall, and the Company shall cause the other Acquired Companies to, promptly make or cause to be made any and all required filings with

Governmental Entities for the change of ownership of the Permits issued to the Acquired Companies by such Governmental Entities, including those noted on Part 3.1(c) of the Disclosure Schedule. Each of Buyer and the Sellers agree to cooperate with and promptly to consult with the other party or its counsel with respect to any filing with any Governmental Entity. Without limitation of the foregoing, promptly following the date hereof, the Sellers and the Buyer shall jointly approach the OIG in relation to the Corporate Integrity Agreement and the transactions contemplated hereby and work together in good faith to jointly make or cause to be made all Governmental Filings, if any, required in connection with the transactions contemplated by this Agreement.

(b) The parties agree to cooperate fully and promptly respond to any inquiries or investigations initiated by such Governmental Entities in connection with any such filings. The Sellers shall deliver (or cause to be delivered) to Buyer at or prior to the Closing all such documents as Buyer may reasonably request to obtain approval of the change of ownership of the Acquired Companies contemplated by <u>Section 4.5(a)</u> of this Agreement.

(c) Each of the parties shall (i) cooperate with one another and respond as promptly as practicable to any inquiries or requests received from the foregoing Governmental Entities, the Attorney General Office of any State, or other applicable Governmental Entity for additional information or documentation, (ii) not enter into any agreement with the foregoing Governmental Entities not to consummate the transactions contemplated by this Agreement, except with the prior written consent of the other party, which consent shall not be unreasonably withheld or delayed, (iii) use reasonable best efforts to obtain any clearances or consents that may be required, (iv) make any further filings or information submissions pursuant thereto that may be reasonably necessary or advisable, and (v) promptly make any filings or submissions required under any applicable state antitrust or trade regulation Law. Each of such parties shall use its reasonable best efforts to obtain any clearance or consents or to resolve any objections that may be asserted by the applicable Governmental Entity, in each case as promptly as practicable. Each of the parties shall (A) promptly notify the other parties of any written communication to that party or its Affiliates from any Governmental Entity to the extent permitted by applicable Law and, subject to applicable Law, permit the other parties to review in advance any proposed written communication to any of the foregoing, (B) to the extent permitted by applicable Law, not agree to participate, or to permit its Affiliates to participate, in any substantive meeting or discussion with any Governmental Entity in respect of any filings, investigation or inquiry concerning the transactions contemplated by this Agreement unless it consults with the other parties in advance and, to the extent permitted by such Governmental Entity, gives the other parties the opportunity to attend and participate thereat, (C) use reasonable best efforts to consult in advance and cooperate with each other in connection with any analyses, appearances, presentations, memoranda, briefs, arguments and proposals to be made or submitted by or on behalf of any party to a Governmental Entity relating to the transactions, and (D) to the extent permitted under applicable Law, furnish the other parties with copies of all correspondence, filings, and written communications between such party and its Affiliates and their respective Representatives, on the one hand, and any Governmental Entity, on the other hand, with respect to this Agreement and the transactions contemplated hereby (unless the furnishing of such information would (1) violate the provisions of any applicable Law or any confidentiality agreement or (2) cause the loss of the attorney-client privilege with respect thereto; provided that each such party shall use its reasonable best efforts to promptly communicate to the other parties the substance

of any such communication, whether by redacting parts of such material communication or otherwise, so that such communication would not violate applicable Law or cause the loss of the attorney-client privilege with respect thereto). Further, any party may, as it deems advisable and necessary, reasonably restrict any competitively sensitive material provided to the other parties under this <u>Section 4.5</u> as "outside counsel only," and with respect to material so designated, outside counsel to whom such material is provided may not disclose the information to others without prior authorization from the party providing the material.

(d) Promptly following the parties' execution and delivery of this Agreement, the Sellers will give (or will cause the Acquired Companies to give) any notices to third parties and will, from the date of this Agreement until the Closing, use reasonable best efforts to obtain (or cause the Acquired Companies to obtain) any third party consents referred to in <u>Part 3.1(f)</u> of the Disclosure Schedule.

(e) Buyer shall be solely responsible for the cost of all filing, consent or approval fees relating to the filings or submissions to be made pursuant to this <u>Section 4.5</u>.

**4.6** <u>Closing Efforts</u>. Upon the terms and subject to the conditions herein provided, except as otherwise provided in this Agreement, and without limiting the obligations of the parties under <u>Section 4.5</u>, each of the parties hereto agrees to act in good faith and use reasonable best efforts to take or cause to be taken all actions, to do or cause to be done and to assist and cooperate with the other party hereto in doing all things necessary, proper or advisable under applicable Laws and Legal Requirements to satisfy the conditions specified in this Agreement necessary to consummate the transactions contemplated hereby. For avoidance of doubt and without otherwise limiting or expanding the foregoing, none of the Acquired Companies, the Sellers nor any of their respective Affiliates shall be obligated to make any payments or otherwise pay any consideration to any third party to obtain any applicable consent, waiver or approval, unless the payment of such consideration is required by applicable Law or the terms of the applicable Contract.

**4.7** <u>Compliance with Employment Laws</u>. On and after the Closing Date, Buyer and the Company shall, and the Company shall cause the other Acquired Companies to, comply with all applicable Laws requiring notice to employees in the event of a plant closing, layoff, discharge, termination, relocation or reassignment of employees, including the WARN Acts. For the avoidance of doubt, Buyer and the Company shall be responsible for notices, payments or other obligations due to any employees, and all notices, payments, fines or assessments due to any Governmental Entity pursuant to any applicable federal, state or local Laws, common law, rule or regulation with respect to the employment, discharge, layoff, termination, relocation or reassignment of any employees of the Acquired Companies by the Acquired Companies occurring or accruing at or after the Closing, including but not limited to foreign, state or local Laws and any rules or regulations as have been issued in connection with the foregoing.

## 4.8 Further Assurances.

(a) On and after the Closing Date, the Sellers, the Company and Buyer shall cooperate and use all of their respective reasonable best efforts to take or cause to be taken all appropriate actions and do, or cause to be done, all things necessary or appropriate to consummate

and make effective the transactions contemplated hereby, including the execution of any additional documents or instruments of any kind, the transfer of assets or property, the obtaining of consents which may be reasonably necessary or appropriate to carry out any of the provisions hereof and the taking of all such other actions as such party may reasonably be requested to take by the other party hereto from time to time, consistent with the terms of this Agreement and the other agreements, instruments and documents contemplated hereby, in order to effectuate the provisions and purposes hereof and thereof and the transactions contemplated hereby.

(b) Neither Buyer, nor any of its Affiliates, will acquire, or agree to acquire, or, engage in substantive negotiations with respect to the acquisition of any business that would reasonably be expected to delay or impede the ability of the parties hereto to make or obtain any of the Governmental Filings required in connection with the consummation of the transactions contemplated by this Agreement.

**4.9** <u>Exclusivity</u>. From the date of this Agreement until the Closing Date or the earlier termination of this Agreement, none of the Sellers, the Company, nor any of their respective Affiliates, employees, officers, directors, members, managers, advisors, agents or other representatives shall, acting in their capacity as such, directly or indirectly, initiate with, solicit from, encourage or respond to (including by way of furnishing non-public information or assistance), or enter into discussions or negotiations of any type, directly or indirectly, or enter into a confidentiality agreement, letter of intent or purchase agreement, merger agreement or other similar agreement with, any Person other than Buyer and its Affiliates and advisors with respect to the acquisition of the business of the Acquired Companies, pursuant to a sale of stock, merger, consolidation or other business combination involving the Acquired Companies, or a sale of all or a material portion of the assets of the Acquired Companies. If, from the date of this Agreement until the Closing Date or the earlier termination of this Agreement, the Company or any of the other Persons referenced above receives an offer or proposal relating to any acquisition of the business of the Acquired Companies, such Person shall notify Buyer of the receipt of such offer, the terms of the offer, and the identity of the offeror.

4.10 **Books and Records.** From and after the Closing, Buyer shall, and shall cause the Acquired Companies to,

preserve and keep the records held by them relating to the Business (which, for the avoidance of doubt, shall not include Tax Returns of Sellers or any other Person other than an Acquired Company) in accordance with the record retention policy of Amedisys, a copy of which has been provided to the Sellers. Without limiting the provisions of <u>Section 4.11</u> below, after the Closing, for so long as such books and records are retained by the Acquired Companies in accordance with the record retention policy of Amedisys, the Sellers and their accountants, lawyers and representatives shall be entitled, upon reasonable advance written notice, to have access to and to make copies (at the Sellers' expense) of the pre-Closing books and records of the Acquired Companies for purposes of the preparation of Tax Returns. In the event of any litigation or threatened litigation between the parties relating to this Agreement or the transactions contemplated hereby, the covenants contained in this <u>Section 4.10</u> shall not be considered a waiver by any party of any right to assert the attorney-client privilege.

# 4.11 <u>Certain Tax Matters</u>.

(a) Sellers shall prepare and file, or cause to be prepared and filed, when due any Income Tax Return that is required to be filed by or with respect to the Acquired Companies after the Closing Date that is an Income Tax Return of any Acquired Company for any Pre-Closing Period. Such Tax Returns shall, at Sellers' option, include safe harbor elections under Revenue Procedure 2011-29 and shall treat, to the maximum extent permitted by Law, Company expenses as arising and deductible in the tax years that end on or prior to the Closing Date. The Sale Bonus Payment shall be treated as a deductible by the Company in the tax year that ends on the Closing Date. Sellers shall provide Buyer with a draft of any such Tax Return not less than thirty (30) days prior to its due date and shall consider in good faith any comments on such draft received by Sellers not less than fifteen (15) days prior to the due date for such Tax Return. Each of Buyer and the Company shall, and the Company shall cause the other Acquired Companies to, provide to Sellers all information in its possession and under its control which is reasonably necessary or requested for Sellers to prepare and file such Pre-Closing Period Income Tax Returns. If Sellers are not authorized under applicable Law to execute and file such Pre-Closing Period Income Tax Return, Buyer shall execute and file (or cause to be filed) such Pre-Closing Period Income Tax Return with the appropriate taxing authority. Buyer shall prepare and file, or cause to be prepared and filed, when due any Tax Return that is required to be filed by or with respect to the Acquired Companies after the Closing Date (i) that is not required to be prepared by Sellers under this Section 4.11(a) and (ii) is for any taxable period that begins after the Closing Date or any Straddle Period (collectively, the "Company Tax Returns"), and shall remit any Taxes due with respect to such Company Tax Returns. All such Company Tax Returns shall be prepared in a manner that is consistent with those prepared for prior taxable periods unless a different treatment of any item is required by applicable law or required by this Agreement. Buyer will furnish a copy of such Company Tax Returns and a copy of related work papers and supporting information to the Sellers and their tax advisers for their review and written approval at least sixty (60) days prior to the due date (or extended due date that has been approved by the Sellers) for filing such Company Tax Returns (or if on the Closing Date less than sixty (60) days remain before filing is due, one-half of the days remaining between the Closing and the filing due date). Simultaneous with furnishing such Company Tax Returns to the Sellers, Buyer shall provide a written statement to the Sellers setting forth the amount of Taxes owing on such Company Tax Returns that, in accordance with Section 4.11(b), it believes is owing by the Sellers. Should the Sellers disagree with the calculation of Taxes shown as being due on such Company Tax Returns or the amount of Taxes owing therein that Buyer believes is owing by the Sellers, it shall provide written notification thereof (such notice, a "Tax Dispute Notice") to Buyer within thirty (30) days of receiving the copy of such Company Tax Return, and Buyer and the Sellers shall cause their respective tax advisers to confer in good faith to seek to resolve such disagreement to the satisfaction of Buyer and the Sellers. If Buyer and the Sellers are unable to resolve such disagreement within the earlier of fifteen (15) days after the date on which the Tax Dispute Notice has been provided or seven (7) days prior to the due date of such Company Tax Return and no extension for the filing of such Company Tax Return is then legally available, Buyer may file the related Company Tax Return prior to the expiration of the deadline for the filing thereof notwithstanding the disagreement being unresolved, with an amended Company Tax Return to be subsequently promptly prepared and filed if required by the final resolution of such disagreement pursuant to this Section 4.11 (a); provided, however, it is understood that the indemnification obligations under this Section 4.11 shall be determined by reference to the post-filing resolution of the dispute by the parties or the decision of the Independent Accounting

Firm pursuant to this <u>Section 4.11(a)</u> rather than the content of any Company Tax Return (and the Tax consequences of such Company Tax Return) filed by Buyer prior to the resolution of the dispute. If resolution of a Tax dispute described in this <u>Section 4.11(a)</u> is not reached prior to the filing of a Company Tax Return and such dispute is not resolved within thirty (30) days of the Company Tax Return being filed, either Buyer or the Sellers may request that the disagreement be resolved by the Independent Accounting Firm. The calculation of the Taxes due and the party(ies) responsible for the payment thereof shall be made by such firm within the range of the respective calculations of Buyer and the Sellers and when so made shall be conclusive, and shall be binding on, and nonappealable by, the parties. The fees and disbursement of such firm shall be borne equally by Buyer and the Sellers.

(b) Subject to the terms and conditions of this Section 4.11, the Sellers will defend and indemnify the Buyer Indemnitees against the following Taxes, but only to the extent such Taxes have not been paid prior to the Closing, adequately provided for by a liability accrued for Taxes on the Latest Balance Sheet or resulted in a payment by Sellers or from the Escrow Fund: (i) the non-payment prior to the Closing of Taxes owed by the Acquired Companies for any Pre-Closing Period and, under the principles of Section 4.11(c), for the portion of any Straddle Period ending on the Closing Date, (ii) the non-payment of Taxes of any Person other than the Acquired Companies that are imposed on the Acquired Companies as a transferee or successor, by contract or pursuant to any Law, which Taxes relate to an event or transaction occurring before the Closing and where the status or relationship of the Acquired Companies giving rise to the liability for such Taxes of another Person existed prior to Closing, and (iii) the non-payment of Taxes owed by the Acquired Companies as a result of their inclusion prior to Closing in a consolidated tax return group; provided, however, that Sellers shall have no obligation to defend and indemnify the Buyer Indemnitees against (I) any Taxes resulting from (x) an election made pursuant to Section 338 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax Law) with respect to Buyer's purchase or deemed purchase of the capital stock of any Acquired Company pursuant to this Agreement (other than the Section 338(h)(10) Election explicitly contemplated by this Agreement), (y) any transactions occurring on the Closing Date after the Closing outside the ordinary course of business (other than as explicitly contemplated by this Agreement), or (z) any breach by Buyer of any of its obligations under Section 4.11(m) of this Agreement or (II) the employer's share of any employment Taxes associated with the Sale Bonus Payment. Except as otherwise provided in this Section 4.11, any Taxes payable by the Sellers pursuant to the foregoing sentence shall be paid to Buyer in a manner consistent with the terms of ARTICLE VII.

(c) In the case of any Taxes of the Acquired Companies that are payable with respect to any Straddle Period, the portion of such Taxes that relates to the portion of such taxable period ending on the Closing Date shall (i) in the case of any real property, personal property or other *ad valorem* Taxes be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction the numerator of which is the number of days in the portion of the period ending on the Closing Date and the denominator of which is the number of days in the entire taxable period, and (ii) in the case of any other Tax be deemed equal to the amount that would be payable if the relevant taxable period (and the taxable period of any Subsidiary of such Acquired Company that is not itself an Acquired Company and which is a pass-through entity for purposes of the relevant Tax) ended on the Closing Date based on a closing of the books as of the close of business on the

Closing Date; *provided, however*, that any transaction occurring after the Closing that is not in the ordinary course of business (other than the transactions contemplated by this Agreement), shall be deemed to occur in a taxable period (or portion thereof) after the Closing Date.

(d) Buyer agrees that (i) the Company and the Non-Wholly Owned Subsidiaries (other than Pathways) will join the consolidated Income Tax Return group of which Amedisys is the parent corporation for U.S. federal income tax purposes (and for purposes of any similar applicable state, local or foreign laws) at the end of the Closing Date pursuant to Treasury Regulation Section 1.1502-76(b)(1)(ii)(A) and (ii) as a result, such Acquired Companies will have a short tax year ending on (and including) the Closing Date and will be included in the consolidated group's U.S. federal (and similar applicable state, local or foreign) Income Tax Returns starting the day after the Closing Date.

(e) In the case of any audit, assessment, suit, proposed adjustment, deficiency, dispute, administrative or judicial proceeding or similar claim relating to Taxes relating to any Pre-Closing Period that, if determined adversely to the Acquired Companies, would be grounds for a claim for indemnification under this <u>Section 4.11</u> (each, a "<u>Tax Claim</u>"), the Sellers shall have the right, at the sole expense of the Sellers, to assume and control the conduct of such Tax Claim and shall have the right to settle such Tax Claim; *provided, however*, (i) that the Sellers will consult with Buyer in the negotiation and settlement of any such Tax Claim, (ii) Buyer shall be permitted to participate, at its own expense, in the defense of any such Tax Claim, and (iii) the Sellers shall not settle or compromise any such Tax Claim without Buyer's prior written consent, with such consent not to be unreasonably withheld, conditioned or delayed.

(f) In the case of any Tax Claim relating to any Straddle Period, Buyer and the Sellers may each participate, at their own expense, in the audit or proceeding, and the audit or proceeding shall be controlled by Buyer or the Sellers, whichever would bear the burden of the greatest portion of the Tax Claim; *provided, however*, that the party controlling the Straddle Period Tax Claim (i) shall not settle such audit or proceeding without the consent of the other party, which consent shall not be unreasonably withheld, conditioned or delayed, and (ii) shall consult with and keep the other party timely informed with respect to the commencement, status and nature of any such Tax Claim.

(g) In the case of any Tax Claim relating to the Taxes of any taxable period beginning after the Closing Date, Buyer shall have the right to control the conduct of such Tax Claim and shall have the right to settle such Tax Claim without the consent of any other party; *provided, however*, that, if such Tax Claim may give rise to a claim for indemnification against Sellers under this Agreement, (i) Sellers shall have the right to participate in, but not control, the defense of such Tax Claim and (ii) any settlement or compromise of such Tax Claim shall require the prior written consent of the Sellers (which consent shall not be unreasonably withheld, conditioned or delayed).

(h) Buyer and the Sellers shall provide each other, and Buyer shall cause the Acquired Companies to provide the Sellers, with such cooperation and assistance as may be reasonably requested by Buyer or a Seller in connection with (i) the preparation of any Tax Return and (ii) any Tax Claims, and until the eighth (8<sup>th</sup>) anniversary of the Closing Date, each will retain

and provide, and Buyer will cause the Acquired Companies to retain and provide, the other with any requested records or information that is relevant to such Tax Return or Tax Claim; provided that Buyer and the Sellers and their respective Affiliates, as applicable, shall have the right to redact any portions of any such records or information that are not solely related to one or more of the Acquired Companies before providing any such records or information to the other party. Such cooperation and assistance shall include making employees reasonably available on a mutually convenient basis during normal business hours to provide additional information; *provided, however*, the party requesting such information available. Furthermore, the parties will reasonably cooperate with each other in connection with the Section 338(h)(10) Election, as described in, and without limiting the provisions of, <u>Section 4.11(j)</u> below. Such cooperation will include providing records and information that are reasonably relevant to the making of the Section 338(h)(10) Election, and making employees available on a mutually convenient basis to provide additional information. Without limiting the foregoing provisions of this <u>Section 4.11(h)</u>, Buyer and the Company agree for the benefit of the Sellers that they will retain or cause to be retained, until the eighth (8<sup>th</sup>) anniversary of the Closing Date, copies of all Tax Returns of the Acquired Companies, and work schedules and other records or information that they possess and that are relevant to such Tax Returns of the Acquired Companies for all periods commencing on or before the Closing Date.

(i) Except as otherwise provided herein, Sellers shall reimburse Buyer for any Taxes of the Acquired Companies that are the responsibility of Sellers pursuant to this Section 4.11 within fifteen (15) business days after Buyer's payment of such Taxes.

(i) Sellers and Buyer shall, and shall cause their Affiliates (as applicable) to: (A) make a timely joint election under Code Section 338(h)(10) (and any corresponding election under state, local, and non-U.S. Tax law) with respect to the purchase and sale of the shares of Capital Stock of the Company provided for in this Agreement (collectively, the "Section 338(h)(10) Election"), (B) provide prior to or at the Closing the information necessary to permit the Section 338(h)(10) Election to be made, (C) take all action necessary and appropriate (including filing such Tax Returns, forms, elections, schedules and other documents that may be required) to effect and preserve the Section 338(h)(10) Election, (D) report Buyer's acquisition of the Company consistently with the Section 338(h)(10) Election, and (E) take no position to the contrary in any Tax Return, any proceeding before a taxing authority or otherwise prejudice the Section 338(h)(10) Election. In furtherance of the foregoing, on the Closing Date, Sellers shall deliver to Buyer an executed IRS Form 8023 or any other analogous or corresponding form required by state or local law, to be filed by Buyer (or, at Buyer's direction, filed by Sellers) with any state, local or foreign Governmental Authority to effect the Section 338(h)(10) Election. For purposes of executing and filing the IRS Form 8023, (X) Buyer shall timely execute two (2) originals of IRS Form 8023, (Y) promptly thereafter Buyer shall provide to Sellers one (1) executed original of IRS Form 8023, and (Z) Buyer shall timely file one (1) executed original IRS Form 8023 with the IRS. The contents of the Section 338(h) (10) Election shall be consistent with the Allocation (as defined in Section 4.11(i)). In the event that the Section 338(h)(10) Election is disputed by any Governmental Authority, the party receiving notice of such dispute shall promptly notify and consult with the other party hereto concerning resolution of such dispute. No election under Code Section 338(h)(10) or any

corresponding provision of state, local or non-U.S. Tax law shall be made with respect to any of the Non-Wholly Owned Subsidiaries. Apart from the Section 338(h)(10) Election, no election under Code Sections 338 or 336(e) (or any corresponding or similar elections under state, local or non-U.S. Tax Law) shall be made in connection with the transactions provided for herein.

(k) The parties agree that for any Acquired Company that (i) is treated as a partnership for U.S. federal income Tax purposes and (ii) will continue to be so treated after the Closing, a timely election pursuant to Section 754 of the Code shall be duly made with respect to such Acquired Company, which election shall be in effect for the taxable period that includes the Closing Date.

(1) The portion of the Purchase Price (and any other consideration paid hereunder for the Shares) and the liabilities (to the extent included in amount realized for U.S. federal income Tax purposes) of the Acquired Companies (other than the Non-Wholly Owned Subsidiaries) allocable for U.S. federal income Tax purposes to the shares of Capital Stock of the Company, plus other relevant items, shall be allocated among the assets of the Company in accordance with Code Sections 338 and 1060 and the Treasury Regulations promulgated thereunder. Within sixty (60) days after the determination of the adjustments, if any, to the Purchase Price hereunder, or, if sooner, ninety (90) days before IRS Form 8883 must be filed, Buyer will provide to Sellers copies of IRS Form 8883 and any required exhibits thereto (the "Allocation") with Buyer's proposed allocation of the Purchase Price consistent with Code Sections 338 and 1060 and the Treasury Regulations promulgated thereunder. Within thirty (30) days after the receipt of such Allocation, Sellers will submit to Buyer in writing any proposed changes to such Allocation or shall indicate its concurrence therewith, which concurrence shall not be unreasonably withheld, conditioned or delayed (and in the event no such changes are proposed in writing to Buyer within such period of time, Sellers will be deemed to have agreed to, and accepted, the Allocation). Buyer and Sellers will endeavor in good faith to resolve any differences with respect to the Allocation within fifteen (15) days after Buyer's receipt of written notice of objection from Sellers. Any unresolved disputes will be resolved by the Independent Accounting Firm, the costs of which shall be borne by each party in the percentage inversely proportionate to the percentage of the total dollar amount of the total items submitted for dispute that are resolved in such party's favor. The determination of the Accounting Firm shall be binding on the parties. Sellers and Buyer shall each timely file IRS Form 8883 by attaching such form to their respective timely filed U.S. federal income Tax Returns reflecting the tax effects of the Section 338(h)(10) Election (or any successor forms, and any forms or schedules similar to Form 8883 that are required to make the Section 338(h)(10) Election effective under provisions of applicable state, local, or foreign law) and otherwise in a manner reflecting the Allocation, and Sellers and Buyer shall reasonably cooperate with each other in connection with such preparation and filing. All Tax Returns and reports filed by Buyer, the Acquired Companies and Sellers will be prepared consistently with the Allocation. None of Buyer, the Acquired Companies or Sellers shall take any position (whether in audits, Tax Returns or otherwise) that is inconsistent with the Allocation unless required to do so by Law. The parties shall cooperate in updating the Allocation and any IRS Form 8883 with respect to any post-Closing adjustments to the purchase price hereunder.

(m) Buyer and Sellers shall not revoke the Company's election to be taxed as an S corporation within the meaning of Code Section 1361 and Code Section 1362. Buyer and Sellers

shall not take or allow any action (other than the sale of the Company's stock pursuant to this Agreement) that would result in the termination of the Company's status as a validly electing S corporation within the meaning of Code Section 1361 and Code Section 1362. Buyer shall not, and shall not cause or permit any of the Acquired Companies to, (i) amend any Tax Returns of the Acquired Companies filed with respect to any Pre-Closing Periods or (ii) make any Tax election that has retroactive effect to any Pre-Closing Periods, in each such case without the prior written consent of Sellers. Other than as required by this Agreement, Buyer shall not cause or permit the Acquired Companies to take any action on the Closing Date that is outside the ordinary course of business (other than any action which the relevant Acquired Company was required to take under Contracts in place prior to Closing).

(n) Any Tax refunds that are received by Buyer or any of its Affiliates (including, following the Closing Date, for the avoidance of doubt, the Acquired Companies), and any amounts credited against any Tax to which Buyer or any of its Affiliates (including, following the Closing, for the avoidance of doubt, the Acquired Companies) become entitled that relate to any Pre-Closing Period or, under the principles of Section 4.11(c), any Straddle Period of any Acquired Company, shall be for the account of Sellers, and Buyer shall pay over to Sellers any such refund or the amount of any such credit within fifteen (15) days after receipt or entitlement thereto.

(o) Buyer shall be responsible for all Transfer Taxes arising in connection with the transactions provided for in this Agreement and shall be responsible for the preparation and filing of all Tax Returns with respect to such Transfer Taxes. Sellers shall provide Buyer with such cooperation as Buyer may reasonably request in connection with the preparation, execution and filing of such Tax Returns.

(p) In the event that Sellers make any indemnity payments hereunder, or amounts are distributed other than to Sellers from the Escrow Fund, the Sellers shall make payments between themselves to properly reflect, in light of their relative direct and indirect interests sold hereunder in respect of the Acquired Companies, the extent to which such payments and distributions are attributable to specific Acquired Companies. This provision shall be enforceable only by the Sellers.

**4.12** <u>Release by the Sellers</u>. Effective as of the Closing, the Sellers release each Acquired Company, its successors and assigns, and its Affiliates, members, managers, directors, and officers from any claim, demand, lien, liability, debt, right, set-off, trespass, tort, wrong, covenant, action, suit, expense, damage, judgment, order and liability of whatever kind or nature, in law or in equity, under contract, in tort, by statute or otherwise, whether known or unknown, vested or contingent, suspected or unsuspected, and whether or not concealed or hidden, that were or could have been asserted in any suit, arbitration or mediation, in any jurisdiction, state, federal or otherwise, under any law, state, federal or otherwise, whether arising out of or relating to, in whole or in part, the Sellers' ownership of the shares of Company Common Stock and the Capital Stock of the Acquired Companies, the Sellers' status as shareholders of the Company, Heching's status as an employee or officer of any Acquired Company, or otherwise, <u>provided</u>, <u>however</u>, that in no event shall the foregoing release apply with respect to (a) any rights with respect to any directors' and officers' insurance policy, fiduciary liability insurance policy or other similar policy maintained by an Acquired Company, (b) any salary or other compensation and benefits accrued or payable through

the Closing, including any rights to receive benefits under any Benefit Plan and any business expense reimbursement in the ordinary course of business, (c) any claims against Buyer or any of its Affiliates (other than the Acquired Companies) unrelated to the transactions contemplated hereby, or (d) any other agreement that has been disclosed to Buyer in the Disclosure Schedules and which is not terminated or superseded at the Closing.

## 4.13 Supplemental Disclosure Schedules.

(a) The Sellers and the Company may, at any time and from time to time not less than five (5) Business Days prior to the Closing, by notice in accordance with the terms of this Agreement, supplement any one or more Parts of the Disclosure Schedule; provided, that, except as expressly provided below in this Section 4.13, such supplements shall not be deemed to amend the Disclosure Schedule or qualify the related representations and warranties of the Sellers or the Company herein. With respect to any item or matter that relates solely to actions, occurrences, facts, developments or events that (i) both arise and become known to the Sellers and the Company, as applicable, after the date hereof and would have been required or permitted to be set forth or described in the Disclosure Schedule had such matter existed as of the date hereof, (ii) does not arise from a breach of this Agreement, and (iii) either (A) is not material to the Acquired Companies, taken as a whole, or to the Sellers, or (B) arises out of or is attributable to any item described in parts (i) through (vi) of the definition of "Material Adverse Effect" in ARTICLE I, the item in such supplemental disclosure shall be deemed to amend the Disclosure Schedule and qualify the representations and warranties of the Sellers and the Company, as applicable; provided, however, that any updates to Parts 3.1(k) and 3.1(r) of the Disclosure Schedule with respect to actions, occurrences, facts, developments or events that arise after the date hereof shall be deemed to amend the Disclosure Schedule and qualify the related representations and warranties of the Company to the extent that the items reflected in such updates were not entered into in breach of this Agreement, without reference to clauses (i) and (iii) of this Section 4.13. Any other supplemental disclosures to the Disclosure Schedule shall not be deemed to amend the Disclosure Schedule, shall be made without effect to or qualification of any of the related representations and warranties of the Sellers and the Company contained in this Agreement, and shall have no effect on the right of the Buyer Indemnitees to indemnification.

(b) Each party will give prompt written notice to the other parties of any notice or development which has caused a material breach of or inaccuracy in any of its own representations and warranties in ARTICLE III. Subject to <u>Section 4.13(a)</u>, no such written notice shall be deemed to have amended and qualified the representations and warranties in ARTICLE III, or to have cured any material breach of or inaccuracy in a representation or warranty that otherwise might have existed.

# 4.14 Additional Covenants.

## (a) Financial Statements.

(i) From the date hereof until the Closing Date, the Sellers shall provide monthly management reports, to include monthly revenues, accounts receivable, accounts payable, other expenses, patient census and executive employee headcount, for the Acquired Companies

(collectively, the "Monthly Financial Statements"), prepared in accordance with past practices, no later than sixty (60) days after the end of each month.

(ii) The Sellers shall use commercially reasonable efforts to cooperate with Buyer's reasonable requests in order to facilitate preparation of the pro forma financial statements and historical financial statements required to be filed by the Buyer with the SEC, including providing such other financial information as is reasonably necessary to prepare pro forma financial statements required to be filed by the Buyer with the SEC. The Sellers shall cause the Company to request the registered accounting firm that audited the Company's 2017 consolidated financial statements deliver, at Buyer's sole cost and expense, all required consents for the inclusion of such financial statements in any filings Amedisys makes with the SEC.

(b) *Maintenance of Insurance*. The Sellers shall cause the Acquired Companies to maintain insurance coverage for the Acquired Companies through the Closing Date no less favorable in the aggregate than the insurance coverages for them in effect as of the date hereof.

## (c) *Employees and Integration.*

(i) To the extent permitted by applicable Law and Legal Requirements, prior to the Closing, the Sellers agree to cause the Acquired Companies to provide to Buyer information and access reasonably requested by it necessary to proceed with its pre-Closing integration processes, including data conversion (including patient, payor and employment data) and information technology installation. The Sellers agrees to cause the Acquired Companies to use commercially reasonable methods to assist Buyer in the transfer of data from the networks of the Acquired Companies (or their Affiliates, as applicable) to Buyer's network, provided that Buyer has executed any instruments necessary to ensure that such transfer complies with HIPAA and other Legal Requirements governing the transfer or sharing of confidential patient information. Buyer shall reimburse the Sellers and the Acquired Companies for any actual out-of-pocket expenses incurred by them in connection with the data conversion, data transfer and information technology installation obligations set forth in this Section 4.14(c) and such reimbursement will occur within fifteen (15) Business Days of the Sellers' presentation of invoices and reasonably detailed supporting documentation to Buyer.

(ii) In addition, the Sellers shall (A) at least thirty-five (35) days prior to the Closing but no more than forty-two (42) days prior to the Closing, and (B) at least fourteen (14) days prior to the Closing but no more than twenty-one (21) days prior to the Closing, provide Buyer with an updated true and correct list of (1) the name, state of residence, title, employer, length of service, hourly pay rate and annualized pay rate, fixed bonuses, and discretionary bonuses of each officer or employee of the Acquired Companies employed or engaged in the Business as of such date, and (2) any other form of compensation (other than salary, bonuses or customary benefits) paid or payable by any of the Acquired Companies to each such officer or employee for the most recent fiscal year. Further, during the period commencing on the date of delivery of the aforementioned list in the above clause (B) and ending upon the Closing, the Sellers shall cause the Acquired Companies not to hire any additional employees without the prior written approval of Buyer.

(iii) For a period of one (1) year following the Closing (or, if earlier, until the date of termination of the employee's employment with the Acquired Companies), Buyer presently intends to (or presently intends to cause the Acquired Companies to) provide (i) salary and target bonus opportunity (or base pay for non-salaried employees) for each employee of the Acquired Companies that is substantially comparable to the salary and target bonus opportunity (or base pay for non-salaried employees) paid to each such employee immediately prior to the Closing (payable by Buyer in accordance with Buyer's customary payroll practices) and (ii) retirement and welfare benefits that are no less favorable than those offered by Buyer to similarly situated employees; provided, however, nothing herein shall limit the Buyer's ability to implement operational changes at any time following the Closing (e.g., Agency closures, changes in staffing ratios, reductions in staffing, etc.). From and after the Closing Date, Buyer shall cause the service of each employee of the Acquired Companies with the Acquired Companies (or any predecessor entities thereof) to be recognized for purposes of eligibility to participate and vesting under any employee benefit plans, programs or arrangements maintained by Buyer and its Affiliates to the same extent recognized by the Acquired Companies prior to the Closing, except to the extent that such credit would result in a duplication of benefits. With respect to welfare benefit plans maintained by Buyer or its Affiliates in which employees of the Acquired Companies are eligible to participate, Buyer shall request that the related insurers waive any waiting periods otherwise applicable to such employees and their dependents under such welfare benefit plans to the extent satisfied under the applicable Benefit Plan prior to the Closing. For the avoidance of doubt, Buyer reserves the right to cause the Acquired Companies to terminate the employment of any "at will" employee at any time following the Closing, with or without cause, and to exercise all termination rights available to it under any employment agreement.

(iv) The provisions contained in clause (iii) this paragraph (d) are included for the sole benefit of the parties hereto and nothing in such clause (iii) whether express or implied, shall (A) create any third-party beneficiary or other rights in any other Person, including any current or former employees, any participant in any Benefit Plan, or any dependent or beneficiary thereof, including, without limitation, any right to employment or continued employment for any specified period or to be covered under or by any employee benefit plan or arrangement, or (B) be construed as an amendment, waiver or creation of or limitation on the ability to terminate any Benefit Plan or benefit plan or agreement of the Buyer or any of its Affiliates.

(d) *Bank Accounts*. At least three (3) Business Days prior to the Closing, the Sellers shall deliver to Buyer all documentation required to change authorizations for the accounts identified on <u>Part 3.1(u)(ii)</u> of the Disclosure Schedule to the individuals designated by Buyer, with such documentation to be held in escrow by Buyer and not released until the Closing.

(e) *Corporate Integrity Agreement*. From and after the Closing, Buyer shall cause the Acquired Companies to comply with the terms and requirements of the Corporate Integrity Agreement, including, without limitation, timely making all reports, repayments and filings with the Office of Inspector General required thereby; provided, however, nothing in this Section 4.14(f) shall relieve the Sellers from any liability hereunder associated with any failure of the Acquired Companies to comply with the terms and requirements of the Corporate Integrity Agreement prior to the Closing.

(f) Section 280G Stockholder Approval. Prior to the Closing Date, the Company shall cause a stockholder vote to be conducted to determine, in a manner reasonably satisfactory to Buyer, the right of any "disqualified individual" (as defined in section 280G(c) of the Code) to receive any and all payments (or other benefits) contingent on the consummation of the transactions contemplated by this Agreement (within the meaning of section 280G(b)(2)(A)(i) of the Code) to the extent necessary so that no payment received by such "disqualified individual" would be a "parachute payment" under section 280G(b) of the Code (determined without regard to section 280G(b)(4) of the Code) as a result of the consummation of the transactions contemplated by this Agreement, in a manner that satisfies the stockholder approval requirements under section 280G(b)(5)(B) of the Code and regulations promulgated thereunder. Consistent with section 280G(b)(5) and the regulations promulgated thereunder, such vote shall establish the "disqualified individual" would be a "parachute payment" under section 280G(b)(5)(B) of the Code and regulations promulgated thereunder. Consistent with section 280G(b)(5) and the regulations promulgated thereunder, such vote shall establish the "disqualified individual" would be a "parachute payment" under section 280G(b) of the Code. At the Closing, the Company shall provide Buyer stockholder resolutions or stockholder consents evidencing that such vote has been conducted in accordance with this <u>Section 4.14(i)</u> (such resolutions or consents, the "<u>280G Resolutions</u>").

(g) *Contracts.* Within thirty (30) days after the Closing, Sellers shall provide Buyer with a list of the Core Service Contracts, listed by provider number, together with true, correct and complete copies thereof.

(h) *Dissolution of Inactive Acquired Companies and Release of Encumbrances.* Sellers and the Company shall use commercially reasonable efforts to cause (i) those Acquired Companies listed on <u>Schedule 4.14(h)</u> to be dissolved prior to the Closing, and (ii) any and all Encumbrances on the assets of the Acquired Companies, including, but not limited, to those set forth in the Disclosure Schedule, to be released and discharged prior to the Closing.

**4.15** Payment and Performance Guarantee by Amedisys. Amedisys hereby absolutely, unconditionally and irrevocably guarantees, as a primary obligor and not merely a surety, to the Company and the Sellers (i) the performance of all obligations of Buyer under this Agreement to the extent such obligations are to be performed prior to or at the Closing and (ii) the due and punctual payment by Buyer of all payments of Purchase Price owing by Buyer pursuant to <u>Section 2.4</u> of the Agreement (clauses (i) and (ii) collectively, the "<u>Guaranteed Obligations</u>"). The guarantee by Amedisys set forth in clause (i) above shall terminate upon consummation of the Closing. The guarantee by Amedisys set forth in this <u>Section 4.15</u> shall continue to be effective or be reinstated, as the case may be, if at any time any payment or performance of the Guaranteed Obligations is rescinded or must otherwise be returned by the Sellers upon the insolvency, bankruptcy or reorganization of Amedisys or otherwise, all as though such payment had not been made. This is a guarantee of payment and performance, as applicable, and not of collection only. The Sellers shall not be required to make any demand upon, or to pursue or exhaust any right or remedy against, the Buyer prior to exercising its rights under this <u>Section 4.15</u>, and no delay or omission on the part of Sellers in exercising rights hereunder shall operate as a waiver or relinquishment of such rights or remedies. To the fullest extent permitted by law, Amedisys hereby expressly, unconditionally and irrevocably waives any and all rights or defenses arising by reason of any law which would otherwise require any election of remedies by the Sellers (but, for the avoidance of doubt, does not waive any

right or defense arising under any provision of this Agreement that requires election of remedies by the Sellers). Amedisys waives promptness, diligence, notice of the acceptance of this guarantee and of the Guaranteed Obligations, presentment, demand for payment, notice of non-performance, default, dishonor and protest, notice of incurrence of the Guaranteed Obligations and all other notices of any kind, any right to require the marshalling of assets of Buyer or any other Person interested in the transactions contemplated by this Agreement, all defenses which may be available by virtue of any valuation, stay, moratorium law or other similar law now or hereafter in effect and all suretyship defenses generally (other than defenses to the payment of the Guaranteed Obligations that are available to Buyer under this Agreement).

## ARTICLE V CONDITIONS OF PURCHASE

**5.1** <u>Conditions to Obligations of the Buyer</u>. The obligation of Buyer to consummate the transactions contemplated by this Agreement is expressly subject to the fulfillment or express written waiver of the following conditions on or prior to the Closing Date:

(a) <u>Obligations of the Sellers and the Company</u>. Each of the Sellers and the Company shall have performed, in all material respects, on or before the Closing Date, all covenants contained in this Agreement which by the terms hereof are required to be performed by him or it on or before the Closing Date.

(b) <u>Representations and Warranties of the Sellers and the Company</u>. Each representation and warranty set forth in <u>Sections 3.1</u> and <u>3.3</u> shall be true and correct at and as of the Closing as though then made except to the extent such representations and warranties by their terms speak of an earlier date in which case they shall be true and correct as of such earlier date, except where the failure of any such representation and warranty to be true and correct (whether as of the Closing or such earlier date) does not have a Material Adverse Effect. For purposes of this <u>Section 5.1(b)</u>, all materiality, Material Adverse Effect and similar qualifiers in the representations and warranties in Section 3.1 and 3.3 shall be disregarded.

(c) <u>Compliance Certificate</u>. The Sellers shall have delivered to Buyer a certificate, dated as of the Closing Date, to the effect that the conditions set forth in <u>Sections 5.1(a)</u>, <u>5.1(b)</u> and <u>5.1(d)</u> have been satisfied as of the Closing together with copies of the resolutions adopted by the Sellers, in their capacities as shareholders of the Company, and the Board of Directors of the Company duly authorizing the transactions contemplated hereby.

(d) <u>No Prohibition</u>. There shall not be any order or injunction of a court of competent jurisdiction or governmental agency (whether temporary, preliminary or permanent) in effect preventing, restraining or invalidating the consummation of the material transactions contemplated hereby.

(e) <u>Governmental Filings; Third Party Consents</u>. All Governmental Entity and other third party filings or registrations with, notifications to, or authorizations, consents or approvals set forth on <u>Schedule I</u> shall have been made or obtained, as applicable.

(f) <u>Restrictive Covenant Agreement</u>. The Sellers shall have delivered to Buyer a restrictive covenant agreement for Heching, in the form of <u>Exhibit E</u>.

(g) <u>Good Standing Certificates</u>. The Sellers shall have delivered to Buyer a Certificate of Status (or its equivalent) in regard to each Acquired Company that holds or operates under any Permit or has other assets and operations issued by the Secretary of State of its state of organization (dated as of a recent date).

(h) <u>Section 2.3 Deliverables</u>. The Sellers shall have delivered the certificates, Distribution Schedule and release letters, if any, required by <u>Sections 2.3(b)</u> and (c).

(i) Escrow Agreement. The Sellers and the Escrow Agent shall have entered into the Escrow Agreement.

(j) <u>Other Transaction Documents</u>. The Sellers shall have delivered (or caused to be delivered) the following documents to the Buyer:

(i) the stock and unit powers substantially in the same forms attached hereto as  $\underline{\text{Exhibit F}}$  (the "Stock  $\underline{\text{Powers}}$ "), duly executed in blank by the Sellers, together with the original stock and/or unit certificate(s), if applicable, evidencing all of the outstanding shares of Company Common Stock and Capital Stock of the Non-Wholly Owned Subsidiaries;

(ii) the resignations, effective as of the consummation of the transactions, or evidence of removal effective as of the consummation of the transactions, of all officers and all members of the Board of Directors of each of the Acquired Companies;

(iii) a certificate of non-foreign status, in a mutually agreeable form, from and duly executed by each Seller, which complies with Section 1445 of the Code;

each Seller;

(iv) an IRS Form W-9 establishing an exemption from backup withholding, from and duly executed by

(v) an IRS Form 8023, executed by each Seller, and such other documentation as may be reasonably requested by Buyer in connection with the Section 338(h)(10) Election;

(vi) a confirmation signed by Heching's present spouse (and any former spouse) that, except in connection with the Trust, she has no interest in or claim to the Shares and the sale proceeds thereof, as the case may be, in form and content reasonably acceptable to Buyer; and

(vii) a certificate in the form of <u>Schedule 5.1(j)(vii)</u>, duly executed by each Seller; and

(viii) the 280G Resolutions.

**5.2** <u>Conditions to Obligations of the Sellers</u>. The obligation of the Sellers to consummate the transactions contemplated by this Agreement is expressly subject to the fulfillment or express written waiver of the following conditions on or prior to the Closing Date:

(a) <u>Representations and Warranties of Buyer and Amedisys</u>. Each of the representations and warranties of Buyer contained in <u>Section 3.2</u> and of Amedisys in <u>Section 3.4</u> shall be true and correct in all material respects at and as of the Closing as though then made except to the extent such representations and warranties by their terms speak of an earlier date in which case they shall be true and correct in all material respects as of such earlier date (except for any representations and warranties that are qualified by the concept of materiality, which shall be true and correct in all respects).

(b) <u>Obligations of the Buyer and Amedisys</u>. Buyer and Amedisys shall have performed in all material respects, on or before the Closing Date, all covenants contained in this Agreement which by the terms hereof are required to be performed by Buyer or Amedisys on or before the Closing Date.

(c) <u>Compliance Certificate</u>. Buyer and Amedisys shall have delivered to the Company a certificate, dated as of the Closing Date, to the effect that the conditions set forth in <u>Sections 5.2(a)</u>, <u>5.2(b)</u> and <u>5.2(d)</u> have been satisfied as of the Closing, together with copies of the resolutions adopted by the Board of Directors of Amedisys duly authorizing the transactions contemplated hereby.

(d) <u>No Prohibition</u>. There shall not be any order or injunction of a court of competent jurisdiction or governmental agency (whether temporary, preliminary or permanent) in effect preventing, restraining or invalidating the consummation of the material transactions contemplated hereby.

(e) <u>Governmental Filings; Third Party Consents</u>. All Governmental Entity and third party filings or registrations with, notifications to, or authorizations, consents or approvals set forth on <u>Schedule I</u> shall have been made or obtained, as applicable.

(f) <u>Closing Payments</u>. Buyer shall have made the payments contemplated by <u>Section 2.4</u>.

(g) <u>Other Agreements</u>. Buyer and the Escrow Agent shall have entered into the Escrow Agreement.

## ARTICLE VI <u>TERMINATION</u>

6.1 <u>Termination of Agreement</u>. This Agreement may be terminated at any time prior to the Closing Date as follows:

(a) by mutual written consent of Buyer and the Sellers;

(b) by Buyer, if (i) any of the representations and warranties of the Company or the Sellers set forth in this Agreement shall have become untrue, or if the Company or the Sellers shall have breached or failed to perform any of their or its obligations, covenants or agreements under this Agreement to the extent required by Section 5.1(a), and (ii) such breach or failure to perform or be true, if occurring or continuing at the Closing, (A) would result in the failure of any of the conditions set forth in Section 5.1 and (B) cannot be or has not been cured by the Company or the Sellers, or waived by Buyer. within thirty (30) days after Buyer gives the Sellers written notice identifying in reasonable detail such breach or failure to perform or be true;

(c) by the Sellers, if (i) any of the representations and warranties of Buyer or Amedisys set forth in this Agreement shall have become untrue, or if Buyer or Amedisys shall have breached or failed to perform any of their obligations, covenants or agreements under this Agreement to the extent required by <u>Section 5.2(b)</u>, and (ii) such breach or failure to perform or be true, if occurring or continuing at the Closing, (A) would result in the failure of any of the conditions set forth in <u>Section 5.2</u> and (B) cannot be or has not been cured by Buyer or Amedisys, or waived by the Sellers, within thirty (30) days after the Sellers gives Buyer written notice identifying in reasonable detail such breach or failure to perform or be true;

(d) by either Buyer or the Sellers, if any court or other Governmental Entity having competent jurisdiction has issued an order, decree or ruling, whether temporary, preliminary or permanent in effect restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated by this Agreement; <u>provided</u> that the right to terminate this Agreement pursuant to this <u>Section 6.1(d)</u> shall not be available to any party whose breach of any provision of this Agreement, results in such order, decree or ruling; or

(e) by either Buyer or the Sellers, if the Closing shall not have occurred within ninety (90) days from the date of this Agreement (the "<u>Outside Date</u>"); <u>provided</u>, that if as of the Outside Date any of the conditions set forth in <u>Section 5.1(e)</u> (solely to the extent such condition has not been satisfied with respect to authorization, consent or approval of, a Governmental Entity) shall not have been satisfied or waived, the Outside Date shall, unless otherwise agreed in writing by Buyer and the Company, be extended on one occasion for a period of ninety (90) days, and such date, as so extended, shall be the Outside Date; <u>provided</u>, <u>further</u>, that (i) the right to terminate this Agreement pursuant to this <u>Section 6.1(e)</u> shall not be available to any party whose breach of any provision of this Agreement, or whose representations and warranties shall have become untrue, results in the failure of the Closing to be consummated by the Outside Date, and (ii) neither party shall be permitted to terminate this Agreement pursuant to this <u>Section 6.1(e)</u> during the Closing Period.

**6.2** <u>Effect of Termination</u>. In the event of termination of this Agreement by a party hereto pursuant to <u>Section 6.1</u> hereof, written notice thereof shall forthwith be given by the terminating party to the other parties hereto specifying the provision hereof pursuant to which such termination is made, and this Agreement shall thereupon terminate and become void and have no effect, except that the provisions of <u>Section 4.2</u> (Publicity), <u>Section 4.3</u> (Confidentiality), <u>Section 4.15</u> (Payment and Performance Guarantee by Amedisys), this <u>Section 6.2</u> (Effect of Termination)

and ARTICLE VIII (Additional Operative Provisions) shall survive the termination of this Agreement; <u>provided</u>, <u>however</u>, that such termination shall not relieve any party hereto of any liability for any willful and material breach of this Agreement prior to such termination.

## ARTICLE VII INDEMNIFICATION

**7.1** <u>Survival</u>. The representations, warranties and covenants contained in this Agreement and in any certificate delivered at the Closing shall survive the Closing and terminate at 5:00 PM Eastern time on the date that is twelve (12) months after the Closing Date; <u>provided</u>, <u>however</u>, that:

(a) all covenants and agreements contained herein which by their terms are to be performed solely, or which prohibit actions, subsequent to the Closing, shall survive the Closing in accordance with their terms or, if no term is specified, in accordance with the applicable statute of limitations;

(b) both (i) the representations and warranties contained in clauses (i) (Taxes), (l) (Employees and Compensation), (m) (Employee Benefit Plans), (s) (Environmental Matters) and (t) (Healthcare Matters) of Section 3.1 (collectively, the "Special Representations"), and (ii) all rights and obligations under Sections 7.2(a)(iii), 7.2(a)(iv), 7.2(a)(v), 7.2(a)(v),

(c) the representations and warranties contained in clauses (a) (Due Organization and Good Standing), (b) (Authorization, Execution and Delivery; Valid and Binding Agreement), (d) (Capital Structure; Subsidiaries), and (p) (Brokers' Fees) of Section 3.1, those contained in clauses (a) (Due Organization and Good Standing), (b) (Authorization and Execution) and (f) (Brokers' Fees) of Section 3.2, those contained in clauses (a) (Execution), (e) (Brokers' Fees) and (f) Ownership of Section 3.3, and those contained in clauses (a) (Due Organization and Good Standing), and (b) (Authorization and Execution) of Section 3.4 (collectively, the "Fundamental Representations"), shall survive the Closing and terminate at 5:00 PM Eastern time on the earlier of (i) the date that is seven (7) years after the Closing Date, or (ii) the date that is thirty (30) days after the expiration of the applicable statute of limitations (as applicable in each of the foregoing cases in this Section 7.1, the "Survival Expiration Date").

No claim for breach of any representation, warranty, covenant or agreement, or claim for indemnification under <u>Section 7.2(a)</u> or <u>Section 7.3(a)</u>, as applicable, may be brought after the applicable Survival Expiration Date, except for claims (1) of which the Sellers have been notified in writing with reasonable specificity by Buyer prior to the applicable Survival Expiration Date, or (2) of which Buyer has been notified in writing with reasonable specificity by the Sellers prior to the applicable Survival Expiration Date.

## 7.2 Indemnification of the Buyer Indemnitees.

(a) If the Closing occurs, subject to the terms of this ARTICLE VII, the Sellers shall, on a joint and several basis, indemnify and hold harmless the Acquired Companies, Buyer,

Amedisys and its Affiliates (collectively, the "<u>Buyer Indemnitees</u>") from and against, and pay to the Buyer Indemnities the amount of, or reimburse the Buyer Indemnitees for, any Losses incurred by the Buyer Indemnitees by reason of:

(i) any inaccuracy or breach of any of the representations or warranties of the Company or the Sellers specifically set forth in Sections 3.1 or 3.3 or contained in any certificate delivered at the Closing by the Company or the Sellers pursuant to this Agreement (with any Material Adverse Effect qualifier contained in any such certificate being disregarded for the purposes of this Section 7.2(a)(i));

(ii) the failure of the Company or the Sellers to perform any of their or its covenants or agreements contained herein required to be performed prior to the Closing, or the failure of the Sellers to perform any covenant or agreement set forth herein which by its terms is to be performed after the Closing;

(iii) any claim by a current or former holder of any security of any Acquired Company, in its capacity as such, challenging this Agreement or the other transactions contemplated hereby;

(iv) the failure of (A) any portion of the Company Expenses or the Indebtedness of the Acquired Companies outstanding as of the Closing to be paid at Closing (unless the failure arises from a breach by the Buyer of its obligations under Sections 2.4(a)(i) or 2.4(a)(iii)) or (B) Closing Cash to be equal to or greater than the Closing Cash Target (except to the extent that the Purchase Price paid at Closing was reduced to reflect all or a portion of such deficiency);

(v) any Medicare and Medicaid Recoupment Claims (provided, however, in the case of any such claims relating to hospice cap liabilities of the Acquired Companies, such claims shall be limited to the extent they exceed the accrual for hospice cap liabilities set forth on Part 7.2(a)(v) of the Disclosure Schedule);

(vi) any Mandatory Repayments (without regard to any matters disclosed on the Disclosure Schedule);

(vii) any Healthcare Liability Claims (without regard to any matters disclosed on the Disclosure Schedule); and

(viii) any and all amounts required to be paid by the Acquired Companies after the Closing Date pursuant to those certain existing and pending settlements with the Government Entities listed on <u>Part 7.2(a)(viii)</u> of the Disclosure Schedule, including any fines, penalties or interest in connection therewith (solely to the extent such fines, penalties or interest were incurred or are attributable to the period prior to the Closing Date), to the extent such amounts, in the aggregate exceed \$5,300,000.

For purposes of determining both (1) whether the Company or the Sellers, as applicable, has breached any of his or its representations and warranties in Sections 3.1 or 3.3 (or any certificate) or whether the Company or the Sellers has breached any covenants or agreements herein, and (2) the amount of Losses suffered or incurred by any Buyer Indemnitee by reason of such breach, qualifications therein referring to "material", "Material Adverse Effect" and other qualifications of similar import or effect shall be disregarded.

(b) In addition to the limitations in Section 7.1, the right of the Buyer Indemnitees to submit claims pursuant to Section 7.2(a) is subject to the following limitations:

(i) no right to be indemnified or held harmless shall exist and no claim may be made against the Sellers under <u>Section 7.2(a)(i)</u> (except for Fundamental Representations and Fraud, neither of which shall be subject to the Tipping Basket or the Threshold (as each such term is hereinafter defined)) unless and until the aggregate amount of all Losses incurred by the Buyer Indemnitees in respect of claims thereunder exceeds \$2,800,000 (the "<u>Tipping Basket</u>"), after which point the Sellers shall be obligated to indemnify the Buyer Indemnitees from and against all such Losses from dollar one, except in the case of the Special Health Care Representations, which are subject to the Threshold;

(ii) no right to be indemnified or held harmless shall exist and no claim may be made against the Sellers (x) under Section 7.2(a)(i) for any inaccuracy or breach of any of the Special Health Care Representations, or (y) under Sections 7.2(a)(v), 7.2(a)(v) and 7.2(a)(vii), unless and until the aggregate amount of all Losses incurred by the Buyer Indemnitees in respect of claims thereunder exceeds \$2,800,000 (the "Threshold"), after which point the Sellers shall be obligated to indemnify the Buyer Indemnitees from and against all such Losses in excess of the Threshold;

(iii) except for indemnification claims made against the Sellers for (w) any inaccuracy or breach of any of the Fundamental Representations, (x) any failure of the Sellers to perform any covenant or agreement set forth herein which by its terms is to be performed after the Closing, (y) any failure by any of the Sellers to pay any of his or its Income Taxes accruing prior to the Closing and (z) Fraud, the sole source of payment for all indemnification claims under <u>Section 7.2(a)</u> shall be limited to the amount of, and in no event exceed, the remaining amount of the Escrow Fund held by the Escrow Agent;

(iv) except for indemnification claims made against the Sellers for (x) any failure of the Sellers to perform any covenant or agreement set forth herein which by its terms is to be performed after the Closing, (y) any failure by any of the Sellers to pay any of his or its Income Taxes accruing prior to the Closing, or (z) Fraud, the Escrow Fund shall be the Buyer Indemnitees' first recourse for all indemnification claims made against them, and the Sellers shall not be responsible to pay for any such indemnification claim until the Escrow Fund has been reduced to zero dollars (\$0); *provided*, however, any indemnification claims made under

<u>Section 7.2(a)(iv)(B)</u>, shall first be paid from the Escrow Fund, for an amount up to \$500,000, and second, from the Sellers with respect to any remaining difference.

(v) except for indemnification claims made against the Sellers for Fraud, the aggregate liability of the Sellers for all indemnification claims shall be limited to, and shall in no event exceed, \$280,000,000;

(vi) no claim shall be made with respect to Losses arising out of any breach of the representations or warranties contained in <u>Section 3.1</u> to the extent that there has been a corresponding reduction in the Purchase Price and such Losses shall be disregarded in determining whether the Tipping Basket or Threshold apply; and

(vii) notwithstanding anything herein to the contrary, the, Sellers shall not be required to indemnify the Buyer Indemnitees in respect of any Losses consisting of or relating to (x) Taxes with respect to any taxable period, or, under the principles of <u>Section 4.11(c)</u>, the portion thereof, beginning after the Closing Date or (y) any Tax withholding requirements imposed following Closing or (z) any Tax Return preparation or filing requirements imposed following Closing, in each case as a result of any breach of the representations and warranties under <u>Section 3.1(i)</u> (Taxes) other than the representations and warranties set forth in the last sentence of paragraph (ix), or paragraphs (x), (xi) or (xii), of such <u>Section 3.1(i)</u> (Taxes).

## 7.3 Indemnification of the Sellers.

(a) If the Closing occurs, subject to the terms of this ARTICLE VII, Buyer agrees to indemnify and hold harmless the Sellers (collectively, the "<u>Seller Indemnitees</u>") from and against, and pay to the Seller Indemnities the amount of, or reimburse the Seller Indemnitees for, any Losses incurred by the Seller Indemnitees by reason of (i) any inaccuracy or breach of any of the representations or warranties of Buyer or Amedisys specifically set forth in <u>Section 3.2</u> or <u>Section 3.4</u>, as applicable, or contained in any certificate delivered at the Closing by Buyer pursuant to this Agreement and (ii) the failure of Buyer or Amedisys to perform any of its covenants or agreements contained herein required to be performed prior to the Closing, or the failure of Buyer to perform any covenant or agreement set forth herein which by its terms is to be performed after the Closing.

(b) In addition to the limitations in Section 7.1, the right of the Seller Indemnitees to submit claims pursuant to Section 7.3(a) is subject to the following limitations:

(i) no right to be indemnified or held harmless shall exist and no claim may be made against Buyer under Section 7.3(a)(i) (except for (x) Fundamental Representations and (y) for Fraud, neither of which shall be subject to the Tipping Basket) unless and until the aggregate amount of all Losses incurred by the Seller Indemnitees in respect of claims thereunder exceeds the Tipping Basket, after which point Buyer shall be obligated to indemnify the Seller Indemnitees from and against all such Losses from dollar one.

(ii) except for indemnification claims made against the Buyer for (x) any inaccuracy or breach of any of the Fundamental Representations, (y) any failure of Buyer or Amedisys to perform any covenant or agreement set forth herein which by its terms is to be performed after the Closing, and (z) Fraud, the aggregate liability of Buyer for all indemnification claims pursuant to Section 7.3(a) shall be limited to the amount of, and in no event exceed, the initial amount of the Escrow Fund; and

(iii) except for indemnification claims made against the Buyer for Fraud, the aggregate liability of Buyer for all indemnification claims shall be limited to, and shall in no event exceed, \$280,000,000.

### 7.4 Indemnification Procedures and Related Provisions and Additional Limitations.

(a) Any party entitled to make a claim for indemnification hereunder (an "<u>Indemnified Party</u>") shall promptly notify (a "<u>Claim Notice</u>") the indemnifying party (the "<u>Indemnifying Party</u>") of the claim in writing upon learning of such claim or the facts constituting such claim, describing the claim in reasonable detail, the amount thereof, and the basis therefor, and in the case of any claims based on a claim by a third party, attach all notices, pleadings and other documents or instruments served upon or received by the Indemnified Party with respect thereto; <u>provided</u> that no delay on the part of the Indemnified Party in giving any such Claim Notice shall relieve the Indemnifying Party of any indemnification obligation hereunder except to the extent that the Indemnifying Party is prejudiced by such delay. The Indemnifying Party's failure to respond to a Claim Notice within thirty (30) days of its delivery will be deemed to be a denial of the claim.

(b) In the event that Buyer has Knowledge of any audit, challenge, claim or charge that would reasonably be expected to give rise to indemnification by the Sellers for any Mandatory Repayments, Medicare and Medicaid Recoupment Claims, or Healthcare Liability Claims incurred by the Buyer Indemnitees, without taking the Threshold into account (a "<u>Notice Matter</u>"), the Buyer Indemnitees shall provide written notice to the Sellers along with all available documentation relating to such matter. Within ten (10) Business Days of their receipt of such documentation, the Sellers shall provide Buyer with comments relevant to reducing or eliminating liability related to the Notice Matter. Buyer will consider any reasonable comments from the Sellers relating to such Notice Matter, and will use commercially reasonable efforts to reduce liability related to the Notice Matter, including submitting responses and/or appeals via the relevant Government Program's or Private Program's formal appeals processes in accordance with the comments from Sellers. Buyer will not cease its pursuit of available appeals processes, or enter into a settlement or compromise with respect to any Notice Matter, except with the prior written consent of Sellers (which consent shall not be unreasonably withheld, conditioned or delayed). Should the amount of Losses due to Mandatory Repayments, Medicare and Medicaid Recoupment Claims or Healthcare Liability Claims be reduced subsequent to any payment of Losses hereunder, the Buyer Indemnitee shall replenish the Escrow Fund or reimburse the Sellers, as applicable hereunder, for the amount thereof previously paid to the Buyer Indemnitee program.

(c) In the event that any Action is commenced by a third party involving a claim (an "Asserted Liability") for which an Indemnified Party is entitled to be indemnified and held harmless hereunder by an Indemnifying Party, the Indemnifying Party shall have thirty (30) days from its receipt of the Claim Notice (the "Notice Period") to notify the Indemnified Party in writing whether or not the Indemnifying Party desires, at the Indemnifying Party's sole cost and expense and with counsel reasonably acceptable to the Indemnified Party, to defend against such Asserted Liability. If the Indemnifying Party undertakes to defend against such Asserted Liability, which shall require the Indemnifying Party to (i) timely notify the Indemnified Party in writing that it has assumed such defense and that it will indemnify the Indemnified Party against any Losses arising out of such Asserted Liability (subject to any other limitations set forth in this Article VII), and (ii) provide evidence of its financial wherewithal to do so should the Indemnifying Party's liability exposure for the Asserted Liability exceed the remaining balance of the Escrow Fund, the Indemnifying Party shall use its commercially reasonable efforts to defend and protect the interests of the Indemnified Party with respect to such Asserted Liability. The Indemnified Party, at its own expense, shall be entitled to participate in the defense of such claim and to employ counsel of its choice for such purpose. If the Indemnifying Party does not undertake to defend the Asserted Liability (or is not entitled to defend the Asserted Liability under Section 7.4(e)), the Indemnifying Party shall be entitled to participate in the defense of such claim and to employ counsel of its choice for such purpose at its own expense. The Indemnified Party shall not consent to the entry of any judgment or enter into any settlement with respect to any Asserted Liability without the prior written consent of the Indemnifying Party (which consent shall not be unreasonably withheld, conditioned or delayed). The Indemnifying Party shall not consent to the entry of any judgment or enter into any settlement with respect to any Asserted Liability without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld, conditioned or delayed), unless such settlement or judgment (x) provides for a full and unconditional release of the Indemnified Party without any admission of wrongdoing or liability, (y) does not require the Indemnified Party to make any monetary payments other than those for which it will be reimbursed in full contemporaneously therewith (subject to the limitations set forth in this ARTICLE VII), and (z) does not include any non-monetary relief.

(d) The Indemnified Party and the Indemnifying Party agree to make available to each other, their counsel and other representatives, all non-privileged information and documents available to them which relate to an Asserted Liability. The Indemnified Party and the Indemnifying Party shall, and shall cause the Acquired Companies and their employees to, render to each other such assistance and cooperation as may reasonably be required to ensure the proper and adequate defense of an Asserted Liability.

(e) Notwithstanding the foregoing terms of <u>Section 7.4(b)</u>, the Indemnifying Party shall not be entitled to assume or maintain control of the defense of any Asserted Liability if (i) the Asserted Liability is a criminal proceeding, action, indictment, allegation or investigation, (ii) the Indemnifying Party has failed to defend or is failing to defend in good faith the Asserted Liability that is not cured within a reasonable time after receiving written notice from the Indemnified Party specifying in reasonable detail the manner in which the Indemnifying Party has so failed or is failing, or (iii) the relief sought in respect of the Asserted Liability includes non-monetary relief

(other than a general boilerplate request for such other and further relief as the court deems just and proper).

(f) To the extent there is any conflict between this <u>Section 7.4</u> and the provisions of <u>Section 4.11</u> with respect to the procedures involving any Tax Claim, <u>Section 4.11</u> shall control.

(g) If any amounts from the Escrow Fund are released to a Buyer Indemnitee pursuant to a claim under Section 7.2(a)(viii) of this Agreement ("Settlement Payment"), and the total amounts available to Buyer Indemnitees in the Escrow Fund are less than \$28,000,000 after giving effect to such Settlement Payment, Heching shall promptly replenish the Escrow Fund by the lesser of (i) the amount necessary to increase the total of the Escrow Fund to \$28,000,000, or (ii) an amount equal to the Settlement Payment.

7.5 Insurance and Other Third Party Recoveries. In calculating the monetary thresholds set forth in Sections 7.2(b) (i), 7.2(b)(ii) and 7.3(b)(i), as applicable, and the amounts otherwise payable to an Indemnified Party, the amount of any indemnified Losses shall be computed net of (i) payments actually recovered by the Indemnified Party under any insurance policy with respect to such Losses (after giving effect to any applicable deductible or retention, any out of pocket costs incurred by the Indemnified Party, and any increase of premium in connection therewith), and (ii) any amounts actually recovered by the Indemnified Party from any other Person with respect to such Losses (after giving effect to any out of pocket costs incurred by the Indemnified Party in connection therewith) including any Tax savings or benefits with respect to such Losses actually realized by the Indemnified Party which savings or benefits is directly attributable to the deductibility of such Loss (as determined on a "with and without" basis), in each case, net of any out of pocket costs incurred by the Indemnified Party in obtaining, receiving or accruing such Tax savings or benefits. In the event an insurance recovery or third party recovery relating to an indemnification payment is received after the Indemnifying Party has made an indemnification payment under this Agreement that did not take into account such insurance recovery or third party recovery, the Indemnified Party shall promptly pay the Indemnifying Party an amount equal to the lesser of such insurance recovery or third party recovery (net of any applicable deductible or retention, any out of pocket costs incurred by the Indemnified Party, and any increase of premium in connection therewith, as applicable) and the amount of the related indemnification payment. Each Indemnified Party shall use commercially reasonable efforts to mitigate the amount of Losses for which it may be entitled to indemnification hereunder.

### 7.6 Manner of Payment.

(a) Escrow Period; Distribution upon Twelve Month Anniversary of Closing and upon Termination of Escrow Period. The Escrow Account shall be in existence as of the Closing and, except as hereinafter provided, shall terminate at 11:59pm, Central Time, on the two (2) year anniversary of the Closing Date (such termination date, the "Second Release Date", and such time period, the "Escrow Period"). Upon the twelve (12) month anniversary of the Closing (the "First Release Date"), Nine Million Five Hundred Thousand (\$9,500,000), less (i) all amounts, if any, previously paid to a Buyer Indemnitee out of the Escrow Fund to satisfy indemnification claims pursuant to ARTICLE VII prior to the First Release Date and (ii) 100% of the amount of any unsatisfied and unresolved claims specified in any Claim Notice delivered in good faith to the

Escrow Agent (each, an "<u>Unresolved Claim</u>") prior to the First Release Date, shall be distributed to the Sellers. Upon the Second Release Date, the remainder of the Escrow Amount, less 100% of the amount of any Unresolved Claims delivered in good faith to the Escrow Agent prior to the Second Release Date, will be distributed to the Sellers <u>provided</u>, <u>however</u>, that the Escrow Period shall not terminate with respect to an Unresolved Claim. As soon as all such claims have been resolved in accordance with this Agreement and the Escrow Agreement, the Escrow Agent shall deliver to the Sellers the remaining portion of the Indemnity Escrow Amount, if any, not required to satisfy such Unresolved Claims.

(b) *Claims for Indemnification Outside of Escrow.* If either (i) the Buyer is permitted to assert an indemnification claim against the Sellers without first depleting the Escrow Fund, as permitted by <u>Section 7.2(b)(iv)</u>, or (ii) the Escrow Fund is no longer available or is not adequate to satisfy any Losses that are subject to indemnification pursuant to <u>Section 7.4</u>, subject to the limitations set forth in <u>Section 7.2(b)(iii)</u>, then after any final decision, judgment or award shall have been rendered by a Governmental Authority of competent jurisdiction and the expiration of the time in which to appeal therefrom, or a settlement shall have been consummated, or Buyer and the Sellers shall have reached an agreement, in each case with respect to an indemnifiable claim hereunder, the Buyer Indemnitee(s) shall forward to the Sellers notice of any sums due and owing by the Sellers pursuant to this Agreement with respect to such matter and the Sellers shall pay all of such remaining sums so due and owing to the Buyer Indemnitee(s) within fifteen (15) days after the final decision, judgment or award was rendered by the Governmental Authority of competent jurisdiction of the time in which to appeal thereform, or a settlement was consummated, or the Buyer and the Sellers notice of award was rendered by the Governmental Authority of competent jurisdiction 7.6. Any such indemnification obligations shall be paid by the Sellers to the Buyer Indemnitee(s) within fifteen (15) days after the final decision, judgment or award was rendered by the Governmental Authority of competent jurisdiction and the expiration of the time in which to appeal thereform, or a settlement was consummated, or the Buyer and the Sellers reached an agreement.

(c) Each of the Parties hereto agrees to deliver such notices to the Escrow Agent as are required to effectuate the disbursements from the Escrow Account as contemplated by this Agreement and the Escrow Agreement.

7.7 <u>Remedies Exclusive</u>. EXCEPT FOR FRAUD AND EQUITABLE RELIEF, THE REMEDIES PROVIDED IN THIS <u>ARTICLE VII</u> SHALL BE THE EXCLUSIVE REMEDIES OF THE PARTIES HERETO AND THE OTHER INDEMNITEES NAMED HEREIN AND THEIR HEIRS, SUCCESSORS AND ASSIGNS AFTER THE CLOSING WITH RESPECT TO THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT INCLUDING, WITHOUT LIMITATION, ANY BREACH OR NON-PERFORMANCE OF ANY REPRESENTATION, WARRANTY, COVENANT OR AGREEMENT CONTAINED HEREIN.

**7.8** <u>**Tax Treatment of Indemnity Payments**</u>. For all Tax purposes, the parties agree to treat indemnity payments made pursuant to this Agreement as an adjustment to the Purchase Price to the extent permitted by applicable Tax Law.

**7.9** <u>No Circular Recovery</u>. Notwithstanding anything to the contrary herein, the Sellers hereby agree they will not make any claim for indemnification against the Buyer Indemnitees by reason of the fact that the Sellers were a controlling person, director/manager, officer, employee or representative of any of the Acquired Companies or was serving as such for another Person at the request of any of the Acquired Companies (whether such claim is for Losses of any kind or otherwise

and whether such claim is pursuant to any Legal Requirement, contract or otherwise) with respect to any claim brought by a Buyer Indemnitee under this Agreement or otherwise relating to this Agreement or any of the transactions contemplated hereby. With respect to any claim brought by a Buyer Indemnitee under this Agreement or otherwise relating to this Agreement or the transactions contemplated hereby, the Sellers expressly waive any right of subrogation, contribution, advancement, indemnification or other claim against the Acquired Companies with respect to any amounts owed by the Sellers hereunder.

## ARTICLE VIII ADDITIONAL OPERATIVE PROVISIONS

**8.1** Assignment; Binding Effect. This Agreement and the rights hereunder are not assignable unless such assignment is consented to in writing by Buyer and the Sellers and, subject to the preceding clause, this Agreement and all the provisions hereof shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors or permitted assigns. Notwithstanding the foregoing, (a) Buyer may make the following assignments in its sole discretion: (i) Buyer may (at any time prior to the Closing) assign, in whole or in part, its rights and obligations pursuant to this Agreement to one or more of its Affiliates (including Affiliates which may be organized subsequent to the date hereof) so long as such assignment would not substantially delay or be likely to cause the rejection of the requisite regulatory approvals; (ii) Buyer may assign its rights under this Agreement, in whole or in part, to any subsequent purchaser of the Capital Stock of the Company; provided, however, no such assignment shall release Buyer or Amedisys of any of their obligations hereunder unless Sellers shall otherwise consent in writing; and (b) either Seller may assign, in whole or in part, its or his rights under this Agreement to any trust, foundation or other Person for estate planning or charitable purposes; provided, however, no such assignment shall release such Sellers of any of their obligations hereunder unless Buyer shall otherwise consent in writing. Any such permitted assignment shall be effected pursuant to a form of assignment and assumption agreement that is reasonably acceptable to the non-assigning parties.

**8.2** <u>Choice of Law</u>. This Agreement and all claims arising from and relating to this Agreement and the transactions contemplated hereby shall be governed by and interpreted and enforced in accordance with the Laws of the State of Delaware, without regard to the conflicts of Laws rules thereof.

**8.3** <u>Consent to Jurisdiction and Service of Process; Waiver of Jury Trial</u>. ALL JUDICIAL PROCEEDINGS BROUGHT AGAINST THE PARTIES ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OBLIGATIONS HEREUNDER, OR THE TRANSACTIONS CONTEMPLATED HEREBY, SHALL BE BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION SITUATED IN WILMINGTON, DELAWARE. BY EXECUTING AND DELIVERING THIS AGREEMENT, THE PARTIES, IRREVOCABLY (A) ACCEPT GENERALLY AND UNCONDITIONALLY THE EXCLUSIVE JURISDICTION AND VENUE OF SUCH COURTS, (B) WAIVE ANY OBJECTIONS WHICH

SUCH PARTY MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY OF THE AFORESAID ACTIONS OR PROCEEDINGS ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, ANY OBLIGATIONS HEREUNDER OR THE TRANSACTIONS CONTEMPLATED HEREBY BROUGHT IN THE COURTS REFERRED TO IN CLAUSE (A) ABOVE AND HEREBY FURTHER IRREVOCABLY WAIVE AND AGREE NOT TO PLEAD OR CLAIM IN ANY SUCH COURT THAT SUCH ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM, (C) AGREE THAT SERVICE OF ALL PROCESS IN ANY SUCH PROCEEDING IN ANY SUCH COURT MAY BE MADE BY REGISTERED OR CERTIFIED MAIL, RETURN RECEIPT REQUESTED, TO SUCH PARTY AT THEIR RESPECTIVE ADDRESSES PROVIDED IN ACCORDANCE WITH SECTION 8.4, AND (D) AGREE THAT SERVICE AS PROVIDED IN CLAUSE (C) ABOVE IS SUFFICIENT TO CONFER PERSONAL JURISDICTION OVER SUCH PARTY IN ANY SUCH PROCEEDING IN ANY SUCH COURT, AND OTHERWISE CONSTITUTES EFFECTIVE AND BINDING SERVICE IN EVERY RESPECT. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.

**8.4** <u>Notices</u>. All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given (a) if delivered personally or actually received, as of the date received, (b) if delivered by certified mail, return receipt requested, five (5) Business Days after being mailed, (c) if delivered by a nationally recognized overnight delivery service, one (1) Business Day after being deposited with such delivery service for next Business Day delivery, (d) if sent via facsimile, upon confirmation of transmission, or (e) if sent via electronic mail, as of the date received (so long as the same is sent via another method of delivery provided for herein within three (3) days of such transmittal), to such party at its address set forth below (or such other address as it may from time to time designate in writing to the other parties hereto):

If to Buyer or, after the Closing, the Company to:

Amedisys Hospice, L.L.C. c/o Amedisys, Inc. 3854 American Way Suite A Baton Rouge, LA 70816-4013 Facsimile: (225) 299-3796 Attn: Paul B. Kusserow Chief Executive Officer Email: paul.kusserow@amedisys.com

Amedisys Hospice, L.L.C. c/o Amedisys, Inc. 209 10th Avenue South, Suite 512 Nashville, Tennessee 37203 Facsimile: (225) 299-3796 Attn: David L. Kemmerly, Esq. General Counsel

Email: dave.kemmerly@amedisys.com

with courtesy copies to (which shall not constitute notice): Kantrow, Spaht, Weaver & Blitzer (APLC)
445 North Boulevard, Suite 300
Baton Rouge, LA 70802
Facsimile: (225) 383-4703
Attn: Lee C. Kantrow, Esq.
Jacob M. Kantrow, Esq.
Email: lee@kswb.com

jacob@kswb.com

If to the Company prior to the Closing, to:

Compassionate Care Hospice Group, Inc. Attn: Elliot D. Ostrove 200 Metroplex Drive Suite 304

Edison, NJ 08817

Email: e.ostrove@epsteinostrove.com

with courtesy copies to (which shall not constitute notice):

Lowenstein Sandler LLP 1251 Avenue of the Americas New York, NY 10020 Facsimile: (973) 535-3357 Attn: Marita A. Makinen, Esq. Email: MMakinen@lowenstein.com

If to the Sellers, to:

Milton Heching P.O. Box 1202 Clifton, NJ 07012 Email: miltonheching@protonmail.com

with courtesy copies to (which shall not constitute notice): Lowenstein Sandler LLP 1251 Avenue of the Americas New York, NY 10020 Facsimile: (973) 535-3357 Attn: Marita A. Makinen, Esq. Email: <u>MMakinen@lowenstein.com</u>

**8.5** <u>Headings</u>. The headings contained in this Agreement are inserted for convenience only and shall not be considered in interpreting or construing any of the provisions contained in this Agreement.

**8.6** <u>Fees and Expenses</u>. Except as otherwise specified in this Agreement, each party hereto shall bear its own costs and expenses (including investment advisory and legal fees and expenses) incurred in connection with this Agreement and the transactions contemplated hereby.

**8.7** Entire Agreement; Schedules. This Agreement (including the Exhibits and Disclosure Schedule hereto), the Escrow Agreement, and the other agreements, instruments and documents executed and delivered among the parties hereto at or in connection with the Closing constitute the entire agreement between the parties hereto with respect to the subject matter hereof and supersede all prior agreements and understandings between the parties with respect to such subject matter; provided, however, that this Agreement shall not supersede the terms and provisions of the Nondisclosure Agreement, which shall survive and remain in effect until expiration or termination thereof in accordance with its terms and Section 4.3 of this Agreement. Information set forth on any Disclosure Schedule shall be deemed to qualify each Section of this Agreement to which such Disclosure Schedule relates (or makes cross-reference), as well as representations and warranties in other Sections of this Agreement but only to the extent that the specific item on any such Disclosure Schedule is reasonably apparent on its face as being applicable to such other Section and only as related to such specific item. No information set forth on any Disclosure Schedule shall be deemed to purposes of this Agreement or otherwise, or that such item is a disclosure required under the Agreement. No disclosure in any Disclosure Schedule relating to any possible breach or violation of any agreement, permit, license, Law or Legal Requirement shall be construed as an admission or indication that any such breach or violation exists or has actually occurred, or shall constitute an admission of liability to any third party.

## 8.8 Interpretation.

(a) When a reference is made to an Article, Section, Exhibit or Schedule, such reference shall be to an Article, Section, Exhibit or Schedule of or to this Agreement unless otherwise indicated. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The conjunction "or" is used in the inclusive sense of "and/or." Unless the context requires otherwise, words using the singular or plural number also include the plural or singular number, respectively, and the use of any gender herein shall be deemed to include the other genders. References to "dollars" or "\$" are

to U.S. dollars. The terms "hereof," "herein," "hereby," "hereto" and derivative or similar words refer to this entire Agreement.

(b) This Agreement was prepared jointly by the parties hereto and no rule that it be construed against the drafter will have any application in its construction or interpretation.

**8.9** <u>Waiver and Amendment</u>. This Agreement may be amended, modified, supplemented or the performance hereof waived only by a written mutual agreement executed and delivered by Buyer and the Sellers. Any amendment, modification, supplement or waiver effected in accordance with this <u>Section 8.9</u> shall be binding on all parties hereto, regardless of whether any such party has consented thereto. No waiver or failure to insist upon strict compliance with any obligations, covenant, agreement or condition shall operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

**8.10** <u>Third-party Beneficiaries</u>. Except as otherwise specifically set forth in this Agreement, including, without limitation, in <u>Article VII</u>, this Agreement is for the sole benefit of the parties hereto and their permitted assigns and nothing herein express or implied shall give or be construed to give to any Person, other than the parties hereto and such permitted assigns, any legal or equitable rights hereunder.

**8.11** Non-Recourse. All claims or causes of action (whether in contract or in tort, in law or in equity) that may be based upon, arise out of or relate to this Agreement or the other agreements, instruments and documents contemplated hereby (the "Transaction Documents"), or the negotiation, execution or performance of this Agreement or the Transaction Documents (including any representation or warranty made in or in connection with this Agreement or the other Transaction Documents or as an inducement to enter into this Agreement or the other Transaction Documents), may be made only against the Persons that are expressly identified as parties hereto and thereto. No Person who is not a named party to this Agreement or the other Transaction Documents, including any past, present or future director, manager, officer, employee, incorporator, member, partner, equityholders (including stockholders and optionholders), Affiliate, agent, attorney or representative of any named party to this Agreement or the Transaction Documents ("Non-Party Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or based upon any theory that seeks to impose liability of an entity party against its owners or affiliates) for any obligations or liabilities arising under, in connection with or related to this Agreement or such other Transaction Documents (as the case may be) or for any claim based on, in respect of, or by reason of this Agreement or such other Transaction Documents (as the case may be) or the negotiation or execution hereof or thereof and each party hereto waives and releases all such liabilities, claims and obligations against any such Non-Party Affiliates. Non-Party Affiliates are expressly intended as third party beneficiaries of this provision of this Agreement.

**8.12** Severability. If any provision of this Agreement or the application of any such provision to any person or circumstance shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction under Section 8.3, such invalidity, illegality or unenforceability shall not affect any other provision hereof.

**8.13** <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in one or more counterpart signature pages, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same

agreement, which shall be binding upon all of the parties hereto notwithstanding the fact that all parties are not signatory to the same counterpart. The exchange of copies of this Agreement and of signature pages by facsimile transmission, by electronic mail in "portable document format" (".pdf") form, or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document bearing an original signature.

## 8.14 Specific Performance.

(a) The parties hereto agree that irreparable damage for which monetary relief, even if available, would not be an adequate remedy, would occur in the event that any provision of this Agreement is not performed in accordance with its specific terms or is otherwise breached, including if the parties hereto fail to take any action required of them hereunder to consummate the transactions contemplated hereby. Subject to the following sentence and <u>Section 7.7</u>, the parties acknowledge and agree that (i) the parties shall be entitled to seek an injunction or injunctions, specific performance or other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in <u>Section 8.3</u>, without proof of damages or otherwise, this being in addition to any other remedy to which they are entitled under this Agreement (including any Actions for damages) and (ii) the right of specific enforcement is an integral part of the transactions contemplated by this Agreement and without that right, none of the parties hereto would have entered into this Agreement. The parties hereto agree not to assert that a remedy of specific enforcement brought in accordance with this Agreement is unenforceable, invalid, contrary to Law or inequitable for any reason, and not to assert that a remedy of monetary damages would provide an adequate remedy or that the parties otherwise have an adequate remedy at law. The parties hereto acknowledge and agree that any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with any such order or injunction.

(b) To the extent any party hereto brings any Action in accordance with this Agreement to enforce specifically the performance of the terms and provisions of this Agreement prior to the Closing, the Outside Date shall automatically be extended by (i) the amount of time during which such Action is pending, plus twenty (20) Business Days, or (ii) such other time period established by the court presiding over such Action.

**8.15** <u>Provision Respecting Representation of the Sellers.</u> Each of the Sellers, the Company, Buyer and Amedisys hereby irrevocably acknowledge and agree that, on its or their own behalf and on behalf of its directors, members, managers, partners, equityholders, officers, employees and Affiliates, the Sellers shall have the right to retain Lowenstein Sandler LLP to represent their interests in any actual or potential dispute or Action arising under or in connection with this Agreement, or the transactions contemplated hereby or thereby (a "Dispute"). The Sellers, the Company, Buyer and Amedisys acknowledge that Lowenstein Sandler LLP has represented and performed services for the Company prior to the Closing and consent to the sharing by Lowenstein Sandler LLP (or any of its successors) with the Sellers of all information obtained during such past representation that Lowenstein Sandler LLP (or any of its successors) believes to be relevant to any

Dispute. Buyer consents and covenants not to assert any objection, based on conflict of interest or otherwise, to any representation of either Seller by Lowenstein Sandler LLP in any Dispute. The Sellers, the Company, Buyer and Amedisys further agree and acknowledge that all communications between or among the Sellers, the Company or any of their respective Affiliates, directors, officers, employees, owners or representatives, on the one hand, and Lowenstein Sandler LLP, on the other hand, made in connection with the negotiation, preparation, execution, delivery and closing under this Agreement, or any Dispute (the "Protected Seller Communications, and the control of the confidentiality and privilege applicable thereto, shall be assigned back to the Sellers for purposes of any Dispute. Notwithstanding the foregoing, in the event that a dispute arises between the Buyer or the Company and a third party other than a party to this Agreement after the Closing, the Company may assert the attorney-client privilege to prevent disclosure of confidential communications by Lowenstein Sandler LLP to such third party; provided, however, that the Company may not waive such privilege for Protected Seller Communications without the prior written consent of the Sellers.

## [Signature Pages Follow]

**IN WITNESS WHEREOF,** the parties hereto have caused this Stock Purchase Agreement to be executed as of the day and year first above written.

## COMPANY:

## COMPASSIONATE CARE HOSPICE GROUP, INC.

By: <u>/s/ Milton Heching</u> Name: Milton Heching Title: COB

## **SELLERS:**

## MILTON HECHING

<u>/s/ Milton Heching</u> Milton Heching, individually

## **HECHING 2012 EXEMPT IRREVOCABLE TRUST**

By: <u>/s/ Bella Heching</u> Name: Bella Heching Title: Trustee **IN WITNESS WHEREOF,** the parties hereto have caused this Stock Purchase Agreement to be executed as of the day and year first above written.

## **BUYER:**

## AMEDISYS HOSPICE, L.L.C.

- By: Amedisys Holding, L.L.C., Member-Manager
- By: Amedisys, Inc., Member-Manager

By: <u>/s/ Paul B. Kusserow</u> Name: Paul B. Kusserow Title: Chief Executive Officer

### AMEDISYS (for purposes of Sections 3.4, 4.3(a), 4.15 and ARTICLE VIII only):

AMEDISYS, INC.

By: <u>/s/ Paul B. Kusserow</u> Name: Paul B. Kusserow Title: Chief Executive Officer

(Back To Top)

## Section 3: EX-10.1 (EXHIBIT 10.1)

Exhibit 10.1

## AMENDMENT TO THE AMEDISYS, INC. 2008 OMNIBUS INCENTIVE COMPENSATION PLAN

This Amendment (this "Amendment") to the Amedisys, Inc. 2008 Omnibus Incentive Compensation Plan (the "Plan") is entered into this 25th day of September, 2018, by the Compensation Committee of the Board of Directors (the "Committee") of Amedisys, Inc. (the "Company"), as authorized pursuant to Section 19 of the Plan.

**WHEREAS**, the Company established and maintains the Plan, as it may be amended and restated from time to time, although no future Awards may be granted pursuant to the Plan as a result of the Company's stockholders approving the Company's 2018 Omnibus Incentive Compensation Plan on June 6, 2018;

WHEREAS, Section 19 of the Plan permits the Committee to amend the Plan without stockholder approval as set forth

therein; and

**WHEREAS**, the Committee now desires to amend the Plan to (i) amend the definition of "Qualifying Termination" to include termination by the Participant (as defined in the Plan) for "Good Reason," (ii) add a definition of "Good Reason" to the Plan, and (iii) change the protection period following a Change in Control (as defined in the Plan) from one year to two years.

**NOW, THEREFORE**, for Awards granted under the Plan that vest on and after September 25, 2018, the Plan is hereby amended as follows:

1.

A new Section 2.23 is hereby added to the Plan, to read as follows:

## 2.23 "Good Reason" shall mean:

- a. If a Participant is a party to an employment agreement with the Company and such agreement provides for a definition of Good Reason, the definition contained in such agreement; or
- b.
- c. If a Participant is a "Covered Executive" as defined in the Amedisys Holding, L.L.C. Severance Plan for Key Employees (the "Key Executive Severance Plan"), the definition of Good Reason contained in the Key Executive Severance Plan; or
- d.
- e. If no such employment agreement exists or if such agreement does not define Good Reason, and the Participant is not a "Covered Executive" under the Key Executive Severance Plan, the occurrence of one or more of the following without the Participant's express written consent: (i) Participant suffers a material diminution in authority, responsibilities, or duties; or (ii) Participant suffers a material reduction in base salary other than in connection with a proportionate reduction in the base salaries of all similarly situated senior officer-level employees.

2.

The Section numbers in Section 2 "Definitions" of the Plan are hereby increased by one, beginning with former Section 2.23, the definition of "Immediate Family," which shall be new Section 2.24.

3.

Section 2.21 of the Plan is hereby amended by replacing the cross-reference to "Section 2.34" to read "Section 2.35."

4.

Section 2.32 of the Plan, as re-numbered to Section 2.33 pursuant to Section 2 of this Amendment, is hereby deleted in its entirety and replaced with the following:

**2.35** "Qualifying Event" means, with respect to a Participant, (i) a termination of such Participant's Employment by the Company (and all then-Affiliates or Subsidiaries) without Cause following a Change in Control of the Company, (ii) a termination of such Participant's Employment by the Participant for Good Reason following a Change in Control of the Company, or (iii) a relocation of the Participant's principal place of employment by more than 50 miles. It is understood that a Participant shall not have a Qualifying Event pursuant to clause (i) above by virtue of ceasing to be Employed by an entity or its subsidiaries undergoing a Change in Control where, following such Change in Control, the Participant remains employed by an entity that is at least one of (i) the Company or (ii) any entity that was an Affiliate or Subsidiary undergoing a Change in Control immediately prior to such Change in Control. Notwithstanding the foregoing, payments on account of a Participant's Qualifying Event that constitute "deferred compensation" within the meaning of Section 409A of the Code shall not commence unless and until the Participant has also incurred a "separation from service" within the meaning of Code Section 409A.

5.

Section 22 of the Plan is hereby amended by replacing "one year" each place it occurs with "two years" and by replacing the defined term "Qualifying Termination" in the first sentence with the term "Qualifying Event."

6.

Except as specifically set forth above, the terms of the Plan shall continue in full force and effect.

[Signature on following page.]

IN WITNESS WHEREOF, the Committee has caused this Amendment to be adopted and executed effective September 25, 2018.

## Amedisys, Inc.

By: <u>/s/ Julie D. Klapstein</u>

Name: Julie D. Klapstein

Title: Compensation Committee Chair

(Back To Top)

# Section 4: EX-10.2 (EXHIBIT 10.2)

Exhibit 10.2

## AMENDMENT TO THE AMEDISYS, INC. 2018 OMNIBUS INCENTIVE COMPENSATION PLAN

This Amendment (this "Amendment") to the Amedisys, Inc. 2018 Omnibus Incentive Compensation Plan (the "Plan") is entered into this 25th day of September, 2018, by the Compensation Committee of the Board of Directors (the "Committee") of Amedisys, Inc. (the "Company"), as authorized pursuant to Section 19 of the Plan.

WHEREAS, the Company established and maintains the Plan, as it may be amended and restated from time to time;

WHEREAS, Section 19 of the Plan permits the Committee to amend the Plan without stockholder approval as set forth therein; and

**WHEREAS**, the Committee now desires to amend the Plan to (i) amend the definition of "Qualifying Termination" to include termination by the Participant (as defined in the Plan) for "Good Reason," (ii) add a definition of "Good Reason" to the Plan, and (iii) change the protection period following a Change in Control (as defined in the Plan) from one year to two years.

**NOW, THEREFORE**, for Awards that vest or are granted on and after September 25, 2018, the Plan is hereby amended as follows:

1.

A new Section 2.25 is hereby added to the Plan, to read as follows:

## 2.25 "Good Reason" shall mean:

- a. If a Participant is a party to an employment agreement with the Company and such agreement provides for a definition of Good Reason, the definition contained in such agreement; or
- b.
- c. If a Participant is a "Covered Executive" as defined in the Amedisys Holding, L.L.C. Severance Plan for Key Employees (the "Key Executive Severance Plan"), the definition of Good Reason contained in the Key

Executive Severance Plan; or

- d.
- e. If no such employment agreement exists or if such agreement does not define Good Reason, and the Participant is not a "Covered Executive" under the Key Executive Severance Plan, the occurrence of one or more of the following without the Participant's express written consent: (i) Participant suffers a material diminution in authority, responsibilities, or duties; or (ii) Participant suffers a material reduction in base salary other than in connection with a proportionate reduction in the base salaries of all similarly situated senior officer-level employees.

The Section numbers in Section 2 "Definitions" of the Plan are hereby increased by one, beginning with former Section 2.25, the definition of "Immediate Family," which shall be new Section 2.26.

Section 2.22 of the Plan is hereby amended by replacing the cross-reference to "Section 2.36" to read "Section 2.37."

## 4.

Section 2.34 of the Plan, as re-numbered to Section 2.35 pursuant to Section 2 of this Amendment, is hereby deleted in its entirety and replaced with the following:

**2.35** "Qualifying Event" means, with respect to a Participant, (i) a termination of such Participant's Employment by the Company (and all then-Affiliates or Subsidiaries) without Cause following a Change in Control of the Company, (ii) a termination of such Participant's Employment by the Participant for Good Reason following a Change in Control of the Company, or (iii) a relocation of the Participant's principal place of employment by more than 50 miles. It is understood that a Participant shall not have a Qualifying Event pursuant to clause (i) above by virtue of ceasing to be Employed by an entity or its subsidiaries undergoing a Change in Control where, following such Change in Control, the Participant remains employed by an entity that is at least one of (i) the Company or (ii) any entity that was an Affiliate or Subsidiary undergoing a Change in Control immediately prior to such Change in Control. Notwithstanding the foregoing, payments on account of a Participant's Qualifying Event that constitute "deferred compensation" within the meaning of Section 409A of the Code shall not commence unless and until the Participant has also incurred a "separation from service" within the meaning of Code Section 409A.

5.

Section 22 of the Plan is hereby amended by replacing "one year" each place it occurs with "two years" and by replacing the defined term "Qualifying Termination" in the first sentence with the term "Qualifying Event."

6.

Except as specifically set forth above, the terms of the Plan shall continue in full force and effect.

IN WITNESS WHEREOF, the Committee has caused this Amendment to be adopted and executed effective September 25, 2018.

## Amedisys, Inc.

By: <u>/s/ Julie D. Klapstein</u>

Name: Julie D. Klapstein\_\_\_\_\_

Title: Compensation Committee Chair\_\_\_\_

(Back To Top)

## Section 5: EX-31.1 (EXHIBIT 31.1)

#### CERTIFICATION

#### I, Paul B. Kusserow, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q for the quarter ended September 30, 2018, of Amedisys, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15 (f) and 15d-15(f)) for the registrant and have:

a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 30, 2018

/S/ Paul B. Kusserow

Paul B. Kusserow President and Chief Executive Officer (Principal Executive Officer)

(Back To Top)

## Section 6: EX-31.2 (EXHIBIT 31.2)

Exhibit 31.2

#### CERTIFICATION

I, Scott G. Ginn, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q for the quarter ended September 30, 2018, of Amedisys, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period

covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15 (f) and 15d-15(f)) for the registrant and have:

a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 30, 2018

/s/ Scott G. Ginn

Scott G. Ginn Chief Financial Officer (Principal Financial Officer)

(Back To Top)

# Section 7: EX-32.1 (EXHIBIT 32.1)

Exhibit 32.1

### CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Amedisys, Inc. (the "Company") on Form 10-Q for the quarter ended September 30, 2018 (the "Report"), I, Paul B. Kusserow, President and Chief Executive Officer of the Company, hereby certify to my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: October 30, 2018

/S/ Paul B. Kusserow

Paul B. Kusserow President and Chief Executive Officer (Principal Executive Officer)

(Back To Top)

# Section 8: EX-32.2 (EXHIBIT 32.2)

Exhibit 32.2

### CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Amedisys, Inc. (the "Company") on Form 10-Q for the quarter ended September 30, 2018 (the "Report"), I, Scott G. Ginn, Chief Financial Officer of the Company, hereby certify to my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: October 30, 2018

/S/ Scott G. Ginn

Scott G. Ginn Chief Financial Officer (Principal Financial Officer)

(Back To Top)